

Volume 18 2024

On the ordinary Hecke orbit conjecture

Pol van Hoften



On the ordinary Hecke orbit conjecture

Pol van Hoften

We prove the ordinary Hecke orbit conjecture for Shimura varieties of Hodge type at primes of good reduction. We make use of the global Serre–Tate coordinates of Chai as well as recent results of D'Addezio about the monodromy groups of isocrystals. The new ingredients in this paper are a general monodromy theorem for Hecke-stable subvarieties for Shimura varieties of Hodge type, and a rigidity result for the formal completions of ordinary Hecke orbits. Along the way, we show that classical Serre–Tate coordinates can be described using unipotent formal groups, generalising a result of Howe.

1.	Introduction	847
2.	Integral models of Shimura varieties of Hodge type	850
3.	Monodromy of Hecke-invariant subvarieties	856
4.	Serre–Tate coordinates and unipotent group actions	870
5.	The formal neighbourhood of an ordinary point	876
6.	Proof of the main theorems	887
Acknowledgements		896
References		896

1. Introduction

Let $A_{g,n}$ be the moduli space of g-dimensional principally polarised abelian varieties (A, λ) with level $n \geq 3$ structure over $\overline{\mathbb{F}}_p$, for a prime number p coprime to n. Recall that there are finite étale prime-to-p Hecke correspondences from $A_{g,n}$ to itself, and that two points $x, y \in A_{g,n}(\overline{\mathbb{F}}_p)$ are said to be *in the same prime-to-p Hecke orbit* if they share a preimage under one of these correspondences. Recall the following result of Chai:

Theorem [Chai 1995]. Let $x \in A_{g,n}(\overline{\mathbb{F}}_p)$ be a point corresponding to an ordinary principally polarised abelian variety. Then the prime-to-p Hecke orbit of x is Zariski dense in $A_{g,n}$.

Our main result is a generalisation of this theorem to Shimura varieties of Hodge type. To state it, we will first introduce some notation.

1.1. *Main results.* Let (G, X) be a Shimura datum of Hodge type with reflex field E and let p be a prime number. Let $K_p \subset G(\mathbb{Q}_p)$ be a hyperspecial subgroup and let $K^p \subset G(\mathbb{A}_f^p)$ be a sufficiently small compact open subgroup. Let Sh_G be the special fibre of the canonical integral model of the Shimura variety of level K^pK_p at a prime v above p of E, constructed in [Kisin 2010; Kim and Madapusi Pera 2016].

MSC2020: primary 11G18; secondary 14G35.

Keywords: Shimura varieties, Hecke orbit conjecture, ordinary locus, monodromy theorems, Serre–Tate coordinates, Rigidity theorem, local stabiliser principle.

^{© 2024} The Author, under license to MSP (Mathematical Sciences Publishers). Distributed under the Creative Commons Attribution License 4.0 (CC BY). Open Access made possible by subscribing institutions via Subscribe to Open.

Let E_v be the v-adic completion of E, which is a finite extension of \mathbb{Q}_p . There is a closed immersion $\operatorname{Sh}_{G,\overline{\mathbb{F}}_p} \to \mathcal{A}_{g,n}$ for some n (see [Xu 2020]), and the intersection $\operatorname{Sh}_{G,\operatorname{ord}}$ of the ordinary locus of $\mathcal{A}_{g,n}$ with Sh_G is nonempty if and only if $E_v = \mathbb{Q}_p$ (see [Lee 2018, Corollary 1.0.2]). Recall that there are prime-to-p Hecke correspondences over Sh_G , which we use to define prime-to-p Hecke orbits.

Theorem I. If $E_v = \mathbb{Q}_p$, then the prime-to-p Hecke orbit of a point $x \in \operatorname{Sh}_{G,\operatorname{ord}}(\overline{\mathbb{F}}_p)$ is Zariski dense in Sh_G .

Theorem I generalises results of Maulik–Shankar–Tang, see [Maulik et al. 2022], who deal with GSpin Shimura varieties associated to a quadratic space over $\mathbb Q$ and GU(1,n-1) Shimura varieties associated to imaginary quadratic fields E with p split in E; their methods are completely disjoint from ours. There is also work of Shankar [2016] for Shimura varieties of type C, using a group-theoretic version of Chai's strategy of using hypersymmetric points and reducing to the case of Hilbert modular varieties. Shankar crucially proves that the Hodge map $\operatorname{Sh}_{G,\overline{\mathbb F}_p} \to \mathcal A_{g,n}$ is a closed immersion over the ordinary locus via canonical liftings, whereas we use work of Xu [2020].

Last we mention work of Zhou [2023], who proves the Hecke orbit conjecture for the μ -ordinary locus of certain quaternionic Shimura varieties. Our results do not imply his, but there is some overlap between the cases that we cover.

A fairly direct consequence of Theorem I is a density result for prime-to-p Hecke orbits of an $\overline{\mathbb{F}}_p$ -point in the μ -ordinary locus of a Shimura variety of abelian type, at primes v above p of the reflex field E where $E_v = \mathbb{Q}_p$, see Corollary 6.4.1.

1.2. *Monodromy theorems.* An important ingredient in our proof is an ℓ -adic monodromy theorem for prime-to-p Hecke-stable subvarieties of special fibres of Shimura varieties, in the style of [Chai 2005, Corollary 3.5]. To state it, let (G, X) be as above and assume for simplicity that G^{ad} is simple over \mathbb{Q} . Let V_{ℓ} be the rational ℓ -adic Tate module of the abelian variety A over Sh_G coming from the map $\mathrm{Sh}_{G,\overline{\mathbb{F}}_p} \to \mathcal{A}_{g,n}$; it is an ℓ -adic local system of rank 2g.

Theorem II. Let $Z \subset \operatorname{Sh}_G$ be a smooth locally closed subvariety that is stable under the prime-to-p Hecke operators. Suppose that Z is not contained in the smallest Newton stratum of Sh_G . Let $z \in Z(\overline{\mathbb{F}}_p)$ and let $Z^{\circ} \subset Z_{\overline{\mathbb{F}}_p}$ be the connected component of Z containing z. Then the neutral component $\operatorname{M}_{\operatorname{geom}}$ of the Zariski closure of the image of the monodromy representation

$$\rho_{\ell,\text{geom}}: \pi_1^{\acute{e}t}(Z^\circ, z) \to \text{GL}_{2g}(\mathbb{Q}_\ell)$$

corresponding to V_{ℓ} , is isomorphic to $G_{\mathbb{Q}_{\ell}}^{\mathrm{der}}$.

This generalises work of Chai [2005] in the Siegel case and others [Kasprowitz 2012; Xiao Xiao 2020] in the PEL case.

In the body of the paper, we work with the integral models of Shimura varieties of Hodge type of level $K_p \subset G(\mathbb{Q}_p)$ constructed in [KMS 2022]. Here K_p is not required to be hyperspecial, for example it is allowed to be any (connected) parahoric subgroup. Our results, namely Theorem 3.2.5 and Corollary 3.2.6,

are proved under the assumption that Hypothesis 2.3.1 holds. This hypothesis holds for example when $G_{\mathbb{Q}_p}$ is quasi-split and has no factors of type D, or when K_p is hyperspecial.

We also prove results about irreducible components of smooth locally closed subvarieties that are stable under the prime-to-p Hecke operators, in the style of [Chai 2005, Proposition 4.4], see Theorem 3.4.10. These results are used to prove irreducibility of Ekedahl–Oort strata in [van Hoften 2020].

1.2.1. An overview of the proof of Theorem II. Since $Z^{\circ} \subset \operatorname{Sh}_G$ is defined over a finite field k we can write it as $Z_k^{\circ} \otimes_k \overline{\mathbb{F}}_p$. We can then consider the Zariski closure \mathbb{M} of the image of

$$\rho_{\ell}: \pi_1^{\text{\'et}}(Z_k^{\circ}, z) \to \mathrm{GL}_{2g}(\mathbb{Q}_{\ell}).$$

An argument from [Chai 2005] proves that \mathbb{M} is (isomorphic to) a normal subgroup of $G_{\mathbb{Q}_{\ell}}$. If $G_{\mathbb{Q}_{\ell}}^{ad}$ was a simple-group, then we would be done if we could show that \mathbb{M} was not central in $G_{\mathbb{Q}_{\ell}}$. However, in general there are *no* primes ℓ such that $G_{\mathbb{Q}_{\ell}}^{ad}$ is simple and so at this point we have to deviate from the strategy of [Chai 2005].

Instead, we control \mathbb{M} by studying the centraliser $I_{x,\ell} \subset G_{\mathbb{Q}_\ell}$ of the image of Frobenius elements $\operatorname{Frob}_x \in \pi_1^{\operatorname{\acute{e}t}}(Z_k^\circ, z)$ corresponding to points $x \in Z_k(\mathbb{F}_q)$. Since the paper [KMS 2022] makes an in-depth study of these Frobenius elements, we can make use of their results about these centralisers. For example, if x is not contained in the basic locus, then they prove that Frob_x is not central. To get more precise results, we need to know that the element $\operatorname{Frob}_x \in G(\mathbb{Q}_\ell)$ is *defined over* \mathbb{Q} , which is what Hypothesis 2.3.1 makes precise.

In this way we can show that $\mathbb{M} \subset G_{\mathbb{Q}_{\ell}}$ is a normal subgroup that surjects onto $G_{\mathbb{Q}_{\ell}}^{\mathrm{ad}}$. The result about $\mathbb{M}_{\mathrm{geom}} \subset \mathbb{M}$ will be deduced from this.

1.3. A sketch of the proof of Theorem I. Let $x \in \operatorname{Sh}_G(\overline{\mathbb{F}}_p)$ be an ordinary point, and let Z be the Zariski closure inside $\operatorname{Sh}_{G,\operatorname{ord}}$ of the prime-to-p Hecke orbit of x. Let $y \in Z(\overline{\mathbb{F}}_p)$ be a smooth point of Z. Recall that it follows from the theory of Serre–Tate coordinates that the formal completion $\mathcal{A}_{g,n}^{/y}$ of $\mathcal{A}_{g,n}$ at y is a formal torus. A special case of the main result of [Shankar and Zhou 2021] tells us that

$$S^{/y} := \operatorname{Sh}_{G,\overline{\mathbb{F}}_n}^{/y} \subset \mathcal{A}_{g,n}^{/y}$$

is a formal subtorus. Work of Chai on the deformation theory of ordinary p-divisible groups [Chai 2003] tells us that the dimension of the smallest formal subtorus of $S^{/y}$ containing $Z^{/y}$, is encoded in the unipotent radical of the p-adic monodromy group of the isocrystal \mathcal{M} associated to the universal abelian variety A over Z.

Using Theorem II and results of D'Addezio [2020; 2023], we compute the monodromy group of \mathcal{M} over Z. It follows from this computation that the smallest formal subtorus of $S^{/y}$ containing $Z^{/y}$ is equal to $S^{/y}$.

We conclude by proving that the formal completion $Z^{/y}$ is a formal subtorus of $S^{/y}$. By the rigidity theorem for p-divisible formal groups of [Chai 2008], it suffices to give a representation-theoretic

description of the Dieudonné module of $S^{/y}$. Unfortunately, the description of the subtorus $S^{/y}$ coming out of the work of Shankar and Zhou [2021] does not readily lend itself to understanding its Dieudonné module.

Instead, we give a different proof that $S^{/y}$ is a subtorus of $\mathcal{A}_{g,n}^{/y}$. We do this by giving a new description of Serre–Tate coordinates in terms of actions of formal unipotent groups on Rapoport–Zink spaces, generalising results of Howe [2020] in the case g = 1. Once we have this perspective, the results of Kim [2019] give an explicit description of the Dieudonné module of the torus $\mathcal{A}_{g,n}^{/y}$ as well as the Dieudonné module of the subtorus $S^{/y}$.

1.4. *Outline.* Sections 2 and 3 form the first part of the paper and work in a more general setting than the rest of the paper. In Section 2 we introduce the integral models of Shimura varieties of Hodge type constructed in [KMS 2022]. We recall results and notation from [loc. cit.], in particular, about the Frobenius elements and their centralisers associated to $\overline{\mathbb{F}}_p$ -points of these models. In Section 3 we prove monodromy theorems for Hecke-stable subvarieties of the special fibres of these integral models, combining results of [KMS 2022] with ideas of Chai [2005].

Section 4 is a standalone section on Serre–Tate coordinates. In it, we show that the classical Serre–Tate coordinates, as described in [Katz 1981], can be reinterpreted using actions of unipotent formal groups as in [Howe 2020]. This section should be of independent interest.

In Section 5, we specialise to the smooth canonical integral models of Shimura varieties of Hodge type at hyperspecial level, and we moreover assume that the ordinary locus is nonempty. We reprove a result of [Shankar and Zhou 2021], which states that the formal completion of the ordinary locus gives a subtorus of the Serre–Tate torus, and give a group-theoretic description of its Dieudonné module. At the end of this section we also give a short interlude on strongly nontrivial actions of algebraic groups on isocrystals, which we will need to confirm the hypotheses of the rigidity theorem of [Chai 2008].

In Section 6, we put everything together and prove Theorem I. We end by deducing a result for Shimura varieties of abelian type.

2. Integral models of Shimura varieties of Hodge type

Let (G, X) be a Shimura datum of Hodge type. In this section we follow [KMS 2022, Section 1.3] and construct integral models for the Shimura varieties associated to (G, X) in a very general situation. The main goal is to introduce various Frobenius elements $\gamma_{x,m,\ell} \in G(\mathbb{Q}_{\ell})$ associated to \mathbb{F}_{q^m} -points of these integral models, and to discuss result of [KMS 2022] about their centralisers $I_{x,m,\ell}$. We end by introducing Hypothesis 2.3.1, which will be assumed throughout Section 3, and prove that it holds under minor assumptions.

2.0.1. Hodge cocharacters. If (G, X) is a Shimura datum, then for each $x \in X$ there is a cocharacter $\mu_x : \mathbb{G}_{m,\mathbb{C}} \to G_{\mathbb{C}}$, see [KMS 2022, Section 1.2.3] for the precise definition. The $G(\mathbb{C})$ -conjugacy class of μ_x does not depend on the choice of x and we will write $\{\mu_X\}$ for this conjugacy class, and denote it by $\{\mu\}$ if X is clear from context. This conjugacy class of cocharacters is defined over a number field $E \subset \mathbb{C}$, called the reflex field.

2.1. The construction of integral models. For a symplectic space (V, ψ) over \mathbb{Q} we write $\mathcal{G}_V := \operatorname{GSp}(V, \psi)$ for the group of symplectic similitudes of V over \mathbb{Q} . It admits a Shimura datum \mathcal{H}_V consisting of the union of the Siegel upper and lower half spaces. Let (G, X) be a Shimura datum of Hodge type with reflex field E and let $(G, X) \to (\mathcal{G}_V, \mathcal{H}_V)$ be a Hodge embedding.

Fix a prime p and choose a $\mathbb{Z}_{(p)}$ -lattice $V_{(p)} \subset V$ on which ψ is $\mathbb{Z}_{(p)}$ -valued, and write $V_p = V_{(p)} \otimes_{\mathbb{Z}_{(p)}} \mathbb{Z}_p$. Write $\mathcal{K}_p \subset \mathcal{G}_V(\mathbb{Q}_p)$ for the stabiliser of V_p in $\mathcal{G}_V(\mathbb{Q}_p)$, and similarly write K_p for the stabiliser of V_p in $G(\mathbb{Q}_p)$. For every sufficiently small compact open subgroup $K^p \subset G(\mathbb{A}_f^p)$ we can find $\mathcal{K}^p \subset \mathcal{G}_V(\mathbb{A}_f^p)$ such that the Hodge embedding induces a closed immersion (see [Kisin 2010, Lemma 2.1.2])

$$\mathbf{Sh}_K(G, X) \to \mathbf{Sh}_K(\mathcal{G}_V, \mathcal{H}_V)_E$$

of Shimura varieties of level $K = K^p K_p$ and $K = K^p K_p$, respectively. We let S_K over $\mathbb{Z}_{(p)}$ be the moduli-theoretic integral model of $\mathbf{Sh}_K(\mathcal{G}_V, \mathcal{H}_V)$; it is a moduli space of (weakly) polarised abelian schemes (A, λ) up to prime-to-p isogeny with level K^p -structure.

Fix a prime $v \mid p$ of E and let

$$\mathscr{S}_K := \mathscr{S}_K(G, X) \to \mathscr{S}_K \otimes_{\mathbb{Z}_{(p)}} \mathscr{O}_{E,(v)}$$

be the normalisation of the Zariski closure of $\mathbf{Sh}_K(G, X)$ in $\mathcal{S}_K \otimes_{\mathbb{Z}_{(p)}} \mathcal{O}_{E,(v)}$. This construction is compatible with changing the level away from p and we define

$$\mathcal{S}_{\mathcal{K}_p} \coloneqq \varprojlim_{\mathcal{K}^p \subset \mathcal{G}_V(\mathbb{A}_f^p)} \mathcal{S}_{\mathcal{K}^p \mathcal{K}_p}, \quad \mathscr{S}_{K_p} \coloneqq \varprojlim_{K^p \subset G(\mathbb{A}_f^p)} \mathscr{S}_{K^p K_p}.$$

Then, as discussed in [KMS 2022, Section 2.1], the transition maps in both inverse systems are finite étale and moreover $G(\mathbb{A}_f^p)$ acts on \mathscr{S}_{K_p} . Let $k = \mathbb{F}_q$ be the residue field of $\mathcal{O}_{E,(v)}$, and write $\operatorname{Sh}_{G,K_p}$ for the special fibre of $\mathscr{S}_{K^pK_p}$; these are both schemes over k and $G(\mathbb{A}_f^p)$ acts on $\operatorname{Sh}_{G,K_p}$. We will write $\operatorname{Sh}_{\mathcal{G}_V,\mathcal{K}^p\mathcal{K}_p}$ for the special fibre of $\mathscr{S}_{\mathcal{K}^p\mathcal{K}_p} \otimes_{\mathbb{Z}_{(p)}} \mathcal{O}_{E,(v)}$ and $\operatorname{Sh}_{\mathcal{G}_V,\mathcal{K}_p}$ for the special fibre of $\mathscr{S}_{\mathcal{K}_p} \otimes_{\mathbb{Z}_{(p)}} \mathcal{O}_{E,(v)}$.

Let \mathbb{V}^p be the prime-to-p adelic Tate module of the universal abelian variety \mathcal{A} over $\mathcal{S}_{\mathcal{K}_p}$; this is a pro-étale local system on $\mathcal{S}_{\mathcal{K}_p}$. For a morphism x: Spec $R \to \mathcal{S}_{\mathcal{K}_p}$ we will write \mathbb{V}_x^p for the pullback along x of \mathbb{V}^p . As explained in [KMS 2022, Section 2.1.1] there is a universal isomorphism

$$\epsilon: V \otimes \underline{\mathbb{A}}_f^p \simeq \mathbb{V}^p,$$

sending the symplectic form ψ to an $\underline{\mathbb{A}}_f^{p,\times}$ -multiple of the Weil pairing. Here $\underline{\mathbb{A}}_f^p$ denotes the pro-étale sheaf associated to the topological group \mathbb{A}_f^p .

¹It is explained in [KMS 2022, Section 1.3.2] that the collection of subgroups $K_p \subset G(\mathbb{Q}_p)$ that can arise from this construction by varying the symplectic space and the Hodge embedding contains all stabilisers of vertices in the extended Bruhat–Tits building of $G_{\mathbb{Q}_p}$. It is moreover explained in [loc. cit.] that this collection is stable under finite intersections.

2.1.1. Tensors. Write V^{\otimes} for the direct sum of $V^{\otimes n} \otimes (V^*)^{\otimes m}$ for all pairs of integers $m \geq 0$, $n \geq 0$. We will also use this notation later for modules over commutative rings and modules over sheaves of rings.

As in [KMS 2022, Section 1.3.4], we fix tensors $\{s_{\alpha} \in V\} \subset V^{\otimes}$ such that G is their pointwise stabiliser in GL(V). Then as explained in [KMS 2022, Sections 1.3.4 and 2.1.2], there are global sections

$$\{s_{\alpha,\mathbb{A}_{f}^{p}}\}\in H^{0}(\mathscr{S}_{K^{p}K_{p}},(\mathbb{V}^{p})^{\otimes})$$

such that if we restrict the isomorphism ϵ via $\mathscr{S}_{K_p} \to \mathscr{S}_{K_p}$ we get an isomorphism

$$\eta: V \otimes \underline{\mathbb{A}}_f^p \to \mathbb{V}^p$$

taking $s_{\alpha} \otimes 1$ to $s_{\alpha, \mathbb{A}_{f}^{p}}$ for all α . In particular, for each $x \in \mathscr{S}_{K_{p}}(\overline{\mathbb{F}}_{p})$ the stabiliser of the tensors $\{s_{\alpha, \mathbb{A}_{f}^{p}, x}\}$ in $GL(\mathbb{V}_{x}^{p})$ is canonically identified with $G \otimes \mathbb{A}_{f}^{p}$.

2.1.2. Let $\overline{\mathbb{F}}_p$ denote an algebraic closure of \mathbb{F}_p . We will use $\check{\mathbb{Z}}_p$ to denote the p-typical Witt vectors $W(\overline{\mathbb{F}}_p)$ of $\overline{\mathbb{F}}_p$ and we set $\check{\mathbb{Q}}_p = \check{\mathbb{Z}}_p[1/p]$. We let $\sigma : \check{\mathbb{Z}}_p \to \check{\mathbb{Z}}_p$ be the automorphism induced by Frobenius on $\overline{\mathbb{F}}_p$, and also denote by σ the induced automorphism of $\check{\mathbb{Q}}_p$.

Let $x \in \operatorname{Sh}_{G,K^pK_p}(\overline{\mathbb{F}}_p)$ and let \mathbb{D}_x be the rational contravariant Dieudonné module of the p-divisible group $A_x[p^\infty]$ of the abelian variety A_x , equipped with its Frobenius ϕ . By [KMS 2022, Proposition 1.3.7] there are ϕ -invariant tensors $\{s_{\alpha,\operatorname{cris},x}\}\subset \mathbb{D}_x^{\otimes}$ and in [KMS 2022, Section 1.3.8] it is argued that there is an isomorphism $\mathbb{Q}_p\otimes V\to \mathbb{D}_x$ sending $1\otimes s_\alpha$ to $s_{\alpha,\operatorname{cris},x}$. See the statement of [KMS 2022, Proposition 1.3.7] for a characterisation of the tensors $s_{\alpha,\operatorname{cris}}$.

Under such an isomorphism, the Frobenius ϕ corresponds to an element $b_x \in G(\mathring{\mathbb{Q}}_p)$, which is well defined up to σ -conjugacy, where $\sigma: G(\mathring{\mathbb{Q}}_p) \to G(\mathring{\mathbb{Q}}_p)$ is induced by $\sigma: \mathring{\mathbb{Q}}_p \to \mathring{\mathbb{Q}}_p$. In other words, we can associate to ϕ a well defined element $[b_x]$ of the Kottwitz set $B(G) = B(G_{\mathbb{Q}_p})$ of [Kottwitz 1985]. By [KMS 2022, Lemma 1.3.9], the element $[b_x]$ is contained in the neutral acceptable set $B(G, \{\mu^{-1}\})$ consisting of the $\{\mu^{-1}\}$ -admissible elements defined in [KMS 2022, Section 1.1.5]. Here we use $\{\mu\}$ to denote the $G(\overline{\mathbb{Q}}_p)$ conjugacy class of cocharacters induced by the place v of E, where we recall that $\{\mu\}$ was introduced in Section 2.0.1.

It follows from [KMS 2022, Theorem 1.3.14] that there are locally closed subschemes $\operatorname{Sh}_{G,[b],K^pK_p}$ of $\operatorname{Sh}_{G,K^pK_p}$, called *Newton strata*, indexed by $[b] \in B(G, \{\mu^{-1}\})$, such that

$$\operatorname{Sh}_{G,[b],K^pK_p}(\overline{\mathbb{F}}_p) = \{x \in \operatorname{Sh}_{G,K^pK_p}(\overline{\mathbb{F}}_p) \mid [b_x] = [b]\}$$

and such that

$$\overline{\operatorname{Sh}_{G,[b],K^pK_p}} \subset \bigcup_{[b]' \leq [b]} \operatorname{Sh}_{G,[b'],K^pK_p}.$$

Here we are using the partial order \leq on $B(G, \{\mu^{-1}\})$ defined in [Rapoport and Richartz 1996, Section 2.3].

2.2. Centralisers. Let $x \in \operatorname{Sh}_{G,K_p}(\overline{\mathbb{F}}_p)$ and choose a sufficiently divisible integer m such that the image of x in $\operatorname{Sh}_{G,K^pK_p}(\overline{\mathbb{F}}_p)$ is defined over \mathbb{F}_{q^m} . Then the geometric q^m -Frobenius $\operatorname{Frob}_{q^m}$ acts on \mathbb{V}_x^p via

tensor-preserving automorphisms and therefore determines an element

$$\gamma_{x,m}^p \in G(\mathbb{A}_f^p),$$

which depends on x and m. For $\ell = p$ there is an element $\delta_{x,m} \in G(\mathbb{Q}_{q^m})$, constructed in [KMS 2022, Section 2.1.7], whose class in $B(G_{\mathbb{Q}_p})$ is equal to $[b_x]$. Moreover there is an element $\gamma_{x,m,p} \in G(\mathbb{Q}_{q^m})$ such that

$$\gamma_{x,m,p} = \delta_{x,m} \sigma(\delta_{x,m}) \cdots \sigma^{rm-1}(\delta_{x,m}),$$

where we write $q = p^r$ and where σ denotes the Frobenius on $G(\mathbb{Q}_{q^m})$.

We define $I_{x,\mathbb{A}_f^p} \subset G_{\mathbb{A}_f^p}$ to be the centraliser of $\gamma_{x,m}^p$, which does not depend on m as long as m is sufficiently divisible. We similarly define $I_{x,\ell} \subset G_{\mathbb{Q}_\ell}$ for $\ell \neq p$ to be the centraliser of the projection $\gamma_{x,m,\ell}$ of $\gamma_{x,m}^p$ to $G_{\mathbb{Q}_\ell}$ for sufficiently divisible m.

We define $I_{x,m,p}$ to be the algebraic group over \mathbb{Q}_p whose functor of points is given by

$$R \mapsto \{g \in G(\mathbb{Q}_{q^m} \otimes_{\mathbb{Q}_p} R) \mid g \delta_{x,m} = \delta_{x,m} \sigma(g)\},$$

where σ is induced by $\sigma: G(\mathbb{Q}_{q^m}) \to G(\mathbb{Q}_{q^m})$. As explained in [KMS 2022, Section 2.1.7], the base change $I_{x,m,p} \otimes \mathbb{Q}_{q^m}$ is naturally identified with the centraliser of the semisimple element $\gamma_{x,m,p}$ in $G(\mathbb{Q}_{q^m})$, and $I_{x,m,p}$ is thus reductive. We similarly define $J_{\delta_{x,m}}$ by its functor of points

$$R \mapsto \{g \in G(\check{\mathbb{Q}}_p \otimes_{\mathbb{Q}_p} R) \mid g\delta_{x,m} = \delta_{x,m}\sigma(g)\}.$$

2.2.1. Consider the decomposition

$$G^{\text{ad}} = \prod_{i=1}^{n} G_i \tag{2.2.1}$$

of G^{ad} into simple groups over \mathbb{Q} . Let $\delta_{x,m,i}$ and $\gamma_{x,m,p,i}$ be the images of $\delta_{x,m}$ and $\gamma_{x,m,p}$ in $G_i(\mathbb{Q}_{q^m})$.

Lemma 2.2.2. Let Z_G be the centre of G. There is a product decomposition

$$I_{x,m,p}/Z_{G,\mathbb{Q}_p} = \prod_{i=1}^n I_{x,m,p,i},$$

where $I_{x,m,p,i}$ represents the functor on \mathbb{Q}_p -algebras sending R to

$$\{g \in G_i(\mathbb{Q}_{q^m} \otimes_{\mathbb{Q}_p} R) \mid g\delta_{x,m,i} = \delta_{x,m,i}\sigma(g)\}.$$

Similarly there is a product decomposition

$$J_{\delta_{x,m}}/Z_G\simeq\prod_{i=1}^nJ_{\delta_{x,m,i}},$$

where $J_{\delta_{x,m,i}}$ represents the functor on \mathbb{Q}_p -algebras sending R to

$$\{g \in G_i(\check{\mathbb{Q}}_p \otimes_{\mathbb{Q}_p} R) \mid g\delta_{x,m,i} = \delta_{x,m,i}\sigma(g)\}.$$

Proof. Consider the commutative diagram

Since the kernel of the bottom map is central and the bottom map is surjective, it follows that the natural map $I_{x,p} \to \prod_{i=1}^n I_{x,m,p,i}$ is surjective. The kernel is given by the intersection of $I_{x,m,p}$ with the kernel of the bottom map and thus has the following functor of points:

$$R \mapsto \{g \in Z_G(\mathbb{Q}_{q^m} \otimes_{\mathbb{Q}_p} R) \mid g\delta_{x,m,i} = \delta_{x,m,i}\sigma(g)\}.$$

This forces $g = \sigma(g)$ and so $g \in Z_G(R) \subset Z_G(\mathbb{Q}_{q^m} \otimes_{\mathbb{Q}_p} R)$. The same proof shows that there is a product decomposition $J_{\delta_{x,m}}/Z_G \simeq \prod_{i=1}^n J_{\delta_{x,m,i}}$.

Note that $I_{x,m,p,i} \otimes \mathbb{Q}_{q^m}$ can be identified with the centraliser of $\gamma_{x,m,p,i}$ in $G_{i,\mathbb{Q}_{q^m}}$ as in the beginning of Section 2.2. The centraliser of $\gamma_{x,m,p,i} \in G(\check{\mathbb{Q}}_p)$ does not depend on m for m sufficiently divisible, and thus the group $I_{x,m,p}$ does not depend on m for m sufficiently divisible. We will write $I_{x,p}$ for the group $I_{x,m,p}$ for sufficiently divisible m and similarly $I_{x,p,i}$ for the group $I_{x,m,p,i}$. We will identify $I_{x,p} \otimes \check{\mathbb{Q}}_p$ with the centraliser of $\gamma_{x,m,p,i}$ in $G(\check{\mathbb{Q}}_p)$ for sufficiently divisible m and similarly we will identify $I_{x,p,i}$ with the centraliser of $\gamma_{x,m,p,i}$ in $G(\check{\mathbb{Q}}_p)$.

2.2.3. Let $x \in \operatorname{Sh}_{G,K_p}(\overline{\mathbb{F}}_p)$ and let $\operatorname{Aut}(A_x)$ be the algebraic group over \mathbb{Q} with functor of points

$$\operatorname{Aut}(A_{\scriptscriptstyle X})(R) = (\operatorname{End}(A_{\scriptscriptstyle X}) \otimes_{\mathbb{Z}} R)^{\times}.$$

Following [KMS 2022, Section 2.1.3], we define I_x^p to be the largest closed subgroup of $\operatorname{Aut}(A_x)$ that fixes the tensors $s_{\alpha,\mathbb{A}_f^p,x}$ and $I_x \subset I_x^p$ to be the largest closed subgroup that also fixes the tensors $s_{\alpha,\operatorname{cris},x}$. There are natural maps $I_{x,\mathbb{Q}_\ell} \to I_{x,\ell}$ for all (including $\ell = p$), see [KMS 2022, Section 2.1.8] for the $\ell = p$ case.

The groups $I_{x,\ell}$ are connected reductive subgroups of $G_{\mathbb{Q}_{\ell}}$ and in fact Levi subgroups over $\overline{\mathbb{Q}}_{\ell}$. By [KMS 2022, Corollary 2.1.9] for all ℓ (including $\ell = p$) the natural map

$$I_{x,\mathbb{Q}_{\ell}} \to I_{x,\ell}$$

is an isomorphism. This induces a closed immersion of groups $I_{x,\mathbb{Q}_p} \to J_{\delta_{x,m}}$ for some sufficiently divisible m.

2.3. *An assumption.* We will need to assume the following hypothesis to prove our main monodromy theorems in Section 3.

Hypothesis 2.3.1. For all points $x \in \operatorname{Sh}_{G,K_p}(\overline{\mathbb{F}}_p)$ and for sufficiently divisible m depending on x, there is an element $\gamma_{x,m} \in G(\overline{\mathbb{Q}})$ that is conjugate to $\gamma_{x,m,\ell}$ in $G(\overline{\mathbb{Q}}_\ell)$ for all ℓ (including $\ell = p$). Moreover the $G(\overline{\mathbb{Q}})$ -conjugacy class of $\gamma_{x,m}$ is stable under the action of $\operatorname{Gal}(\overline{\mathbb{Q}}/\mathbb{Q})$.

If the $G(\overline{\mathbb{Q}})$ -conjugacy class of $\gamma_{x,m}$ contains an element of $G(\mathbb{Q})$, then it is clearly Galois stable. However the converse does not necessarily hold.

Lemma 2.3.2. The hypothesis holds when K_p is hyperspecial.

Proof. If K_p is hyperspecial, then [Kisin 2017, Corollary 2.3.1] tells us that there is an element $\gamma_{x,m} \in G(\mathbb{Q})$ that is conjugate to $\gamma_{x,m,\ell}$ in $G(\overline{\mathbb{Q}}_{\ell})$ for all ℓ (including $\ell = p$).

Remark 2.3.3. By [KMS 2022, Corollary 2.2.14], an element $\gamma_{x,m} \in G(\mathbb{Q})$ satisfying the requirements of Hypothesis 2.3.1 exists when $G_{\mathbb{Q}_p}$ is quasi-split and has no factors of type D.

If K_p is very special, the group $G_{\mathbb{Q}_p}$ is tamely ramified and satisfies $p \nmid 2 \cdot \#\pi_1(G^{\operatorname{der}})$ and $\pi_1(G)_I$ is torsion free, where $I \subset \operatorname{Gal}(\overline{\mathbb{Q}}_p/\mathbb{Q}_p)$ is the inertia group, then the existence of an element $\gamma_{x,m} \in G(\mathbb{Q})$ satisfying the requirements of Hypothesis 2.3.1 follows from Theorem I of [van Hoften 2020].

If p > 2, if K_p is a very special parahoric subgroup and if the triple (G, X, K_p) is acceptable in the sense of [Kisin and Zhou 2021, Definitions 5.2.6 and 5.2.9], then [Kisin and Zhou 2021, Theorem 6.1.4] proves the existence of an element $\gamma_{x,m} \in G(\overline{\mathbb{Q}})$ satisfying the requirements of Hypothesis 2.3.1.

Remark 2.3.4. When $G_{\mathbb{Q}_p}$ is not quasi-split, one should probably not expect that the $G(\overline{\mathbb{Q}})$ -conjugacy class of $\gamma_{x,m}$ always contains an element of $G(\mathbb{Q})$. This is because CM lifts do not exist in general when $G_{\mathbb{Q}_p}$ is not quasi-split. However, we expect Hypothesis 2.3.1 to hold in full generality.

For example, let $x \in \operatorname{Sh}_{G,K_p}(\overline{\mathbb{F}}_p)$ be a point corresponding to the good reduction of an abelian variety *defined over a number field* and assume that p > 2. Then [Kisin and Zhou 2021, Theorem 7.2.4] tells us that there is an element $\gamma_{x,m} \in G(\overline{\mathbb{Q}})$ satisfying the requirements of Hypothesis 2.3.1.

2.3.5. We end by deducing a consequence of Hypothesis 2.3.1 that will be used in Section 3. Let G^* denote the quasi-split inner form of G over \mathbb{Q} and let $\Psi: G \otimes \overline{\mathbb{Q}} \to G^* \otimes \overline{\mathbb{Q}}$ be an inner twisting. This means that every $\tau \in \operatorname{Gal}(\overline{\mathbb{Q}}/\mathbb{Q})$ satisfies

$$\Psi(\tau(g)) = h_{\tau}\tau(\Psi(g))h_{\tau}^{-1}$$

for some element $h_{\tau} \in G^*(\overline{\mathbb{Q}})$. A direct consequence of this definition is that the image under ψ of a $Gal(\overline{\mathbb{Q}}/\mathbb{Q})$ -stable $G(\overline{\mathbb{Q}})$ -conjugacy class is a $Gal(\overline{\mathbb{Q}}/\mathbb{Q})$ -stable $G^*(\overline{\mathbb{Q}})$ -conjugacy class.

Lemma 2.3.6. Suppose that Hypothesis 2.3.1 holds and let $\gamma_{x,m} \in G(\overline{\mathbb{Q}})$ be the element that is guaranteed to exist by that Hypothesis. Then for sufficiently divisible m the element $\Psi(\gamma_{x,m})$ is $G^*(\overline{\mathbb{Q}})$ -conjugate to an element in $G^*(\mathbb{Q})$.

Proof. If m is sufficiently divisible, then the centraliser of $\gamma_{x,m}$ is connected because this is true for $\gamma_{x,m,\ell}$ and the formation of centralisers commutes with base change. Since G^* is quasi-split and the element $\Psi(\gamma_{x,m})$ is semisimple with connected centraliser, we may apply [Kottwitz 1982, Theorem 4.7(2)] which tells us that the $G^*(\mathbb{Q})$ -conjugacy class of $\Psi(\gamma_{x,m})$ contains an element of $G^*(\mathbb{Q})$.

3. Monodromy of Hecke-invariant subvarieties

In this section we prove an ℓ -adic monodromy theorem in the style of Chai, see [Chai 2005] and [Xiao 2020; Kasprowitz 2012], for prime-to-p Hecke stable subvarieties of Shimura varieties of Hodge type in characteristic p. We expect the results in this section to be of independent interest, at least beyond the hyperspecial case that we will use in the rest of this article.

In Section 3.1 we establish formal properties of subvarieties Z of Shimura varieties of Hodge type in characteristic p that are stable under prime-to-p Hecke operators. Using techniques from [Chai 2005], we prove that the ℓ -adic monodromy groups of the universal abelian variety over such Z are normal subgroups of $G_{\mathbb{Q}_{\ell}}$, this is stated as Corollary 3.1.16.

In Section 3.2 we use the results from [KMS 2022] in combination with Hypothesis 2.3.1 to prove Theorem 3.2.5 and Corollary 3.2.6; the latter is a generalisation of Theorem II. In Section 3.3 we combine this theorem with results of D'Addezio [2020] to deduce results about the *p*-adic monodromy groups of the universal abelian variety over Hecke stable subvarieties.

Finally, in Section 3.4 we prove results about irreducible components of Hecke stable subvarieties in the style of [Chai 2005, Proposition 4.5.4]. We will not use these results in the rest of this article and so this section can safely be skipped for the reader only interested in the proof of Theorem I.

3.1. Arithmetic monodromy groups I. Let the notation be as in Section 2. In this section we are going to study arithmetic monodromy groups of Hecke stable subvarieties of Sh_{G,K^pK_p} . For maximal generality, we do not assume that these are defined over $k = \mathbb{F}_q$ and so from now on we will implicitly base change the Shimura variety Sh_{G,K^pK_p} to an unspecified finite extension of k, which we will also denote by k.

The morphism $\pi: \operatorname{Sh}_{G,K_p} \to \operatorname{Sh}_{G,K^pK_p}$ is a pro-étale K^p -torsor over $\operatorname{Sh}_{G,K^pK_p}$ such that the action of $K^p \subset G(\mathbb{A}_f^p)$ extends to an action of $G(\mathbb{A}_f^p)$. Let $Z \subset \operatorname{Sh}_{G,K^pK_p}$ be a locally closed subscheme and let \widetilde{Z} be the inverse image of Z under π . We say that Z is *stable under the prime-to-p-Hecke operators*, or that Z is $G(\mathbb{A}_f^p)$ -stable, if \widetilde{Z} is $G(\mathbb{A}_f^p)$ -stable.

For the rest of this section ℓ will be used to denote a prime number not equal to p. For such ℓ we let K_{ℓ} be the image of K^p in $G(\mathbb{Q}_{\ell})$ under the projection $G(\mathbb{A}_f^p) \to G(\mathbb{Q}_{\ell})$. We let

$$\pi_{\ell}: \operatorname{Sh}_{G,K_n} \times^{K^p} K_{\ell} \to \operatorname{Sh}_{G,K^pK_n}$$
 (3.1.1)

be the induced pro-étale K_ℓ -torsor. For $Z \subset \operatorname{Sh}_{G,K^pK_p}$ a locally closed subscheme, we will write Z_ℓ for the inverse image of Z under π_ℓ . We say that Z is *stable under the* ℓ -adic Hecke operators, or that Z is $G(\mathbb{Q}_\ell)$ -stable, if Z_ℓ is $G(\mathbb{Q}_\ell)$ -stable. When discussing $G(\mathbb{Q}_\ell)$ -stable Z we will always implicitly work with $\ell \neq p$. If Z is $G(\mathbb{A}_f^p)$ -stable, then it is automatically $G(\mathbb{Q}_\ell)$ -stable for all $\ell \neq p$.

All the results in this section will be stated for smooth Z, and the following lemma will be used to reduce to the smooth case in the proof of Theorem I.

Lemma 3.1.1. Let $Z \subset \operatorname{Sh}_{G,K^pK_p}$ be a locally closed subscheme that is stable under the action of $G(\mathbb{A}_f^p)$ (respectively $G(\mathbb{Q}_\ell)$), then the smooth locus $U \subset Z$ is also stable under this action.

Proof. For $g \in G(\mathbb{A}_f^p)$ and $K^p \subset G(\mathbb{A}_f^p)$ there is a finite étale correspondence



and the assumption that \widetilde{Z} is stable under the action of g is equivalent to the statement that the inverse image of Z under p_1 is the same as the inverse image of Z under $g \circ p_2$ for all choices of K^p . Because all the maps in the diagram are finite étale, the same is true for the smooth locus U of Z. Therefore the inverse image \widetilde{U} of U under π is stable under the action of $g \in G(\mathbb{A}_f^p)$.

Lemma 3.1.2. Let $Z \subset Y \subset \operatorname{Sh}_{G,K^pK_p}$ be locally closed and $G(\mathbb{A}_f^p)$ -stable (resp. $G(\mathbb{Q}_\ell)$ -stable) subvarieties. Then the closure of Z in Y is also stable under $G(\mathbb{A}_f^p)$ (resp. $G(\mathbb{Q}_\ell)$).

Proof. This follows in the same way as in the proof of Lemma 3.1.1 from the fact that the prime-to-p Hecke correspondences are finite étale; indeed finite étale maps are open and closed, and thus take closures to closures.

3.1.3. Some general topology. Let $\{X_i\}_{i\in I}$ be a countably indexed cofiltered inverse system of finite type schemes over a field k with surjective affine transition maps. Let $X = \varprojlim_i X_i$ be the inverse limit, it is a nonempty quasi-compact scheme by [Stacks 2020, Lemma 01Z2]. Recall that for a quasi-compact scheme Y there is a profinite topological space $\pi_0(Y)$ of connected components of Y.

Lemma 3.1.4. The natural map

$$\pi_0(X) \to \varprojlim_i \pi_0(X_i)$$
(3.1.2)

is a homeomorphism.

Proof. The left hand side of (3.1.2) is a profinite topological space by [Stacks 2020, Lemma 0906] and the right hand side of (3.1.2) is visibly an inverse limit of finite sets. Hence both sides are compact Hausdorff topological spaces and to show that the map is a homeomorphism it suffices to show that it is a bijection.

To show that the natural map is a bijection, we construct an explicit inverse. Any compatible system of connected components $\{V_i\}_{i\in I}$ of $\{X_i\}_{i\in I}$ has nonempty and quasi-compact inverse limit $V\subset X$ by [Stacks 2020, Lemma 0A2W]. To prove that V is connected we suppose that there are nonempty open and closed subsets W and W' of V such that $V=W\coprod W'$. Then W and W' are quasi-compact open because V is quasi-compact.

Now [Stacks 2020, Lemma 0A30.(1)] tells us that we can find i and (nonempty) constructible quasicompact open subsets Z, Z' of V_i such that W is the inverse image of Z under $V \to V_i$ and similarly W'is the inverse image of Z' under $V \to V_i$. In particular, the subsets Z and Z' are disjoint nonempty open subsets of V_i , which gives us a contradiction since V_i is connected.

We have produced a map $\varprojlim_i \pi_0(X_i) \to \pi_0(X)$ and it is not hard to check that it is an inverse of the natural map from the lemma; this concludes the proof.

Corollary 3.1.5. Let $Z \subset \operatorname{Sh}_{G,K^pK_p}$ be a $G(\mathbb{A}_f^p)$ -stable (resp. $G(\mathbb{Q}_\ell)$ -stable) locally closed subscheme of $\operatorname{Sh}_{G,K^pK_p}$ and let \widetilde{Z} be as above. Then $\pi_0(\widetilde{Z})$ is equipped with a continuous action of $G(\mathbb{A}_f^p)$ (respectively $\pi_0(Z_\ell)$ is equipped with a continuous action of $G(\mathbb{Q}_\ell)$).

Proof. The existence of the action follows from the existence of the action on \widetilde{Z} (resp. Z_{ℓ}). The continuity follows from the continuity of the action of K^p on $\varprojlim_{K^p} Z_{K^p}$ (resp. the continuity of the action of K_{ℓ} on Z_{ℓ}) and Lemma 3.1.4.

The following lemma is only a slight generalisation of [Chai 2005, Lemma 2.8], but we include a proof for the benefit of the reader.

Lemma 3.1.6. Let X be a second-countable compact Hausdorff topological space with a transitive and continuous action of a locally profinite topological group \mathbb{G} . Let $x \in X$ with stabiliser $\mathbb{G}_x \subset \mathbb{G}$, then the orbit map

$$O: \mathbb{G}/\mathbb{G}_r \to X$$

is a homeomorphism.

Proof. We can write \mathbb{G} as the increasing union of countably many compact open sets, for example by using finite unions of cosets of a compact open subgroup $\mathbb{K} \subset \mathbb{G}$. Since the quotient map $\mathbb{G} \to \mathbb{G}/\mathbb{G}_x$ is open for any topological group, it follows that \mathbb{G}/\mathbb{G}_x can be written as the increasing union of countably many compact open subsets.

Since the orbit map is surjective, the topological space X can be written as a countable union of the compact subsets O(U) for $U \subset \mathbb{G}/\mathbb{G}_x$ compact open. Because X is second-countable it is metrisable by Urysohn's metrisation theorem and thus the Baire category theorem tells us that there exists a compact open subset U of \mathbb{G}/\mathbb{G}_x such that O(U) contains an open subset W of X.

Choose a compact open subset $V \subset U$ such that $O(V) \subset W$. Then $O: V \to O(V)$ is a continuous bijection between compact Hausdorff topological spaces and hence a homeomorphism. Now note that \mathbb{G} acts transitively on both \mathbb{G}/\mathbb{G}_x and on X. Hence by moving around V we see that any point of $V \in \mathbb{G}/\mathbb{G}_x$ has an open neighbourhood V_V such that the natural map $V: V_V \to O(V_V)$ is a homeomorphism, and we conclude that $V: V_V \to O(V_V)$ is a homeomorphism. $V: V_V \to O(V_V)$

3.1.7. Lie groups over ℓ -adic local fields. Recall that a topological group M is called an ℓ -adic Lie group if it admits the (necessarily unique) structure of an ℓ -adic Lie group; see [Glockner 2016, Definition 2.1, Proposition 2.2]. If M is an ℓ -adic Lie group, then by definition there is a finite-dimensional \mathbb{Q}_{ℓ} -Lie algebra Lie M, an open neighbourhood $U \subset \operatorname{Lie} M$ of the identity and an exponential map $\operatorname{Exp}: U \to M$ that is a homeomorphism onto a compact open subgroup of M. For example for an algebraic group H over \mathbb{Q}_{ℓ} the topological group $H(\mathbb{Q}_{\ell})$ is an ℓ -adic Lie group with Lie algebra $\operatorname{Lie} H(\mathbb{Q}_{\ell}) = \operatorname{Lie} H$.

Lemma 3.1.8. Let H be an algebraic group over \mathbb{Q}_{ℓ} and let $M \subset H(\mathbb{Q}_{\ell})$ be a subgroup that is compact in the subspace topology. Then M is an ℓ -adic Lie group and the morphism $M \to H(\mathbb{Q}_{\ell})$ is a morphism of ℓ -adic Lie groups. Moreover, the induced Lie subalgebra Lie $M \subset \text{Lie } H(\mathbb{Q}_{\ell}) = \text{Lie } H$ satisfies

[Lie
$$M$$
, Lie M] = [Lie \mathbb{M} , Lie \mathbb{M}],

where the bracket notation means the commutator of two Lie subalgebras and where \mathbb{M} is the Zariski closure of M.

Proof. The group M is an ℓ -adic Lie group by [Glockner 2016, Proposition 2.3] and the morphism $M \to H(\mathbb{Q}_{\ell})$ is a morphism of ℓ -adic Lie groups by [Glockner 2016, Proposition 2.2]. This implies that there is an induced morphism on Lie algebras Lie $M \to \text{Lie } H(\mathbb{Q}_{\ell}) = \text{Lie } H$.

Since $M \subset \mathbb{M}(\mathbb{Q}_{\ell})$ is Zariski dense, it follows that the smallest algebraic subgroup of H whose Lie algebra contains Lie M is equal to \mathbb{M} ; indeed, if there is a smaller algebraic subgroup $\mathbb{M}' \subset \mathbb{M}$ with Lie $M \subset \operatorname{Lie} \mathbb{M}'$, then we see using the ℓ -adic exponential map that there is a compact open (hence finite index) subgroup of M contained in $\mathbb{M}'(\mathbb{Q}_{\ell})$. This contradicts the fact that M is Zariski dense in \mathbb{M} .

The fact that the smallest algebraic subgroup of H whose Lie algebra contains Lie M is equal to \mathbb{M} is expressed as $\mathfrak{a}(\text{Lie }M) = \text{Lie }\mathbb{M}$ in the notation of [Borel 1991, Section 7.1]. By [Borel 1991, Corollary 7.9] we have the following equality of Lie subalgebras of Lie H

[Lie
$$M$$
, Lie M] = [$\mathfrak{a}(\text{Lie }M)$, $\mathfrak{a}(\text{Lie }M)$] = [Lie \mathbb{M} , Lie \mathbb{M}].

Lemma 3.1.9. Let \mathbb{M} be a semisimple algebraic group over \mathbb{Q}_{ℓ} and let $M \subset \mathbb{M}(\mathbb{Q}_{\ell})$ be a subgroup closed in the ℓ -adic topology. If M equipped with the subspace topology is compact and M is Zariski dense in \mathbb{M} , then M is a compact open subgroup of $\mathbb{M}(\mathbb{Q}_{\ell})$.

Proof. It follows from Lemma 3.1.8 that M is an ℓ -adic Lie group, that $M \to \mathbb{M}(\mathbb{Q}_{\ell})$ is a morphism of ℓ -adic Lie groups and that the Lie algebra of M is equal to the Lie algebra of \mathbb{M} , since \mathbb{M} is semisimple. Now we can use the exponential map for ℓ -adic Lie groups to show that M contains a compact open subgroup of $\mathbb{M}(\mathbb{Q}_{\ell})$. Since M is itself compact, this implies that M is also a compact open subgroup of $\mathbb{M}(\mathbb{Q}_{\ell})$. \square

3.1.10. The main theorem of Galois theory for schemes tells us that the category of finite-étale covers of a smooth connected scheme Z over k is equivalent to the category of finite sets equipped with a continuous action of $\pi_1^{\text{\'et}}(Z,z)$. Under this equivalence, a finite étale cover $f:Y\to Z$ is sent to the finite set $f^{-1}(z)$ equipped with its action of $\pi_1^{\text{\'et}}(Z,z)$. In particular, the set of connected components of Y is in bijection with the set of orbits of $\pi_1^{\text{\'et}}(Z,z)$ on $f^{-1}(z)$.

If $f: Y \to Z$ is a countably indexed inverse limit of finite étale covers $f_i: Y_i \to Z$ with surjective transition maps, then we can associate to f the profinite set

$$f^{-1}(z) = \varprojlim_{i} f_{i}^{-1}(z),$$

equipped with its natural continuous action of $\pi_1^{\text{\'et}}(Z, z)$. By Lemma 3.1.4 it follows that the profinite set of orbits of $\pi_1^{\text{\'et}}(Z, z)$ on $f^{-1}(z)$ is homeomorphic to the topological space of connected components of Y.

3.1.11. Now let Z be a smooth $G(\mathbb{Q}_{\ell})$ -stable locally closed subscheme of $\operatorname{Sh}_{G,K^pK_p}$, let $Z^{\circ} \subset Z$ be a connected component of Z, and let $z \in Z^{\circ}(\overline{\mathbb{F}}_p)$. Let π_{ℓ} be as in (3.1.1) and write Z_{ℓ} for the inverse image

of Z under π_{ℓ} as above; it is stable under the action of $G(\mathbb{Q}_{\ell})$ by assumption. Denote by Z_{ℓ}° the inverse image of Z° under π_{ℓ} , then $Z_{\ell}^{\circ} \to Z^{\circ}$ is a profinite étale K_{ℓ} -torsor.

By the Galois theory for schemes discussed above, the cover $\pi_{\ell}: Z_{\ell}^{\circ} \to Z^{\circ}$ corresponds to the profinite set $\pi_{\ell}^{-1}(z)$ equipped with its natural action of $\pi_{1}^{\text{\'et}}(Z^{\circ}, z)$. In particular, the set of connected components of Z_{ℓ}° corresponds to the set of orbits of $\pi_{1}^{\text{\'et}}(Z^{\circ}, z)$ on $\pi_{\ell}^{-1}(z)$.

Choose an element $\tilde{z} \in \pi_{\ell}^{-1}(z)$. Then using the simply transitive action of K_{ℓ} on $\pi_{\ell}^{-1}(z)$ we can identify $\pi_{\ell}^{-1}(z)$ with K_{ℓ} ; under this identification the chosen element \tilde{z} is send to $1 \in K_{\ell}$. This defines a continuous group homomorphism

$$\rho_{\ell}: \pi_1^{\text{\'et}}(Z^{\circ}, z) \to K_{\ell},$$

whose conjugacy class does not depend on the choice of \tilde{z} . Let $y \in \pi_0(Z_\ell^\circ)$ be the connected component containing \tilde{z} . Then the stabiliser of y in K_ℓ is equal to the image of ρ_ℓ .

Let $P_y \subset G(\mathbb{Q}_\ell)$ be the stabiliser of y in $G(\mathbb{Q}_\ell)$. It is a closed topological subgroup by the continuity of the action and the fact that $\pi_0(Z_\ell)$ is Hausdorff. Its intersection with K_ℓ gives us the stabiliser of y in K_ℓ . The action map gives us a continuous map

$$G(\mathbb{Q}_{\ell})/P_{\nu} \to \pi_0(Z_{\ell}), \quad g \mapsto g \cdot y$$

with image the orbit Orb(y) of y.

Lemma 3.1.12. The orbit Orb(y) is open and closed inside of $\pi_0(Z_\ell)$. Moreover the orbit map induces a homeomorphism $G(\mathbb{Q}_\ell)/P_y \cong Orb(y)$; in particular, $G(\mathbb{Q}_\ell)/P_y$ is compact.

Proof. The identification

$$\pi_0(Z_\ell)/K_\ell \simeq \pi_0(Z)$$
.

tells us that there are finitely many K_{ℓ} -orbits on $\pi_0(Z_{\ell})$, and that each of them is open and closed. The $G(\mathbb{Q}_{\ell})$ -orbit of a point y is then a union of finitely many K_{ℓ} -orbits, and thus also open and closed. Lemma 3.1.12 shows that Orb(y) is open and closed inside a second-countable profinite topological space. Therefore Orb(y) is profinite and second-countable. The result now follows from Lemma 3.1.6.

Let M be the image of ρ_{ℓ} and let \mathbb{M} be the neutral component of its Zariski closure inside $G(\mathbb{Q}_{\ell})$. Let $\rho_{\ell,\text{geom}}$ be the restriction of ρ_{ℓ} to

$$\pi_1^{\text{\'et}}(Z_{\overline{\mathbb{F}}_n}^{\circ}, z) \subset \pi_1^{\text{\'et}}(Z^{\circ}, z),$$

let M_{geom} be its image and let \mathbb{M}_{geom} be the neutral component of its Zariski closure inside $G(\mathbb{Q}_{\ell})$.

Lemma 3.1.13. The groups \mathbb{M} and \mathbb{M}_{geom} are connected reductive groups over \mathbb{Q}_{ℓ} .

Proof. There is a short exact sequence (e.g., by [D'Addezio 2020, Proposition 3.2.7])

$$1 \to \mathbb{M}'_{\text{geom}} \to \mathbb{M} \to Q \to 1, \tag{3.1.3}$$

where Q is a commutative algebraic group of multiplicative type and where \mathbb{M}'_{geom} is a closed subgroup of \mathbb{M} with neutral component given by \mathbb{M}_{geom} . In particular, it follows that \mathbb{M} is reductive if \mathbb{M}_{geom} is

reductive. The representation

$$\pi_1^{\text{\'et}}(Z^{\circ}, z) \to K_{\ell} \to G(\mathbb{Q}_{\ell}) \to GL(V)(\mathbb{Q}_{\ell})$$

is the monodromy representation of the (rational) ℓ -adic Tate module of the abelian scheme $\pi: A \to \operatorname{Sh}_{G,K^pK_p}$ coming from the Hodge embedding $\operatorname{Sh}_{G,K^pK_p} \to \operatorname{Sh}_{\mathcal{G}_V,\mathcal{K}^p\mathcal{K}_p}$. This is an ℓ -adic sheaf \mathcal{F}_0 on Z° which is pure of weight one. Then [Deligne 1980, Theorem 3.4.1.(iii)] tells us that the basechange \mathcal{F} of \mathcal{F}_0 to $Z^\circ_{\mathbb{F}_p}$ is semisimple. This base change corresponds to the composition of $\rho_{\ell,\text{geom}}$ with $K_\ell \to G(\mathbb{Q}_\ell) \to \operatorname{GL}(V)(\mathbb{Q}_\ell)$. Now [Deligne 1980, Corollary 1.3.9] tells us that \mathbb{M}_{geom} is a semisimple algebraic group, and thus that it is reductive.

3.1.14. Let \mathbb{N} be the normaliser of \mathbb{M} in $G_{\mathbb{Q}_{\ell}}$ and let \mathbb{N}° be its neutral component. The group \mathbb{N}° is a connected reductive group because we are working with reductive groups in characteristic zero; see, e.g., [Conrad et al. 2015, Proposition A.8.12].

Lemma 3.1.15. *The group* P_v *is contained in* \mathbb{N} .

Proof. Let $\gamma \in P_y$, then we want to show that γ normalises \mathbb{M} . If V is a compact open subgroup of $G(\mathbb{Q}_\ell)$ contained in K_ℓ , then $V \cap P_y \subset M$. Moreover for every $\gamma \in P_y$ we can find an open subgroup $U \subset G(\mathbb{Q}_\ell)$ such that $\gamma U \gamma^{-1} \subset V$. For example, we can just take the intersection of V with $\gamma V \gamma^{-1}$. For such U the open subgroup $M \cap U$ of M satisfies $\gamma (M \cap U) \gamma^{-1} \subset M$.

Since conjugation by γ is a homeomorphism in the Zariski topology, we see that the Zariski closure of $M \cap U$ is moved under conjugation by γ into the Zariski closure of M. But since $M \cap U$ is an open subgroup of M it is also a closed subgroup and thus compact and thus of finite index in M. This means that the Zariski closure of $M \cap U$ and the Zariski closure of M have the same identity component, both of which are equal to M. Since conjugation preserves 1, this mean it sends M to M.

Corollary 3.1.16. The Zariski closure \mathbb{M} of M is a normal subgroup of $G_{\mathbb{Q}_{\ell}}$.

Proof. The group $G(\mathbb{Q}_{\ell})/\mathbb{N}(\mathbb{Q}_{\ell})$ is compact, because it is a quotient of $G(\mathbb{Q}_{\ell})/P_y$ which is compact. Since $\mathbb{N}^{\circ}(\mathbb{Q}_{\ell})$ is finite index in $\mathbb{N}(\mathbb{Q}_{\ell})$, it follows that $G(\mathbb{Q}_{\ell})/\mathbb{N}^{\circ}(\mathbb{Q}_{\ell})$ is also compact. Since \mathbb{N}° is connected it follows from [Borel and Tits 1965, Propositions 8.4 and 9.3] that it contains a parabolic subgroup of $G_{\mathbb{Q}_{\ell}}$ and because it is reductive it follows that $\mathbb{N}^{\circ} = G_{\mathbb{Q}_{\ell}}$. Therefore $\mathbb{N}^{\circ} = \mathbb{N} = G_{\mathbb{Q}_{\ell}}$ and we find that \mathbb{M} is a normal subgroup of $G_{\mathbb{Q}_{\ell}}$.

3.2. Arithmetic monodromy groups II. So far we have not excluded the possibility that \mathbb{M} is contained in the centre of $G_{\mathbb{Q}_{\ell}}$. In fact, this happens when Z is the supersingular locus inside the modular curve. Thus we will need additional assumptions on Z to prove that \mathbb{M} is not central.

We will show, using the results of [KMS 2022], that if Z contains a point $x \in Z(\mathbb{F}_{q^m})$ not contained in the smallest Newton stratum, then the image of Frob_x under ρ_ℓ is noncentral. If $G_{\mathbb{Q}_\ell}^{\operatorname{ad}}$ were a simple group over \mathbb{Q}_ℓ , then this would force \mathbb{M} to contain $G_{\mathbb{Q}_\ell}^{\operatorname{der}}$. However G^{ad} is generally not a simple group over \mathbb{Q} , and even if it were simple then there would generally be no primes ℓ where $G_{\mathbb{Q}_\ell}^{\operatorname{ad}}$ is simple. To deal with these issues, we will make use of Hypothesis 2.3.1.

3.2.1. Recall that for a point $x \in Z^{\circ}(\mathbb{F}_{q^m})$, there is a Frobenius element

$$\operatorname{Frob}_{x} \in \pi_{1}^{\operatorname{\acute{e}t}}(Z^{\circ}, z),$$

whose image under ρ_{ℓ} is the element $\gamma_{x,m,\ell} \in G(\mathbb{Q}_{\ell})$ from Section 2.2. If m is sufficiently divisible, then its centraliser is equal to the group $I_{x,\ell}$.

3.2.2. The decomposition $G^{ad} = \prod_{i=1}^{n} G_i$ of (2.2.1) induces maps

$$B(G_{\mathbb{Q}_p}) \to B(G_{\mathbb{Q}_p}^{\mathrm{ad}}) \to \prod_{i=1}^n B(G_{i,\mathbb{Q}_p}).$$

For an element $[b] \in B(G_{\mathbb{Q}_p})$ we will write $[b_i]$ for its image in $B(G_{i,\mathbb{Q}_p})$ under this map. Recall, see[Kret and Shin 2021, Definition 5.3.2], that an element $[b] \in B(G, \{\mu^{-1}\})$ is called \mathbb{Q} -nonbasic if $[b_i]$ is nonbasic for all i. A Newton stratum $\mathrm{Sh}_{G,[b],K^pK_p}$ is called \mathbb{Q} -nonbasic if [b] is \mathbb{Q} -nonbasic.

Proposition 3.2.3. Let $x \in Z^{\circ}(\mathbb{F}_{q^m})$ for some sufficiently divisible m and let $[b] = [b_x] \in B(G, \{\mu^{-1}\})$. Assume that Hypothesis 2.3.1 holds. If $[b_i]$ is nonbasic, then the image of $\rho_{\ell}(\operatorname{Frob}_{x,m})$ under

$$G(\mathbb{Q}_{\ell}) \to G^{\mathrm{ad}}(\mathbb{Q}_{\ell}) \to G_i(\mathbb{Q}_{\ell})$$

is nontrivial. Moreover, the image of $\rho_{\ell}(\operatorname{Frob}_{x,m})$ in each simple factor of $G_{i,\mathbb{Q}_{\ell}}$ over \mathbb{Q}_{ℓ} is nontrivial.

Proof. Let m be sufficiently divisible and let $\gamma_{x,m} \in G(\overline{\mathbb{Q}})$ be the element guaranteed to exist by Hypothesis 2.3.1. Let G^* denote the quasi-split inner form of G over \mathbb{Q} and let $\Psi: G \otimes \overline{\mathbb{Q}} \to G^* \otimes \overline{\mathbb{Q}}$ be an inner twisting. Then by Lemma 2.3.6 there is an element $\gamma'_{x,m} \in G^*(\mathbb{Q})$ that is conjugate to $\Psi(\gamma_{x,m})$ in $G^*(\overline{\mathbb{Q}})$. We will write $I'_x \subset G^*$ for the centraliser of $\gamma'_{x,m}$. Recall that by Hypothesis 2.3.1 for all ℓ (including $\ell = p$) the element $\Psi^{-1}(\gamma_{x,m})$ is conjugate to $\gamma_{x,m,\ell}$ in $G(\overline{\mathbb{Q}}_{\ell})$.

By the classification of adjoint algebraic groups we can find number fields² F_1, \ldots, F_n and absolutely simple adjoint algebraic groups H_i over F_i for each $i = 1, \ldots, n$ such that

$$G^{\operatorname{ad}} = \prod_{i=1}^{n} \operatorname{Res}_{F_i/\mathbb{Q}} H_i = \prod_{i=1}^{n} G_i.$$

We have a similar decomposition for $G^{*,ad}$ with H_i replaced by its quasi-split inner form H_i^* and we will write G_i^* for the restriction of scalars from F_i to $\mathbb Q$ of H_i^* .

Let $\gamma'_{x,m,i} \in G_i^*(\mathbb{Q}) = H_i^*(F_i)$ be the image of $\gamma'_{x,m}$ and let $C_{x,i} \subset H_i^*$ be its centraliser. Then there is a product decomposition

$$I'_x/Z_G \simeq \prod_{i=1}^n \operatorname{Res}_{F_i/\mathbb{Q}} C_{x,i} = \prod_{i=1}^n I'_{x,i}.$$

Let $b = b_x \in G(\tilde{\mathbb{Q}}_p)$ for sufficiently divisible m be as in Section 2.1.2. Then Equation (2.2.1) shows that there is a product decomposition $J_b/Z_{G,\mathbb{Q}_p} \simeq \prod_{i=1}^n J_{b_i}$, where each $J_{b,i}$ is the twisted centraliser

²By the classification of Shimura varieties of abelian type in [Milne 2005, Appendix B], each F_i is totally real.

of the image b_i of b in $G_i(\tilde{\mathbb{Q}}_p)$. Moreover the natural inclusion $I_{x,p} \to J_b$ induces closed immersions $I_{x,p,i} \to J_{b_i}$. As in [KMS 2022, Section 1.1.4], there is an inclusion $J_{b,i,\tilde{\mathbb{Q}}_p} \to G_{i,\tilde{\mathbb{Q}}_p}$, identifying its image with the centraliser $M_{v_{b_i}}$ of the fractional cocharacter v_{b_i} of $G_i \otimes \tilde{\mathbb{Q}}_p$ attached to b_i . If $[b_i]$ is not basic, then v_{b_i} is not central and so $Dim J_{b,i} = Dim M_{v_{b_i}} < Dim G_i$ and so $Dim I_{x,p,i} < Dim G_i$.

The subgroup

$$\Psi^{-1}(I'_{x,\overline{\mathbb{Q}}_p})\subset G_{\overline{\mathbb{Q}}_p}$$

can be identified with the centraliser of $\gamma_{x,m,p}$ for m sufficiently divisible, and it follows that

$$\Psi^{-1}(I'_{x,i,\overline{\mathbb{Q}}_p}) \subset G_{i,\overline{\mathbb{Q}}_p}$$

can be identified with the centraliser of the image of $\gamma_{x,m,p,i}$ in $G_{i,\overline{\mathbb{Q}}_p}$. In particular, the group $I'_{x,i,\overline{\mathbb{Q}}_p}$ is conjugate to $I_{x,i,p,\overline{\mathbb{Q}}_p}$ and therefore of the same dimension.

The upshot of the above discussion is that $\operatorname{Dim} I'_{x,i} < \operatorname{Dim} G_i$ if b_i is not basic. It follows that the inclusion $I_{x,i,\mathbb{Q}_\ell} \subset G_{i,\mathbb{Q}_\ell}$ is not an equality for $\ell \neq p$ and thus that the image of $\rho_\ell(\operatorname{Frob}_{x,m})$ in $G_i(\mathbb{Q}_\ell)$ is nontrivial.

To deduce the last statement of Proposition 3.2.3, we note that it suffices to show that the image of $\gamma_{x,m,i}$ in every simple factor of $G_{i,\mathbb{Q}_{\ell}}$ is noncentral. For this, we fix i and a prime number ℓ .

Then we can write $G_{i,\mathbb{Q}_{\ell}}^*$ as a product indexed by primes \mathfrak{p} of F_i dividing ℓ

$$G_{i,\mathbb{Q}_{\ell}}^{*} = \prod_{\mathfrak{p}|\ell} \operatorname{Res}_{F_{i,\mathfrak{p}}/\mathbb{Q}_{\ell}} H_{i,F_{i,\mathfrak{p}}}^{*}.$$

The element $\gamma_{x,m,i}$ is noncentral in $H_i^*(F_i)$ and thus also noncentral in $H_i^*(F_{i,p})$ for all primes \mathfrak{p} of F_i dividing ℓ , and thus we are done.

Remark 3.2.4. When b_i is basic then the inclusion $I_{x,i,\mathbb{Q}_p} \subset J_{b,i}$ should be an equality and the image of $\rho_{\ell}(\operatorname{Frob}_{x,m})$ in $G_i(\mathbb{Q}_{\ell})$ should be trivial for all $\ell \neq p$. This is true when K_p is a very special parahoric subgroup, see the proof of [He et al. 2021, Proposition 5.2.10].

We now state and prove our main arithmetic monodromy theorem.

Theorem 3.2.5. Let $Z \subset \operatorname{Sh}_{G,K^pK_p}$ be a smooth $G(\mathbb{Q}_\ell)$ -stable locally closed subvariety. Let $Z^{\circ} \subset Z$ be a connected component and choose a point $z \in Z^{\circ}(\overline{\mathbb{F}}_p)$. Let \mathbb{M} be the neutral component of the Zariski closure of the image of

$$\rho_{\ell}: \pi_1^{\acute{e}t}(Z^{\circ}, z) \to K_{\ell} \to G(\mathbb{Q}_{\ell}).$$

Assume that Hypothesis 2.3.1 holds. Then \mathbb{M} is a normal subgroup of $G_{\mathbb{Q}_{\ell}}$ surjecting onto $G_{i,\mathbb{Q}_{\ell}}$ for all i such that there is a point $x \in Z^{\circ}(\overline{\mathbb{F}}_{p})$ with $b_{x,i}$ nonbasic.

When $(G, X) = (\mathcal{G}_V, \mathcal{H}_V)$, then this is closely related to [Chai 2005, Corollary 3.5].

Proof. Corollary 3.1.16 proves that $\mathbb{M} \subset G_{\mathbb{Q}_{\ell}}$ is a normal subgroup. For $x \in Z^{\circ}(\overline{\mathbb{F}}_p)$ we have the Frobenius element $\rho_{\ell}(\operatorname{Frob}_{x,m}) = \gamma_{x,m,\ell} \in M$, for all sufficiently divisible m, where we recall that M is the image

of ρ_{ℓ} . Thus $\gamma_{x,m,\ell}$ is contained in the Zariski closure of M and thus after replacing m by a power we may assume that $\gamma_{x,m,\ell} \in \mathbb{M}$, the neutral component of the Zariski closure of M.

If $x \in Z^{\circ}(\overline{\mathbb{F}}_p)$ is a point with $b_{x,i}$ nonbasic, then Proposition 3.2.3 tells us that the image of $\gamma_{x,m,\ell}$ in $G_{i,\mathbb{Q}_{\ell}}$ is nonzero and thus the image of \mathbb{M} in $G_{i,\mathbb{Q}_{\ell}}$ is nonzero. This image is moreover normal, so to show that it is equal to $G_{i,\mathbb{Q}_{\ell}}$ it suffices to show that it maps nontrivially to every simple factor of $G_{i,\mathbb{Q}_{\ell}}$ over \mathbb{Q}_{ℓ} . But this follows from the last line of the statement of Proposition 3.2.3.

Corollary 3.2.6. Let $Z \subset \operatorname{Sh}_{G,K^pK_p,\overline{\mathbb{F}}_p}$ be a smooth $G(\mathbb{Q}_\ell)$ -stable locally closed subscheme as before, let $Z^{\circ} \subset Z$ be a connected component and fix $z \in Z^{\circ}(\overline{\mathbb{F}}_p)$. Let \mathbb{M}_{geom} be the neutral component of the Zariski closure of the image of

$$\pi_1^{\acute{e}t}(Z_{\overline{\mathbb{F}}_n}^{\circ},z) \to K_{\ell} \to G(\mathbb{Q}_{\ell}).$$

Assume that Hypothesis 2.3.1 holds. Then \mathbb{M}_{geom} is a normal subgroup of $G_{\mathbb{Q}_{\ell}}$ surjecting onto $G_{i,\mathbb{Q}_{\ell}}$ for all i such that there is a point $x \in Z^{\circ}(\overline{\mathbb{F}}_p)$ with $b_{x,i}$ nonbasic. If we can find such a point for all i, then $\mathbb{M}_{geom} = G_{\mathbb{Q}_{\ell}}^{der}$.

Proof. The subscheme Z is defined over a finite extension of k, and so we can speak of its arithmetic monodromy group \mathbb{M} . Theorem 3.2.5 tells us that \mathbb{M} surjects onto $G_{i,\mathbb{Q}_{\ell}}$ for all i such that there is a point $x \in Z^{\circ}(\overline{\mathbb{F}}_p)$ with $b_{x,i}$ nonbasic. We now claim that \mathbb{M}_{geom} and \mathbb{M} have the same image in $G_{i,\mathbb{Q}_{\ell}}$ for all such i.

It follows from the short exact sequence (3.1.3) that $\mathbb{M}_{\text{geom}} \subset \mathbb{M}$ is a normal subgroup with abelian cokernel. Let $\mathbb{M}_{\text{geom},i}$ be the image of \mathbb{M}_{geom} in $G_{i,\mathbb{Q}_{\ell}}$ and let \mathbb{M}_{i} be the image of \mathbb{M} in $G_{i,\mathbb{Q}_{\ell}}$. Then $\mathbb{M}_{\text{geom},i} \subset \mathbb{M}_{i}$ is a normal subgroup with abelian cokernel. Given an integer i with $1 \le i \le n$ such that there is a point $x \in Z^{\circ}(\overline{\mathbb{F}}_{p})$ with $b_{x,i}$ nonbasic, then $\mathbb{M}_{i} = G_{i,\mathbb{Q}_{\ell}}$ and therefore \mathbb{M}_{i} has no nontrivial abelian quotients. Thus it follows that the inclusion $\mathbb{M}_{\text{geom},i} \subset \mathbb{M}_{i}$ is an equality.

If we can find a point x with $b_{x,i}$ nonbasic for all i, then \mathbb{M}_{geom} surjects onto $G_{\mathbb{Q}_{\ell}}^{ad}$ and is moreover semisimple by [Deligne 1980, Corollary 1.3.7]. It must therefore be equal to $G_{\mathbb{Q}_{\ell}}^{der}$.

3.3. *p-adic monodromy groups.* In this subsection we record a consequence of Theorem 3.2.5 in combination with the main results of [D'Addezio 2020; 2023].

Recall the following notions from [D'Addezio 2020, Section 2.2]. Write F-Isoc(S) for the \mathbb{Q}_p -linear Tannakian category of F-isocrystals over a smooth finite type scheme S over $\overline{\mathbb{F}}_p$ and write F-Isoc $^{\dagger}(S)$ for the \mathbb{Q}_p -linear Tannakian category of overconvergent F-isocrystals over S. There is a natural fully faithful embedding F-Isoc $^{\dagger}(S) \subset F$ -Isoc(S) which sends an overconvergent F-isocrystal \mathcal{M}^{\dagger} to the underlying F-isocrystal \mathcal{M} . Similarly we write Isoc $^{\dagger}(S)$ and Isoc(S) for the \mathbb{Q}_p -linear category of (overconvergent) isocrystals over S. There are natural faithful forgetful functors from (overconvergent) F-isocrystals to (overconvergent) isocrystals.

3.3.1. The morphism $\mathrm{Sh}_{G,K^pK_p} \to \mathrm{Sh}_{\mathcal{G}_V,\mathcal{K}^p\mathcal{K}_p}$ gives us an abelian scheme $\pi:\mathcal{A}\to\mathrm{Sh}_{G,K^pK_p}$ and we consider the *F*-isocrystal

$$\mathcal{M} := R^1 \pi_* \mathcal{O}_{\mathrm{cris},A}[1/p],$$

which is overconvergent by [Étesse 2002, Theorem 7]. Then [KMS 2022, Corollary 1.3.13] proves that there is an exact \mathbb{Q}_p -linear tensor functor (the *p-adic realisation functor*)

$$\operatorname{Rel}_{p} : \operatorname{Rep}_{\mathbb{Q}_{p}} G \to F\operatorname{-Isoc}(\operatorname{Sh}_{G,K^{p}K_{p}})$$
(3.3.1)

such that the representation $G_{\mathbb{Q}_p} \to \mathcal{G}_V \to GL_V$ coming from the choice of Hodge embedding is sent to the *F*-isocrystal \mathcal{M} .

Lemma 3.3.2. This morphism factors via an exact \mathbb{Q}_p -linear tensor functor

$$\operatorname{Rel}_p : \operatorname{Rep}_{\mathbb{Q}_p} G \to F\operatorname{-Isoc}^{\dagger}(\operatorname{Sh}_{G,K^pK_p}),$$
 (3.3.2)

which we will also denote by Rel_p.

Proof. Since F-Isoc $^{\dagger}(\operatorname{Sh}_{G,K^pK_p}) \subset F$ -Isoc $(\operatorname{Sh}_{G,K^pK_p})$ is a full subcategory, it suffices to show that $\operatorname{Rel}_p(W)$ is overconvergent for each representation W of $G_{\mathbb{Q}_p}$. We follow the proof of [KMS 2022, Corollary 1.3.13]. As explained there, each W can be written as the kernel of a map $e: V_{m,n} \to V_{m',n'}$, where

$$V_{m,n} = V^{\otimes m} \otimes V^{*,\otimes n}$$
.

Since $\mathcal{M} = \operatorname{Rel}_p(V)$ is overconvergent and the category of overconvergent isocrystals is stable under tensor products, duals and direct sums by [Berthelot 1996, Remark 2.3.3(iii)], we see that $\operatorname{Rel}_p(V_{m,n})$ is overconvergent. Since Rel_p is exact, we see that $\operatorname{Rel}_p(W)$ can be written as the kernel of a map between overconvergent *F*-isocrystals, and is thus overconvergent.

Given a smooth locally closed subscheme $Z \subset \operatorname{Sh}_{G,K^pK_p,\overline{\mathbb{F}}_p}$ and a point $z \in Z(\overline{\mathbb{F}}_p)$, there are monodromy groups

$$Mon(Z, \mathcal{M}, z) \subset Mon(Z, \mathcal{M}^{\dagger}, z),$$

which are algebraic groups over $\check{\mathbb{Q}}_p$, see the introduction of [D'Addezio 2023]. They are defined to be the Tannakian groups corresponding to the smallest Tannakian subcategory of Isoc(Z) respectively Isoc[†](Z) containing \mathcal{M} , via the fibre functor ω_z

$$\omega_z : \operatorname{Isoc}(Z) \to \operatorname{Isoc}(\overline{\mathbb{F}}_p) = \operatorname{Vect}_{\tilde{\mathbb{Q}}_p}$$

We will often omit the chosen point z from the notation since the monodromy group does not depend on z up to isomorphism.

Fix an isomorphism $\mathbb{D}_z := \omega_z(\mathcal{M}^{\dagger}) \to V \otimes \check{\mathbb{Q}}_p$ sending $\omega_z(s_{\alpha}) = s_{\alpha, \mathrm{cris}, z}$ to $s_{\alpha} \otimes 1$. This identifies the composite

$$\omega_z \circ \operatorname{Rel}_p : \operatorname{Rep}_{\mathbb{Q}_p} G \to \operatorname{Vect}_{\check{\mathbb{Q}}_p}$$

with the standard fibre functor, tensored up to \mathbb{Q}_p . Thus if we apply Tannakian duality to (3.3.1) and (3.3.2), we get inclusions

$$\operatorname{Mon}(Z, \mathcal{M}, z) \subset \operatorname{Mon}(Z, \mathcal{M}^{\dagger}, z) \subset G_{\check{\mathbb{Q}}_n}.$$

Corollary 3.3.3. Let $Z \subset \operatorname{Sh}_{G,K^pK_p}$ be a smooth locally closed subscheme and assume that there is a prime $\ell \neq p$ such that Z is $G(\mathbb{Q}_{\ell})$ -stable. Suppose that Z° contains a point x such that $b_{x,i}$ is nonbasic for all i. If Hypothesis 2.3.1 holds, then there is an equality of subgroups of $G_{\mathbb{Q}_p}$

$$\operatorname{Mon}(Z^{\circ}, \mathcal{M}^{\dagger}) = G_{\check{\mathbb{Q}}_{p}}^{\operatorname{der}}.$$

Proof. Let ℓ be as in the statement of the Corollary. Then it follows from Corollary 3.2.6 that the ℓ -adic monodromy group \mathbb{M}_{geom} of the abelian variety over Z is equal to $G_{\mathbb{Q}_{\ell}}^{\text{der}}$. It follows from [D'Addezio 2020, Theorem 1.2.1] (compare [Pál 2022]) that there is an isomorphism of algebraic groups

$$\operatorname{Mon}(Z^{\circ}, \mathcal{M}^{\dagger}) \otimes \overline{\mathbb{Q}}_{p} \simeq G^{\operatorname{der}} \otimes \overline{\mathbb{Q}}_{p}.$$

Therefore $\operatorname{Mon}(Z^{\circ}, \mathcal{M}^{\dagger}) \subset G \otimes \check{\mathbb{Q}}_p$ is a subgroup which is isomorphic to G^{der} over $\overline{\mathbb{Q}}_p$. It follows that $\operatorname{Mon}(Z^{\circ}, \mathcal{M}^{\dagger})$ is equal to its own derived subgroup and therefore contained in $G^{\operatorname{der}} \otimes \check{\mathbb{Q}}_p$. This inclusion has to be an isomorphism for dimension reasons, because both groups are connected.

From now on we will assume that Z is contained in a single Newton stratum $\operatorname{Sh}_{G,[b],K^pK_p}$ of $\operatorname{Sh}_{G,K^pK_p}$. This means that for every representation W of $G_{\mathbb{Q}_p}$ the Newton polygon of $\operatorname{Rel}_p(W)$ is constant. As explained in [D'Addezio 2023, Section 4.3] (cf. [Katz 1979]), this implies that $\operatorname{Rel}_p(W)$ admits a (unique) slope filtration $\operatorname{Rel}_p(W)_{\bullet}$. There is an induced slope filtration on $\omega_z(\operatorname{Rel}_p(W))$, which gives a fractional cocharacter λ_W of $\operatorname{GL}(\omega_z(\operatorname{Rel}_p(W)))$. Since this construction is functorial in W, it defines a fractional cocharacter λ of $G_{\mathbb{Q}_p}$. On the other hand, there is an element $b=b_z\in G(\mathbb{Q}_p)$ correspond to the Frobenius of $\mathbb{D}_z=\omega_z(\mathcal{M}^{\dagger})=V\otimes\mathbb{Q}_p$; let ν_b be the Newton cocharacter of b.

Lemma 3.3.4. There is an equality $\lambda = v_b$.

Proof. It suffices to show that $\lambda = \nu_b$ after composing with $G \otimes \mathbb{Q}_p \to \operatorname{GL}(V) = \operatorname{GL}(\omega_z(\mathcal{M}^{\dagger}))$. But both of these fractional cocharacters of $\operatorname{GL}(V)$ are per definition the slope cocharacters of the *F*-isocrystal $\omega_z(\mathcal{M}^{\dagger})$. Indeed, this is true for λ per definition and for ν_b by its construction; see [Kottwitz 1985, Section 4].

Under our assumption that Z is contained in a single Newton stratum $Sh_{G,[b],K^pK_p}$ of Sh_{G,K^pK_p} we note that the monodromy group

$$\operatorname{Mon}(Z^{\circ}, \mathcal{M}) \subset G_{\check{\mathbb{Q}}_p}$$

of a connected component Z° of Z is contained in the parabolic subgroup $P(\lambda)$ associated to λ , as explained in [D'Addezio 2023, Section 4.1].

Corollary 3.3.5. Let $Z \subset \operatorname{Sh}_{G,K^pK_p}$ be a smooth locally closed subscheme and assume that there is a prime $\ell \neq p$ such that Z is $G(\mathbb{Q}_{\ell})$ -stable. Let Z° be a connected component of Z and suppose that Z° is contained in a single \mathbb{Q} -nonbasic Newton stratum $\operatorname{Sh}_{G,[b],K^pK_p}$. If Hypothesis 2.3.1 holds, then the p-adic monodromy group

$$\operatorname{Mon}(Z^{\circ}, \mathcal{M}) \subset \operatorname{Mon}(Z^{\circ}, \mathcal{M}^{\dagger}) = G_{\check{\mathbb{Q}}_{p}}^{\operatorname{der}} \subset G_{\check{\mathbb{Q}}_{p}}$$

is equal to the intersection

$$G_{\check{\mathbb{Q}}_n}^{\mathrm{der}} \cap P(\lambda).$$

In particular, the unipotent radical of $Mon(Z^{\circ}, \mathcal{M})$ is isomorphic to the unipotent radical of the parabolic subgroup P_{ν_b} of $G_{\mathring{\mathbb{Q}}_n}$.

Proof. The first assertion is a direct consequence of Corollary 3.3.3 and [D'Addezio 2023, Theorem 1.1.1]. The second assertion follows from Lemma 3.3.4

3.4. *Irreducible components of Hecke stable subvarieties.* In this section we will study irreducible components of Hecke stable subvarieties and prove results in the style of [Chai 2005, Proposition 4.5.4]. The results proved in this section will not be used in the rest of this article, but they are used to prove irreducibility results for EKOR strata in [van Hoften 2020].

Let $\rho: G^{\mathrm{sc}} \to G^{\mathrm{der}}$ be the simply connected cover of the derived group G^{der} of G and note that ρ induces an action of $G^{\mathrm{sc}}(\mathbb{A}_f^p)$ on Sh_{G,K_p} . From now on we will need another assumption:

Hypothesis 3.4.1. For each finite extension F of the reflex field E and any place w of F extending v, the natural maps

$$\pi_0(\mathbf{Sh}_{K^pK_n}(G,X)\otimes_E F) \to \pi_0(\mathscr{S}_{K^pK_n}\otimes_{\mathcal{O}_{E,(v)}}\mathcal{O}_{F,(w)}) \leftarrow \pi_0(\mathbf{Sh}_{G,K^pK_n}\otimes_k k(w))$$

are isomorphisms.

Lemma 3.4.2. Hypothesis 3.4.1 holds if Sh_{G,K^pK_n} has geometrically integral connected components.

Proof. This is [Madapusi Pera 2019, Corollary 4.1.11].

Remark 3.4.3. The variety Sh_{G,K^pK_p} has geometrically integral connected components if K_p is hyperspecial because then the integral models are smooth by work of Kisin [2010]. More generally the Kisin–Pappas integral models [2018] have geometrically integral connected components if K_p is very special; see [Kisin and Pappas 2018, Corollary 4.6.26].

3.4.4. Connected components. The following result is well known.

Lemma 3.4.5. Let Y_{∞} be a connected component of the scheme

$$\varprojlim_{U\subset G(\mathbb{A}_f)} \mathbf{Sh}_{U,\mathbb{C}}(G,X).$$

Then Y_{∞} is stable under the action of $G^{sc}(\mathbb{A}_f)$.

Proof. This is a direct consequence of the description of connected components of Shimura varieties and strong approximation for $G^{sc}(\mathbb{Q})$, see [Kisin et al. 2021, Section 5.5.1, Lemma 5.5.4].

³Our results do not literally generalise Chai's results because he works with $\operatorname{Sp}_{2g}(\mathbb{A}_f^p)$ -stable subvarieties while we work with $\operatorname{GSp}_{2g}(\mathbb{A}_f^p)$ -stable subvarieties.

Corollary 3.4.6. Let Y_{∞} be a connected component of

$$\varprojlim_{U^p\subset G(\mathbb{A}_f^p)} \mathbf{Sh}_{U^pK_p,\mathbb{C}}(G,X).$$

Then Y_{∞} is stable under the action of $G^{\text{sc}}(\mathbb{A}_f^p)$.

Proof. We consider

$$\varprojlim_{U\subset G(\mathbb{A}_f^p)}\mathbf{Sh}_U(G,X)\to \varprojlim_{U^p\subset G(\mathbb{A}_f^p)}\mathbf{Sh}_{U^pK_p}(G,X)$$

and we let Y'_{∞} be a connected component of the left hand side mapping to Y_{∞} . Then Y'_{∞} is stable under the action of $G^{\text{sc}}(\mathbb{A}_f^p)$.

Lemma 3.4.7. Suppose that Hypothesis 3.4.1 holds and let $Y_{\infty} \subset \operatorname{Sh}_{G,K_p,\overline{\mathbb{F}}_p}$ be a connected component. Then Y_{∞} is stable under the action of $G^{\operatorname{sc}}(\mathbb{A}_f^p)$.

Proof. It suffices to prove this for Shimura varieties over \mathbb{C} , because the connected components are defined over an algebraic closure \overline{E} of the reflex field E and the result can be transported to the special fibre using Hypothesis 3.4.1. The result over \mathbb{C} is Corollary 3.4.6.

3.4.8. Let $Z \subset \operatorname{Sh}_{G,K^pK_p}$ be a $G(\mathbb{A}_f^p)$ -stable locally closed subscheme with inverse image $\widetilde{Z} \subset \operatorname{Sh}_{G,K_p}$. A finite étale cover $X \to Z$ is called $G(\mathbb{A}_f^p)$ -equivariant if $\widetilde{X} := \widetilde{Z} \times_Z X$ has an action of $G(\mathbb{A}_f^p)$ making the natural map $\widetilde{X} \to \widetilde{Z}$ equivariant for the action of $G(\mathbb{A}_f^p)$. If Hypothesis 3.4.1 is satisfied, then by Lemma 3.4.7 the group $G^{\operatorname{sc}}(\mathbb{A}_f^p)$ acts on the fibres of

$$\pi_0(\widetilde{X}) \to \pi_0(\operatorname{Sh}_{G,K_n,\overline{\mathbb{F}}_n}).$$

Lemma 3.4.9. If Hypothesis 3.4.1 holds, then $G(\mathbb{A}_f^p)$ acts continuously on $\pi_0(\widetilde{X})$.

Proof. The assumption that $X \to Z$ is finite étale implies that $\pi_0(\widetilde{X}) \to \pi_0(\widetilde{Z})$ is a finite map with discrete fibres, and therefore the action of $G(\mathbb{A}_f^p)$ on $\pi_0(\widetilde{X})$ is continuous because the action on $\pi_0(\widetilde{Z})$ is, see Corollary 3.1.5.

Let Σ be a finite set of places of $\mathbb Q$ containing p and containing all places ℓ where G_ℓ^{ad} has a compact factor. From now on we will work with $G(\mathbb A_f^p)$ -stable subvarieties Z defined over $\overline{\mathbb F}_p$ and with geometric monodromy groups.

Theorem 3.4.10. Let $X \to Z$ be a $G(\mathbb{A}_f^p)$ -equivariant finite étale cover of a smooth $G(\mathbb{A}_f^p)$ -stable locally closed subscheme $Z \subset \operatorname{Sh}_{G,K^pK_p,\overline{\mathbb{F}}_p}$, and suppose that each connected component of Z intersects a \mathbb{Q} -nonbasic Newton stratum. If Hypotheses 2.3.1 and 3.4.1 hold, then $G(\mathbb{A}_f^{\Sigma})$ acts trivially on the fibres of

$$\pi_0(\widetilde{X}) \to \pi_0(\operatorname{Sh}_{G,K_p,\overline{\mathbb{F}}_p}).$$

For a prime $\ell \notin \Sigma$ we will write K_ℓ for the image of $K^p \to G(\mathbb{Q}_\ell)$ and $\pi_\ell : \operatorname{Sh}_{G,K^pK_p,\ell,\overline{\mathbb{F}}_p} \to \operatorname{Sh}_{G,K^pK_p,\overline{\mathbb{F}}_p}$ for the induced K_ℓ -torsor over $\operatorname{Sh}_{G,K^pK_p,\overline{\mathbb{F}}_p}$.

Lemma 3.4.11. Suppose that Hypothesis 2.3.1 holds and let $Y_{\infty} \subset \operatorname{Sh}_{G,K^pK_p,\ell,\overline{\mathbb{F}}_p}$ be a connected component with image $Y \subset \operatorname{Sh}_{G,K^pK_n,\overline{\mathbb{F}}_p}$. Then $Y_{\infty} \to Y$ is a torsor for a compact open subgroup of $G^{\operatorname{der}}(\mathbb{Q}_{\ell})$.

Proof. It follows from profinite Galois theory for schemes, see Section 3.1.11, that the stabiliser K_{∞} of Y_{∞} in $G(\mathbb{Q}_{\ell})$ can be identified with the image of

$$\pi_1^{\text{\'et}}(Y, y) \to G(\mathbb{Q}_\ell)$$

for some point $y \in Y(\overline{\mathbb{F}}_p)$. If we apply Corollary 3.2.6 and Lemma 3.1.9 to $Z = \operatorname{Sh}_{G,K^pK_p}$, it follows that this image contains a compact open subgroup of $G^{\operatorname{der}}(\mathbb{Q}_\ell)$ and that it is contained in $G^{\operatorname{der}}(\mathbb{Q}_\ell)$.

Proof of Theorem 3.4.10. We write $Z_{\ell} \to Z$ for the induced K_{ℓ} torsor and $X_{\ell} \to Z_{\ell}$ for $Z_{\ell} \times_{Z} X$. Then the action of $G(\mathbb{A}_{f}^{p})$ on \widetilde{X} and \widetilde{Z} induces an action of $G(\mathbb{Q}_{\ell})$ on X_{ℓ} , and it suffices to show that $G^{\text{sc}}(\mathbb{Q}_{\ell})$ acts trivially on the fibres of

$$a_{\ell}: \pi_0(X_{\ell}) \to \pi_0(\operatorname{Sh}_{G,K^pK_n,\ell,\overline{\mathbb{F}}_p})$$

for all $\ell \notin \Sigma$.

Let $x \in \pi_0(X_\ell)$ and let Z° be a connected component of Z containing the image of x. Moreover let $Z_\ell^\circ \subset \mathbb{Z}_\ell$ be the inverse image of Z° . Fix a point $z \in Z^\circ(\overline{\mathbb{F}}_p)$, then Hypothesis 2.3.1 and Corollary 3.2.6 tell us that the image of

$$\rho_{\ell}: \pi_1^{\text{\'et}}(Z^{\circ}, z) \to K_{\ell}$$

is a compact subgroup $M_{\text{geom},\ell}$ whose Zariski closure $\mathbb{M}_{\text{geom},\ell}$ has neutral component equal to $G_{\mathbb{Q}_{\ell}}^{\text{der}}$. It follows from Lemma 3.1.9 that the image of ρ_{ℓ} contains a compact open subgroup $V_{\ell} \subset G^{\text{der}}(\mathbb{Q}_{\ell})$. The upshot of this discussion is that the stabiliser in $G(\mathbb{Q}_{\ell})$ of a connected component of Z_{ℓ} contains a compact open subgroup of $G^{\text{der}}(\mathbb{Q}_{\ell})$ and this implies that the stabiliser in $G(\mathbb{Q}_{\ell})$ of x contains a compact open subgroup of $G^{\text{der}}(\mathbb{Q}_{\ell})$.

Let Y_{∞} be a connected component of $\operatorname{Sh}_{G,K^pK_p,\ell,\overline{\mathbb{F}}_p}$ such that the image Y of Y_{∞} in $\operatorname{Sh}_{G,K^pK_p}$ contains Z° . Then it follows from Hypothesis 2.3.1 and Lemma 3.4.11 that $Y_{\infty} \to Y$ is a pro-étale torsor for a compact open subgroup $U_{\ell} \subset G^{\operatorname{der}}$, and from Hypothesis 3.4.1 and Lemma 3.4.7 that Y_{∞} is stable under the action of $G^{\operatorname{sc}}(\mathbb{Q}_{\ell})$.

We will write $X_{\infty} \subset X_{\ell}$ for the inverse image of Y_{∞} in X_{ℓ} and let $X' \subset X$ be its image. Note that $x \in \pi_0(X_{\infty})$ by construction. Then $X_{\infty} \to X'$ is a pro-étale U_{ℓ} torsor and X_{∞} is stable under the action of $G^{\text{sc}}(\mathbb{Q}_{\ell})$. This action is moreover continuous by Lemma 3.4.9 and the inclusion

$$\pi_0(X_\infty) \subset \pi_0(X_\ell)$$

is closed since $\{Y_\infty\} \subset \pi_0(\operatorname{Sh}_{G,K^pK_p,\ell,\overline{\mathbb{F}}_p})$ is closed. In particular, the topological space $\pi_0(X_\infty)$ is compact Hausdorff.

Let U'_{ℓ} be the inverse image of U_{ℓ} in $G^{\mathrm{sc}}(\mathbb{Q}_{\ell})$. Then the quotient

$$U'_{\ell} \backslash \pi_0(X_{\infty}) = U_{\ell} \backslash \pi_0(X_{\infty}) = \pi_0(X')$$

is finite. This means that there are finitely many (open and closed) U'_{ℓ} orbits on $\pi_0(X_{\infty})$. Therefore the $G^{\mathrm{sc}}(\mathbb{Q}_{\ell})$ orbit of x on $\pi_0(X_{\infty})$ is a union of finitely many U'_{ℓ} -orbits and thus closed; in particular it is compact Hausdorff. It then follows from Lemma 3.1.6 that the $G^{\mathrm{sc}}(\mathbb{Q}_{\ell})$ orbit of x is homeomorphic to $G^{\mathrm{sc}}(\mathbb{Q}_{\ell})/P_x$, where $P_x \subset G^{\mathrm{sc}}(\mathbb{Q}_{\ell})$ is the stabiliser of x. In particular, it follows that $G^{\mathrm{sc}}(\mathbb{Q}_{\ell})/P_x$ is compact.

The group P_x contains a compact open subgroup of $G^{\operatorname{sc}}(\mathbb{Q}_\ell)$ because the stabiliser of x in $G(\mathbb{Q}_\ell)$ contains a compact open subgroup of $G^{\operatorname{der}}(\mathbb{Q}_\ell)$ and $G^{\operatorname{sc}}(\mathbb{Q}_\ell) \to G^{\operatorname{der}}(\mathbb{Q}_\ell)$ has finite fibres. This implies that $G^{\operatorname{sc}}(\mathbb{Q}_\ell)/P_x$ has the discrete topology, and we conclude that $G^{\operatorname{sc}}(\mathbb{Q}_\ell)/P_x$ is a finite set or equivalently that P_x is a finite index subgroup. The assumption that $G^{\operatorname{sc}}_{\mathbb{Q}_\ell}$ has no compact factors implies, by [Platonov and Rapinchuk 1994, Theorems 7.1 and 7.5], that the group $G^{\operatorname{sc}}(\mathbb{Q}_\ell)$ has no finite index subgroups. Therefore $G^{\operatorname{sc}}(\mathbb{Q}_\ell)/P_x$ is a singleton which is precisely what we wanted to prove.

4. Serre-Tate coordinates and unipotent group actions

In this section we show that the classical Serre–Tate coordinates, as described in [Katz 1981], can be reinterpreted using the action of a unipotent formal group, as in [Howe 2020]. Our results are more-or-less a direct generalisation of the results of [Howe 2020], except that we construct the action of unipotent formal groups using Rapoport–Zink spaces, while in [loc. cit.] this action is constructed using Igusa varieties.

In Section 4.1, we recall the classical theory of Serre–Tate coordinates following [Katz 1981], which shows that the formal deformation space Def(Y) of an ordinary p-divisible group Y over $\overline{\mathbb{F}}_p$ has the structure of a commutative formal group. We then compute the scheme-theoretic p-adic Tate-module of the p-divisible group $\mathcal{H}_{0,1}$ associated to this formal group. In Section 4.2 we use Rapoport–Zink spaces to describe an action of the universal cover $\widetilde{\mathcal{H}}_{0,1}$ on the formal scheme $\widehat{Def}(Y)$ associated to Def(Y). In Section 4.3 we identify this action with the projection from the universal cover to $\mathcal{H}_{0,1}$ followed by the left-translation action of $\mathcal{H}_{0,1}$ on $\widehat{Def}(Y)$.

4.0.1. We consider the category Art of Artin local \mathbb{Z}_p -algebras R such that the natural map $\mathbb{F}_p \to R/\mathfrak{m}_R$ is an isomorphism. Here \mathfrak{m}_R is the unique maximal ideal of R and we write $\alpha: R \to \mathbb{F}_p$ for the composition of the natural map $R \to R/\mathfrak{m}$ with the inverse of the natural isomorphism $\mathbb{F}_p \to R/\mathfrak{m}$. Note that α is functorial for morphisms in Art. We similarly consider the category Nilp of \mathbb{Z}_p -algebras in which $p^n = 0$ for some n. The category Art is naturally a full subcategory of Nilp.

For a p-divisible group \mathscr{G} over an algebra $R \in \text{Nilp}$ we define the p-adic Tate module to be the functor $T_p\mathscr{G} := \varprojlim_n \mathscr{G}[p^n]$, which is representable by a flat affine scheme over Spec R by [Scholze and Weinstein 2013, Proposition 3.3.1].

4.1. Classical Serre-Tate theory. Let Y be an ordinary p-divisible group of dimension g and height 2g over $\overline{\mathbb{F}}_p$. In other words, let Y be a p-divisible group isomorphic to $(\mathbb{Q}_p/\mathbb{Z}_p)^{\oplus g} \oplus \mu_{p^{\infty}}^{\oplus g}$.

Let $\operatorname{Def}(Y)$ be the functor on Art sending (R,α) to the set of isomorphism classes of pairs (X,β) where X is a p-divisible group over $\operatorname{Spec} R$ and $\beta: X \otimes_{R,\alpha} \overline{\mathbb{F}}_p \to Y_{\overline{\mathbb{F}}_p}$ is an isomorphism. This functor is (pro)-representable by a formally smooth formal scheme $\operatorname{Def}(Y)$ of relative dimension g^2 over $\operatorname{Spf} \check{\mathbb{Z}}_p$.

By [Katz 1981, Theorem 2.1], this functor lifts to a functor valued in abelian groups such that the formal group Def(Y) is p-divisible.⁴

There is a canonical direct sum decomposition $Y = Y_0 \oplus Y_1$ where Y_0 is the maximal étale quotient of Y and where Y_1 is equal to the formal completion of Y at the origin. Since Y_0 is étale there is a unique lift to a p-divisible formal group over \mathbb{Z}_p , which we will denote by Y_0^{can} . Similarly Y_1 has a unique lift to a p-divisible formal group over \mathbb{Z}_p , for example because the Serre dual of Y_1 is étale. We will denote this lift by Y_1^{can} and we will use $Y^{\operatorname{can}} = Y_0^{\operatorname{can}} \oplus Y_1^{\operatorname{can}}$ to denote the canonical lift of Y to \mathbb{Z}_p .

Let Y^{\vee} be the Serre-dual of Y and consider the free \mathbb{Z}_p -modules of rank g given by $T_pY(\overline{\mathbb{F}}_p)$ and $T_pY^{\vee}(\overline{\mathbb{F}}_p)$. By [Katz 1981, Theorem 2.1], the formal group $\mathrm{Def}(Y)$ is isomorphic to the functor on Art sending R to

$$\hom(T_p Y(\overline{\mathbb{F}}_p) \otimes_{\mathbb{Z}_p} T_p Y^{\vee}(\overline{\mathbb{F}}_p), \widehat{\mathbb{G}}_m(R)).$$

Let S be the complete Noetherian local \mathbb{Z}_p -algebra representing $\mathrm{Def}(Y)$ on Art. Then the abelian group structure on $\mathrm{Def}(Y)$ induces a (continuous) cocommutative Hopf algebra structure on S. In particular the formal scheme $\widehat{\mathrm{Def}}(Y) := \mathrm{Spf}\, S$, considered as a functor on Nilp, has the structure of a formal group. We will write $\mathcal{H}_{0,1} := \varinjlim \mathrm{Spf}\, S[p^n]$ for the corresponding p-divisible group over $\mathrm{Spf}\,\mathbb{Z}_p$. Note that it acts via left translation on $\widehat{\mathrm{Def}}(Y)$; we will denote this action by a_{ST} (for Serre–Tate).

Remark 4.1.1. The natural map $\mathcal{H}_{0,1} \to \widehat{\mathrm{Def}}(Y)$ is an isomorphism of formal schemes, since both of them are formally smooth formal schemes of the same dimension. Nevertheless, it is useful to treat them as different objects, for example because the notation $\widehat{\mathrm{Def}}(Y)$ is somewhat unwieldy, especially when passing to universal covers of p-divisible groups.

Lemma 4.1.2. The p-adic Tate module of $\mathcal{H}_{0,1}$ is isomorphic to the sheaf $\mathscr{H}om(Y_0^{can}, Y_1^{can})$ on Nilp of homomorphisms from Y_0^{can} to Y_1^{can} .

Proof. Let us prove the stronger assertion that there are isomorphisms $\mathcal{H}_{0,1}[p^n] \simeq \mathscr{H}om(Y_0^{\operatorname{can}}, Y_1^{\operatorname{can}})[p^n]$ for all n, compatible with changing n. Note that $\mathcal{H}_{0,1}[p^n]$ is represented by the spectrum of an Artin local \mathbb{Z}_p -algebra. The same is true for $\mathscr{H}om(Y_0^{\operatorname{can}}, Y_1^{\operatorname{can}})[p^n]$, since $\mathscr{H}om(\mathbb{Z}/p^n\mathbb{Z}, \mu_{p^n}) \simeq \mu_{p^n}$. Thus it suffices to show that the functors $\mathcal{H}_{0,1}[p^n]$ and $\mathscr{H}om(Y_0^{\operatorname{can}}, Y_1^{\operatorname{can}})[p^n]$ are isomorphic as functors on Art. In [Katz 1981, p. 152] it is explained that $\operatorname{Def}(Y)$ is isomorphic to the functor (on Art) sending R to

$$hom(T_pY(\overline{\mathbb{F}}_p), Y_1^{can}(R)).$$

Note that $T_pY(\overline{\mathbb{F}}_p)=T_pY_0(\overline{\mathbb{F}}_p)=T_pY_0^{\operatorname{can}}(\overline{\mathbb{F}}_p)$ and that because $T_pY_0^{\operatorname{can}}$ is an inverse limit of étale group schemes, the natural map $T_pY_0^{\operatorname{can}}(R)\to T_pY_0(\overline{\mathbb{F}}_p)$ is an isomorphism for $R\in\operatorname{Art}$. Thus there is a natural isomorphism

$$\hom(T_n Y(\overline{\mathbb{F}}_n), Y_1^{\operatorname{can}}(R)) \simeq \hom(T_n Y_0^{\operatorname{can}}(R), Y_1^{\operatorname{can}}(R)).$$

⁴Recall that a commutative formal group X is called *p-divisible* if $[p]: X \to X$ is finite flat.

The p^n -torsion of this group is given by

$$\hom(T_p Y_0^{\operatorname{can}}(R), Y_1^{\operatorname{can}}(R))[p^n] = \hom(T_p Y_0^{\operatorname{can}}(R), Y_1^{\operatorname{can}}[p^n](R)) = \hom(Y_0^{\operatorname{can}}[p^n](R), Y_1^{\operatorname{can}}[p^n](R)).$$

We see that there is an isomorphism $\operatorname{Def}(Y)[p^n] \simeq \mathscr{H}om(Y_0^{\operatorname{can}}[p^n], Y_1^{\operatorname{can}}[p^n])$ of functors on Art, which induces an isomorphism $\mathcal{H}_{0,1}[p^n] \simeq \mathscr{H}om(Y_0^{\operatorname{can}}[p^n], Y_1^{\operatorname{can}}[p^n])$ of functors on Nilp. It is straightforward to check that these isomorphisms are compatible with increasing n, which concludes the proof.

4.2. Rapoport–Zink spaces and unipotent formal groups. Let $\widetilde{Y} \to Y$ be the universal cover of Y, defined as the inverse limit of the projective system

$$\underbrace{\lim}_{[p]:G\to G} Y.$$

It is representable by a formal scheme by [Scholze and Weinstein 2013, Proposition 3.1.3(iii)]. By the proof of [Caraiani and Scholze 2017, Proposition 4.2.11], the automorphism group functors of Y and \widetilde{Y} on Nilp can be described as follows:

$$\mathbf{Aut}(Y) = \begin{pmatrix} \mathbf{Aut}(Y_0) & 0 \\ \mathscr{H}om(Y_0, Y_1) & \mathbf{Aut}(Y_1) \end{pmatrix}, \quad \mathbf{Aut}(\widetilde{Y}) = \begin{pmatrix} \mathbf{Aut}(\widetilde{Y}_0) & 0 \\ \mathscr{H}om(Y_0, Y_1)[1/p] & \mathbf{Aut}(\widetilde{Y}_1) \end{pmatrix}.$$

Moreover the functors $\operatorname{Aut}(Y_i)$ are pro-étale group schemes which are noncanonically isomorphic to the group schemes associated to the profinite group $\operatorname{GL}_g(\mathbb{Z}_p)$. Let $\widetilde{\mathcal{H}}_{0,1}$ be the universal cover of $\mathcal{H}_{0,1}$. Then by the discussion after [Caraiani and Scholze 2017, Definition 4.1.1], we can identify the fpqc sheaves

$$\widetilde{\mathcal{H}}_{0,1} = (T_p \mathcal{H}_{0,1})[1/p].$$

Moreover, by the proof of [Caraiani and Scholze 2017, Proposition 4.1.2], there is a short exact sequence of fpqc sheaves

$$0 \to T_p \mathcal{H}_{0,1} \to \widetilde{\mathcal{H}}_{0,1} \to \mathcal{H}_{0,1} \to 0.$$

By Lemma 4.1.2, we can identify this with

$$0 \to \mathcal{H}om(Y_0,Y_1) \to \mathcal{H}om(Y_0,Y_1)[1/p] \to \mathcal{H}_{0,1} \to 0.$$

Note that $\mathcal{H}om(Y_0, Y_1)[1/p]$ is isomorphic to $\widetilde{\mathcal{H}}_{0,1}$, and thus representable by a formal scheme by [Scholze and Weinstein 2013, Proposition 3.1.3(iii)] as above. In particular, this means that $\mathbf{Aut}(\widetilde{Y})$ is representable by a formal scheme.

4.2.1. Let RZ_Y be the Rapoport–Zink space associated to Y. It is defined to be the functor on Nilp sending R to the set of isomorphism classes of pairs (X, f), where X is a p-divisible group over Spec R and $f: X \dashrightarrow Y_R$ is a quasi-isogeny (or equivalently, by [Katz 1981, Lemma 1.1.3.3], a quasi-isogeny $f_0: X_{R/pR} \dashrightarrow Y_{R/pR}$). The functor RZ_Y is representable by a formally smooth formal scheme over Spf \mathbb{Z}_p by [Rapoport and Zink 1996, Theorem 2.16]. The group functor $\mathbf{Aut}(\widetilde{Y})$ acts on RZ_Y via postcomposition, where we note that an automorphism of \widetilde{Y} is the same thing as a self-quasi-isogeny of Y.

Let y be the $\overline{\mathbb{F}}_p$ -point of RZ_Y corresponding to the identity map $Y \to Y$ and let

$$RZ_Y^{/y} \subset RZ_Y$$

be the formal completion of RZ_Y in $\{y\}$, in the sense of formal algebraic spaces as in [Stacks 2020, Tag0GVR]. By definition this is the subfunctor of RZ_Y corresponding to those morphisms $\operatorname{Spec} R \to RZ_Y$ that factor through $\{y\}$ on the level of topological spaces. In other words, it consists of those morphisms $\operatorname{Spec} R \to RZ_Y$ such that the induced morphism $\operatorname{Spec} R^{\operatorname{red}} \subset \operatorname{Spec} R \to RZ_Y$ factors through $y:\operatorname{Spec} \overline{\mathbb{F}}_p \to RZ_Y$.

In terms of the moduli description, this means that we are looking at those quasi-isogenies $f: X \dashrightarrow Y_R$ such that: There is a (necessarily unique) isomorphism $\beta: X_{R^{\text{red}}} \to Y_{R^{\text{red}}}$ making the following diagram commute:

$$\begin{array}{ccc} X_{R^{\mathrm{red}}} & \stackrel{\beta}{\longrightarrow} Y_{R^{\mathrm{red}}} \\ & \downarrow^{f} & & \parallel \\ Y_{R^{\mathrm{red}}} & = = & Y_{R^{\mathrm{red}}} \end{array} \tag{4.2.1}$$

Now restrict this moduli description to the full subcategory Art \subset Nilp. Then $RZ_Y^{/y}$ can be described as the functor on Art sending (R, α) to the set of isomorphisms classes of triples (X, β, f) , where X is a p-divisible group over R equipped with an isomorphism $\beta: X \otimes_{R,\alpha} \overline{\mathbb{F}}_p \to Y$ and where f is a quasi-isogeny $f: X \dashrightarrow Y_R$ such that (4.2.1) commutes.

Lemma 4.2.2. The natural forgetful map $RZ_Y^{/y} \to Def(Y)$ sending (X, β, f) to (X, β) is an isomorphism. In particular, there is an isomorphism of formal schemes $\widehat{Def}(Y) \simeq RZ_Y^{/y}$.

Proof. The commutativity of (4.2.1) expresses the fact that f is a lift of the quasi-isogeny $Y \to Y$ given by the identity. But since quasi-isogenies lift uniquely by [Katz 1981, Lemma 1.1.3.3], the data of f is superfluous and we see that the forgetful map $RZ_Y^{/y}(R) \to Def(Y)(R)$ is a bijection for all $R \in Art$. \square

The subgroup

$$\begin{pmatrix} \mathbf{Aut}(Y_0) & 0 \\ \widetilde{\mathcal{H}}_{0,1} & \mathbf{Aut}(Y_1), \end{pmatrix} \subset \mathbf{Aut}(\widetilde{Y})$$

preserves the point $y \in RZ_Y(\overline{\mathbb{F}}_p)$ and therefore acts on $\widehat{Def}(Y)$. In particular, the profinite group

$$\operatorname{Aut}(Y_0)(\overline{\mathbb{F}}_p) \times \operatorname{Aut}(Y_1)(\overline{\mathbb{F}}_p) = \operatorname{Aut}(Y)(\overline{\mathbb{F}}_p)$$

acts on $\widehat{\mathrm{Def}}(Y)$. This induces an action of $\mathrm{Aut}(Y)(\overline{\mathbb{F}}_p)$ on $\mathrm{Def}(Y)$ because $\overline{\mathbb{F}}_p$ is an object of $\mathrm{Art}\subset\mathrm{Nilp}$.

Corollary 4.2.3. This action sends a pair $(X, \beta) \in \text{Def}(Y)(R)$, where X is a p-divisible group over Spec R and $\beta : X \otimes_{R,\alpha} \overline{\mathbb{F}}_p \to Y_{\overline{\mathbb{F}}_p}$ is an isomorphism, to $(X, g \circ \beta)$ for $g \in \text{Aut}(Y)(\overline{\mathbb{F}}_p)$.

Proof. This follows from Lemma 4.2.2 and the uniqueness of the isomorphism $\beta: X_{\overline{\mathbb{F}}_p} \to Y$ given $f: X \dashrightarrow Y_R$.

4.2.4. Since the action of $\widetilde{\mathcal{H}}_{0,1}$ on RZ_Y preserves the point y, there is an induced action

$$\widetilde{a}_{RZ}: \widetilde{\mathcal{H}}_{0,1} \times \widehat{\operatorname{Def}}(Y) \to \widehat{\operatorname{Def}}(Y).$$

The goal of the rest of this section is to prove the following proposition, our proof of which was heavily inspired by the proof of [Howe 2020, Theorem 6.2.1], which deals with the g = 1 case.

Proposition 4.2.5. The action \tilde{a}_{RZ} factors through an action of $\mathcal{H}_{0,1}$ via the natural quotient map $\widetilde{\mathcal{H}}_{0,1} \to \mathcal{H}_{0,1}$. Moreover the induced action of $\mathcal{H}_{0,1}$ is given by a_{ST} .

4.3. *Proof of Proposition 4.2.5.* Choose isomorphisms

$$T_p Y_0 \simeq \mathbb{Z}_p^{\oplus g}, \quad Y_1 \simeq (\mu_{p^{\infty}})^{\oplus g},$$

which induce isomorphisms of functors on Art

$$\operatorname{Def}(Y) \simeq \mathscr{H}om(\mathbb{Z}_p^{\oplus g}, (\mu_{p^{\infty}})^{\oplus g}).$$

In fact if we let $x_1, \ldots, x_g \in \mathbb{Z}_p^{\oplus g}$ be the standard basis vectors, then we can in fact identify

$$\operatorname{Def}(Y) \simeq (\mu_{p^{\infty}})^{\oplus g^2}$$

with coordinates $q_{i,j}$ for $1 \le i, j \le g$ and similarly

$$\mathcal{H}_{0.1} \simeq (\mu_{p^{\infty}})^{\oplus g^2}$$

with coordinates $\zeta_{i,j}$ for $1 \le i, j \le g$. For R in Art a morphism

Spec
$$R \to Def(Y)$$

corresponds to elements $q_{i,j} \in 1 + \mathfrak{m}_R$, and the corresponding deformation of Y is the p-divisible group X_q corresponding to the pushout of (see [Katz 1981, p. 152])

$$0 \to \mathbb{Z}_p^{\oplus g} \to \mathbb{Q}_p^{\oplus g} \to \frac{\mathbb{Q}_p^{\oplus g}}{\mathbb{Z}_p^{\oplus g}} \to 0$$

via the morphism $\mathbb{Z}_{p,R}^{\oplus g} \to \mu_{p^{\infty},R}^{\oplus g}$ given by $x_i \mapsto (q_{i,1},\ldots,q_{i,g})$. In fact, there is a pushout diagram

$$0 \longrightarrow \mathbb{Z}^{\oplus g} \longrightarrow \mathbb{Z}[1/p]^{\oplus g} \longrightarrow \frac{\mathbb{Q}_{p}^{\oplus g}}{\mathbb{Z}_{p}^{\oplus g}} \longrightarrow 0$$

$$\downarrow \qquad \qquad \downarrow \qquad \qquad \parallel$$

$$0 \longrightarrow \mathbb{Z}_{p}^{\oplus g} \longrightarrow \mathbb{Q}_{p}^{\oplus g} \longrightarrow \frac{\mathbb{Q}_{p}^{\oplus g}}{\mathbb{Z}_{p}^{\oplus g}} \longrightarrow 0$$

and so we can also think of X_q as the quotient of $\mu_{p^{\infty},R}^{\oplus g} \oplus \mathbb{Z}[1/p]^{\oplus g}$ by the image of the map

$$h_q: \mathbb{Z}^{\oplus g} \to \mu_{p^{\infty}, R}^{\oplus g} \oplus \mathbb{Z}[1/p]^{\oplus g}$$

given by $x_i \mapsto ((q_{i,1}, ..., q_{i,g}), (x_i)).$

Let N be an integer such that $q_{i,j}^{p^N} = 1$ for all i, j, which exists since R is Artinian. Then the isogeny

$$\mu_{p^{\infty},R}^{\oplus g} \oplus \mathbb{Z}[1/p]^{\oplus g} \to \mu_{p^{\infty},R}^{\oplus g} \oplus \mathbb{Z}[1/p]^{\oplus g}, \quad (A,B) \mapsto (p^N A, p^N B)$$

maps $h_q(\mathbb{Z}^{\oplus g})$ into $h_1(\mathbb{Z}^{\oplus g})$. Thus it induces a quasi-isogeny

$$f_{q,N}: X_q \dashrightarrow X_{\underline{1}} = Y_R, \tag{4.3.1}$$

and the induced quasi-isogeny $X_{\underline{q},\overline{\mathbb{F}}_p}=X_{\underline{1},\overline{\mathbb{F}}_p}\to X_{\underline{1},\overline{\mathbb{F}}_p}$ is given by p^N . It follows that the quasi-isogeny $p^{-N}f_{\underline{q},N}$ is the unique quasi-isogeny lifting the identity $X_{\underline{q},\overline{\mathbb{F}}_p}=X_{\underline{1},\overline{\mathbb{F}}_p}\to X_{\underline{1},\overline{\mathbb{F}}_p}$. Let us write $\underline{q}\in\mathrm{RZ}_Y(R)$ for $p^{-N}f_{q,N}:X_q\dashrightarrow Y_R$.

A morphism Spec $R \to \mathcal{H}_{0,1}$ corresponds to elements $\zeta_{i,j} \in 1 + \mathfrak{m}_R$. The left translation action of $\mathcal{H}_{0,1}$ via the Serre–Tate action is given by

$$a_{\rm ST}(\underline{\zeta},q) = \underline{\zeta}q,$$

where $(\underline{\zeta}q)_{i,j} = (\zeta_{i,j} \cdot q_{i,j})$ and where $(\zeta_{i,j} \cdot q_{i,j})$ denotes the multiplication in $\mu_{p^{\infty}}(R) = 1 + \mathfrak{m}_R$. In terms of p-divisible groups, this correspond to the p-divisible group $X_{\underline{\zeta}q}$. We will write $\underline{\zeta}q \in RZ_Y(R)$ for the element corresponding to $X_{\zeta q}$.

Proof of Proposition 4.2.5. By definition of the action \tilde{a}_{RZ} , it suffices to show that for every fpqc cover Spec $\tilde{R} \to \text{Spec } R$ and every lift

$$\tilde{\zeta} \in \widetilde{\mathcal{H}}_{0,1}(\tilde{R})$$

of $\underline{\zeta} \in \mathcal{H}_{0,1}(\tilde{R})$, we have $\tilde{a}_{RZ}(\underline{\tilde{\zeta}},\underline{q}) = \underline{\zeta}\underline{q}$. There is a universal such lift over the fpqc cover \tilde{R} given by formally adjoining all the p-power roots of all $\zeta_{i,j}$, and it suffices to prove the result for this choice of \tilde{R} . Recall that $X_{q,\tilde{R}}$ is defined as the quotient of

$$\mu_{p^{\infty},\tilde{R}}^{\oplus g} \oplus \mathbb{Z}[1/p]^{\oplus g}$$

by the image of the map h_q which sends the standard basis element $x_i \in \mathbb{Z}^{\oplus g}$ to

$$((q_{i,1},\ldots,q_{i,g}),(x_i))\in\mu_{p^{\infty},\tilde{R}}^{\oplus g}\oplus\mathbb{Z}[1/p]^{\oplus g}.$$

The *p*-divisible group $X_{\underline{\zeta}\underline{q},\tilde{R}}$ is defined similarly but then using the map $h_{\underline{\zeta}\underline{q}}$. The compatible sequence of *p*-power roots of ζ defined by $\tilde{\zeta}$ defines a map

$$\psi_{\tilde{\underline{\zeta}}}: \mu_{p^{\infty}, \tilde{R}}^{\oplus g} \oplus \mathbb{Z}[1/p]^{\oplus g} \to \mu_{p^{\infty}, \tilde{R}}^{\oplus g} \oplus \mathbb{Z}[1/p]^{\oplus g}, \quad (A, B) \mapsto (A \cdot L_{\tilde{\underline{\zeta}}}(B), B),$$

where $L_{\tilde{\underline{\zeta}}}: \mathbb{Z}[1/p]^{\oplus g} \to \mu_{p^{\infty}, \tilde{R}}^{\oplus g}$ is the morphism sending

$$\frac{x_i}{p^n} \mapsto (\zeta_{i,1}^{1/p^n}, \zeta_{i,2}^{1/p^n}, \dots, \zeta_{i,g}^{1/p^n}).$$

It is straightforward to check that this map satisfies

$$\psi_{\tilde{\zeta}}(h_{\underline{q}}(\mathbb{Z}^{\oplus g}) = h_{\underline{\zeta}\underline{q}}(\mathbb{Z}^{\oplus g})$$

and that it thus induces an isomorphism on quotients

$$\phi_{\underline{\tilde{\zeta}}}: X_{\underline{q}, \tilde{R}} \to X_{\underline{\zeta}\underline{q}, \tilde{R}}.$$

Choose N sufficiently large such that $\zeta_{i,j}^{p^N}=1$ and $q_{i,j}^{p^N}=1$ for all i,j. Let

$$p^{-2N}f_{\zeta q,2N}:X_{\zeta q}\dashrightarrow Y_R$$

be the unique quasi-isogeny lifting the identity map $X_{\underline{\zeta}\underline{q},\overline{\mathbb{F}}_p}=Y_{\overline{\mathbb{F}}_p}\to Y_{\overline{\mathbb{F}}_p}$ as described in (4.3.1). To prove the proposition it suffices to show that the following diagram commutes:

$$X_{\underline{q},\tilde{R}} \xrightarrow{\phi_{\underline{\xi}}} X_{\underline{\xi}\underline{q},\tilde{R}}$$

$$\downarrow^{p^{-N}f_{\underline{q},N}} \qquad \qquad \downarrow^{p^{-2N}f_{\underline{\xi}\underline{q},2N}}$$

$$\left(\frac{\mathbb{Q}_p^{\oplus g}}{\mathbb{Z}_p^{\oplus g}}\right)_{\tilde{R}} \oplus \mu_{p^{\infty},\tilde{R}}^{\oplus g} \xrightarrow{-\frac{\xi_{\underline{\xi}}}{2}} \left(\frac{\mathbb{Q}_p^{\oplus g}}{\mathbb{Z}_p^{\oplus g}}\right)_{\tilde{R}} \oplus \mu_{p^{\infty},\tilde{R}}^{\oplus g}$$

$$(4.3.2)$$

Here $\xi_{\tilde{\xi}}$ is given by the matrix $(\frac{1}{\tilde{\xi}} \frac{0}{1})$, see the beginning of Section 4.2 for the matrix notation. To show that this diagram commutes we consider the auxiliary commutative diagram

$$\mu_{p^{\infty},\tilde{R}}^{\oplus g} \oplus \mathbb{Z}[1/p]^{\oplus g} \xrightarrow{\psi_{\tilde{\xi}}} \mu_{p^{\infty},\tilde{R}}^{\oplus g} \oplus \mathbb{Z}[1/p]^{\oplus g}$$

$$\downarrow^{p^{N}} \qquad \qquad \downarrow^{p^{2N}}$$

$$\mu_{p^{\infty},\tilde{R}}^{\oplus g} \oplus \mathbb{Z}[1/p]^{\oplus g} \xrightarrow{p^{N}\psi_{\tilde{\xi}}} \mu_{p^{\infty},\tilde{R}}^{\oplus g} \oplus \mathbb{Z}[1/p]^{\oplus g}$$

$$(4.3.3)$$

The diagram of quasi-isogenies (4.3.2) is obtained from the diagram (4.3.3) by quotienting by the subgroups

$$h_{\underline{q}}(\mathbb{Z}^{\oplus g}) \xrightarrow{\psi_{\underline{\zeta}}} h_{\underline{\zeta}\underline{q}}(\mathbb{Z}^{\oplus g})$$

$$\downarrow^{p^N} \qquad \downarrow^{p^{2N}}$$

$$h_{1}(\mathbb{Z}^{\oplus g}) \xrightarrow{p^{N}\psi_{\underline{\zeta}}} h_{1}(\mathbb{Z}^{\oplus g})$$

and formally inverting certain powers of p. It follows that (4.3.2) is commutative.

5. The formal neighbourhood of an ordinary point

The goal of this section is to give Serre–Tate coordinates for the formal completions of points in the ordinary locus of Shimura varieties of Hodge type.

In Section 5.1 we specialise to the smooth canonical integral models of Shimura varieties of Hodge type at hyperspecial level, and we moreover assume that the ordinary locus is nonempty. In Section 5.2 we recall a small amount of covariant Dieudonné theory for semiperfect rings, following [Scholze and Weinstein 2013].

In Section 5.4 we prove that the formal completion of the ordinary locus gives a subtorus of the Serre–Tate torus, reproving a special case of [Shankar and Zhou 2021, Theorem 1.1]. We also give a group-theoretic description of the Dieudonné module of the associated *p*-divisible group. In Section 5.5 we introduce strongly nontrivial actions of algebraic groups on isocrystals, which we will need to confirm the hypotheses of the rigidity theorem of [Chai 2008].

5.1. Integral models at hyperspecial level. Let the notation be as in Section 2. In particular, we have a Shimura datum (G, X) of Hodge type with reflex field E, a prime p and a prime v of E above p. Moreover there is a symplectic space V and a Hodge embedding $(G, X) \to (\mathcal{G}_V, \mathcal{H}_V)$ and for every sufficiently small $K^p \subset G(\mathbb{A}_f^p)$ there is a sufficiently small $\mathcal{K}^p \subset \mathcal{G}_V(\mathbb{A}_f^p)$ and a finite morphism

$$\mathscr{S}_K := \mathscr{S}_K(G, X) \to \mathscr{S}_K \otimes_{Z_{(p)}} \mathscr{O}_{E, (v)}. \tag{5.1.1}$$

Recall that there is a $\mathbb{Z}_{(p)}$ -lattice $V_{(p)}$ of V on which the symplectic form is $\mathbb{Z}_{(p)}$ -valued, and recall that we have defined $K_p \subset G(\mathbb{Q}_p)$ to be its stabiliser. *From now on we will assume that* K_p *is a hyperspecial subgroup*, in which case \mathscr{S}_K is the canonical integral model of $\mathbf{Sh}_{K^pK_p}(G, X)$ over \mathcal{O}_{E_v} . Moreover the main theorem of [Xu 2020] tells us that the map (5.1.1) is a closed immersion.

5.1.1. The Zariski closure $\mathcal{G}_{\mathbb{Z}_{(p)}}$ of G inside $\mathcal{G}_{V_{(p)}}$ is a reductive group scheme over $\mathbb{Z}_{(p)}$. By [Kisin 2010, Proposition 1.3.2], we can choose tensors $\{s_{\alpha}\}\subset V_{(p)}^{\otimes}$ whose stabiliser is $\mathcal{G}_{\mathbb{Z}_{(p)}}$. All the results of Section 2 still go through with this choice of tensors.

For $x \in \operatorname{Sh}_{G,K^pK_p}(\overline{\mathbb{F}}_p)$ we have seen in Section 2.1.1 that there are tensors

$$\{s_{\alpha,\mathrm{cris}}\}\subset \mathbb{D}_r^{\otimes},$$

where \mathbb{D}_x^{\otimes} is the rational contravariant Dieudonné module of $A_x[p^{\infty}]$. Now let $\mathbb{D}(A_x[p^{\infty}])$ be the *integral* contravariant Dieudonné module. Then as explained in [Shankar and Zhou 2021, Section 6.3], the tensors $\{s_{\alpha, \text{cris}}\}$ lie in

$$\mathbb{D}(A_x[p^{\infty}])^{\otimes}.$$

It is moreover explained there that there is an isomorphism

$$\mathbb{D}(A_x[p^{\infty}]) \simeq V_p \otimes_{\mathbb{Z}_{(p)}} \check{\mathbb{Z}}_p$$

taking $s_{\alpha, \text{cris}}$ to $s_{\alpha} \otimes 1$.

5.1.2. Let us now drop the level from the notation and write \mathscr{S}_G and \mathscr{S}_{GSp} respectively for the base changes of \mathscr{S}_K and \mathscr{S}_K respectively to $\check{\mathbb{Z}}_p$ for some choice of $\mathscr{O}_{E,v} \to \check{\mathbb{Z}}_p$. Similarly write Sh_G for the special fibre of \mathscr{S}_G and Sh_{GSp} for the special fibre of \mathscr{S}_{GSp} . Let $\operatorname{Sh}_{GSp, \operatorname{ord}} \subset \operatorname{Sh}_{GSp}$ be the dense open ordinary locus and define the ordinary locus of Sh_G by $\operatorname{Sh}_{G,\operatorname{ord}} := \operatorname{Sh}_G \cap \operatorname{Sh}_{GSp,\operatorname{ord}}$. It is an open subset which is nonempty if and only if $E_v = \mathbb{Q}_p$, by [Lee 2018, Corollary 1.0.2]. We will assume from now on that $E_v = \mathbb{Q}_p$.

Lemma 5.1.3. The ordinary locus $Sh_{G,ord}$ is open and dense and equal to the Newton stratum $Sh_{G,[b],K^pK_p}$ for $[b] \in B(G, \{\mu^{-1}\})$ the maximal element in the partial order introduced in Section 2.1.2.

The maximal element [b] is known as the μ -ordinary element, and the maximal Newton stratum is known as the μ -ordinary locus.

Proof. The μ -ordinary locus and the ordinary locus are equal in this case by the proof of [Lee 2018, Corollary 4.3.2], as explained in [Lee 2018, Remark 4.3.3]. The density of the μ -ordinary locus is [Wortmann 2013, Theorem 1.1]; see [KMS 2022, Theorem 3] for a published reference.

5.1.4. Let $x \in \operatorname{Sh}_G(\overline{\mathbb{F}}_p)$ be an ordinary point and consider the closed immersions of formal neighbourhoods (considered as functors on the category $\operatorname{Nilp}_{\mathbb{Z}_p}$ of \mathbb{Z}_p -algebras where p is nilpotent)

$$\mathscr{S}_{G}^{/x} := \operatorname{Spf} \widehat{\mathcal{O}}_{\mathscr{S}_{G,x}} \hookrightarrow \operatorname{Spf} \widehat{\mathcal{O}}_{\mathcal{S}_{\operatorname{GSp}},x} =: \mathcal{S}_{\operatorname{GSp}}^{/x}. \tag{5.1.2}$$

Let A be the universal abelian scheme over \mathcal{S}_{GSp} and let $X = A[p^{\infty}]$ be the associated p-divisible group over \mathcal{S}_{GSp} . Let $\widehat{Def}(A_x)$ be the formal deformation space of the abelian variety A_x , that is, the formal scheme representing the functor $Def(A_x)$ on the category Art of deformations of the abelian variety A_x . Similarly let $\widehat{Def}(Y)$ be the deformation space of the p-divisible group $X_x =: Y$. There are natural morphisms

$$\mathcal{S}_{GSp}^{/x} \to \widehat{\operatorname{Def}}(A_x) \to \widehat{\operatorname{Def}}(Y).$$

The first is a closed immersion by the moduli description of S_{GSp} , and the second morphism is an isomorphism by [Katz 1981, Theorem 1.2.1]. Now [Shankar and Zhou 2021, Theorem 1.1] (see [Noot 1996] for closely related results) implies that the closed formal subscheme

$$\mathscr{S}_G^{/x} \subset \widehat{\mathrm{Def}}(Y)$$

is a p-divisible formal subgroup. The goal of this section is to compute the Dieudonné module of $Sh_G^{/x}$. We do this by giving a new proof that

$$\operatorname{Sh}_G^{/x} \subset \widehat{\operatorname{Def}}(Y)$$

is a p-divisible formal subgroup, using the methods of Section 4 and results of [Kim 2019].

5.2. Some covariant Dieudonné theory.

5.2.1. A caveat. In the rest of this subsection we are going to recall some covariant Dieudonné theory for semiperfect rings following [Scholze and Weinstein 2013]. The reason we do this is that the references [Caraiani and Scholze 2017; Kim 2019] are written in this language. Moreover we feel that results such as Lemma 5.2.5 are most naturally stated using the covariant theory.

To avoid potential confusion, we will always write a subscript cov when using covariant Dieudonné theory. The covariant theory and the contravariant theory will interact only once, in Section 5.3, and we will warn the reader again there.

5.2.2. Recall that an \mathbb{F}_p -algebra A is *semiperfect* if it is the quotient of a perfect \mathbb{F}_p -algebra B and that it is *f-semiperfect* if it is the quotient of a perfect \mathbb{F}_p -algebra by a finitely generated ideal. Let A be a semiperfect \mathbb{F}_p -algebra and let $A_{cris}(A)$ be Fontaine's ring of crystalline periods (see [Scholze and Weinstein 2013, Proposition 4.1.3]) with $\varphi: A_{cris}(A) \to A_{cris}(A)$ induced by the absolute Frobenius on A.

Definition 5.2.3. A *covariant Dieudonné module* over a semiperfect \mathbb{F}_p -algebra A is a pair (M, φ_M) , where M is a finite locally free $A_{\text{cris}}(A)$ -module and where

$$\varphi_M: \varphi^*M\left[\frac{1}{p}\right] \to M\left[\frac{1}{p}\right]$$

is an isomorphism such that

$$M \subseteq \varphi_M(M) \subseteq \frac{1}{p}M$$
.

Remark 5.2.4. Usually one instead asks that

$$pM \subseteq \varphi_M(M) \subseteq M$$
.

The reasons for our conventions is that they agree with the conventions in [Caraiani and Scholze 2017; Kim 2019].

A *p*-divisible group \mathscr{G} over A has a covariant⁵ Dieudonné module ($\mathbb{D}_{cov}(\mathscr{G})$, $\varphi_{\mathscr{G}}$). For Spec $A' \to \operatorname{Spec} A$ there is a canonical isomorphism

$$(\mathbb{D}_{\text{cov}}(\mathscr{G}_{A'}), \varphi_{\mathscr{G}_{A'}}) \simeq (\mathbb{D}_{\text{cov}}(\mathscr{G}), \varphi_{\mathscr{G}}) \otimes_{A_{\text{cris}}(A)} A_{\text{cris}}(A').$$

Our covariant Dieudonné modules are normalised as in [Caraiani and Scholze 2017]. In particular, this means that the covariant Dieudonné module of $\mathbb{Q}_p/\mathbb{Z}_p$ over A is $A_{\text{cris}}(A)$ equipped with the trivial Frobenius, and the covariant Dieudonné module of $\mu_{p^{\infty}}$ is $A_{\text{cris}}(A)$ equipped with Frobenius given by 1/p. This also means that the contravariant Dieudonné module is isomorphic to the dual of the covariant Dieudonné module, see [Caraiani and Scholze 2017, footnote on page 692].

Now let \mathscr{G} be a p-divisible group over \mathbb{F}_p with universal cover $\widetilde{\mathscr{G}}$ in the sense of [Scholze and Weinstein 2013, Section 3.1]. If we consider $\widetilde{\mathscr{G}}$ as a functor on Nilp then it is a filtered colimit of spectra of f-semiperfect \mathbb{F}_p -algebras by [Scholze and Weinstein 2013, Proposition 3.1.3(iii)] and is thus determined by its restriction to the category of semiperfect \mathbb{F}_p -algebras. We can describe it explicitly on the category of f-semiperfect \mathbb{F}_p -algebras as follows:

Lemma 5.2.5. There is a commutative diagram of natural transformation of functors on the category of f-semiperfect \mathbb{F}_p -algebras, which evaluated at an object A gives

$$\widetilde{\mathscr{G}}(A) \stackrel{\simeq}{\longrightarrow} \left(B_{\mathrm{cris}}^+(A) \otimes_{\check{\mathbb{Q}}_p} \mathbb{D}_{\mathrm{cov}}(\mathscr{G}) \left[\frac{1}{p} \right] \right)^{\varphi=1},$$

where φ is given by the diagonal Frobenius and where $B_{\text{cris}}^+(A) := A_{\text{cris}}(A)[1/p]$.

⁵We write \mathbb{D}_{cov} to distinguish from the contravariant Dieudonné theory that we used in Section 3.

Proof. Let *A* be f-semiperfect, then [Scholze and Weinstein 2013, Theorem 4.1.4] tells us that the covariant Dieudonné module functor over *A* is fully faithful after inverting *p*. There is a natural map

$$T_{p}\mathscr{G}(A) = \operatorname{Hom}_{A}((\mathbb{Q}_{p}/\mathbb{Z}_{p})_{A}, \mathscr{G}_{A}) \to \operatorname{Hom}_{A_{\operatorname{cris}}, F} \left(A_{\operatorname{cris}}(A), A_{\operatorname{cris}}(A) \otimes_{\check{\mathbb{Z}}_{p}} \mathbb{D}_{\operatorname{cov}}(\mathscr{G}) \right)$$
$$\simeq \left(A_{\operatorname{cris}}(A) \otimes_{\check{\mathbb{Z}}_{p}} \mathbb{D}_{\operatorname{cov}}(\mathscr{G}) \right)^{\varphi = 1},$$

where the latter bijection is induced by evaluation at 1. After inverting p we get a natural isomorphism

$$\widetilde{\mathscr{G}}(A) = \operatorname{Hom}_{A}((\mathbb{Q}_{p}/\mathbb{Z}_{p})_{A}, \mathscr{G}_{A})\left[\frac{1}{p}\right] \to \left(B_{\operatorname{cris}}^{+}(A) \otimes_{\check{\mathbb{Q}}_{p}} \mathbb{D}_{\operatorname{cov}}(\mathscr{G})\left[\frac{1}{p}\right]\right)^{\varphi = 1}.$$

5.3. The Dieudonné module of the Serre-Tate torus. Let $x \in \operatorname{Sh}_{G,\operatorname{ord}}(\overline{\mathbb{F}}_p)$ be as above and let $Y = A_x[p^\infty]$ be the corresponding p-divisible group. Recall from Section 4 that $Y = Y_0 \oplus Y_1$ and that both Y_0 and Y_1 lift uniquely to p-divisible groups Y_0^{can} and Y_1^{can} over \mathbb{Z}_p . Let $\operatorname{Def}(Y)$ be the formal deformation space of Y, considered as a functor on Art together with its extension $\widehat{\operatorname{Def}}(Y)$ to Nilp. We have seen that $\widehat{\operatorname{Def}}(Y)$ has the structure of a p-divisible formal group, and we use $\mathcal{H}_{0,1}$ to denote the corresponding p-divisible group over $\operatorname{Spf}\mathbb{Z}_p$.

Consider the special fibre $\mathcal{H}_{0,1,\overline{\mathbb{F}}_p}$. Then by Lemma 4.1.2 its *p*-adic Tate module is given by $\mathscr{H}om(Y_0,Y_1)$. Therefore by [Caraiani and Scholze 2017, Lemmas 4.1.7 and 4.1.8], we have an isomorphism

$$\mathbb{D}_{\text{cov}}(\mathcal{H}_{0,1,\overline{\mathbb{F}}_p})[1/p] \simeq \mathscr{H}om(\mathbb{D}_{\text{cov}}(Y_0)[1/p], \mathbb{D}_{\text{cov}}(Y_1)[1/p])^{\leq 0},$$

where $\mathscr{H}om$ denotes the internal hom in F-isocrystals and where $(\cdot)^{\leq 0}$ denotes the slope at most 0 part of an F-isocrystal.

5.3.1. Choose an isomorphism (here we use contravariant Dieudonné theory!) $\mathbb{D}(Y) \to V_p \otimes_{\mathbb{Z}_p} \check{\mathbb{Z}}_p$ sending $s_\alpha \otimes 1$ to $s_{x, \text{cris}}$ as in Section 5.1.1. This induces an isomorphism from $V_p^* \otimes_{\mathbb{Z}_p} \check{\mathbb{Z}}_p$ to the covariant Dieudonné module $\mathbb{D}_{\text{cov}}(Y)$ and thus gives us Frobenius invariant tensors $\{s_{\alpha, \text{cris}}\} \subset \mathbb{D}_{\text{cov}}(Y)^{\otimes}$. Let $b \in G(\check{\mathbb{Q}}_p) \subset GL(V^*)(\check{\mathbb{Q}}_p)$ be the element corresponding to the Frobenius in $\mathbb{D}_{\text{cov}}(Y)[1/p]$. Then there is an inclusion of *F*-isocrystals

$$\mathscr{H}om(\mathbb{D}_{cov}(Y_0)[1/p], \mathbb{D}_{cov}(Y_1)[1/p]) \subset \mathscr{H}om(\mathbb{D}_{cov}(Y)[1/p], \mathbb{D}_{cov}(Y)[1/p]), \tag{5.3.1}$$

which sends $f: \mathbb{D}_{cov}(Y_0)[1/p] \to \mathbb{D}_{cov}(Y_1)[1/p]$ to

$$\mathrm{Id} + f: \mathbb{D}_{\mathrm{cov}}(Y_0)[1/p] \oplus \mathbb{D}_{\mathrm{cov}}(Y_1)[1/p] \to \mathbb{D}_{\mathrm{cov}}(Y_0)[1/p] \oplus \mathbb{D}_{\mathrm{cov}}(Y_1)[1/p].$$

The map in equation (5.3.1) realises the source as the slope -1 part of the target.

5.3.2. Write $\mathfrak{gl}(V^*)$ for the Lie algebra of the algebraic group $GL(V^*) \otimes \check{\mathbb{Q}}_p$ and identify it with the vector space of endomorphisms of $V^* \otimes \check{\mathbb{Q}}_p$ equipped with the commutator bracket. We can equip $\mathfrak{gl}(V^*)$ with the structure of an F-isocrystal by letting Frobenius act by conjugation by $b \in GL(V^*)(\check{\mathbb{Q}}_p)$. Let us write $(\mathfrak{gl}(V^*), \operatorname{Ad} b\sigma)$ for this isocrystal.

Using the isomorphism $V_p^* \otimes_{\mathbb{Z}_p} \check{\mathbb{Z}}_p \simeq \mathbb{D}_{cov}(Y)$ as above, we can identify the *F*-isocrystal on the right hand side of (5.3.1) with $(\mathfrak{gl}(V^*), \operatorname{Ad} b\sigma)$. There is a sub-*F*-isocrystal

$$(\mathfrak{g}, \operatorname{Ad} b\sigma) \subset (\mathfrak{gl}(V^*), \operatorname{Ad} b\sigma),$$

where $\mathfrak{g} = \operatorname{Lie} G \otimes \check{\mathbb{Q}}_p$. By Lemma 5.3.3 below, the subspace $\mathfrak{g} \subset \mathfrak{gl}(V^*)$ is precisely the subspace of those endomorphisms g of $V^* \otimes \check{\mathbb{Q}}_p$ that satisfy $g^*(s_\alpha \otimes 1) = 0$ for all tensors s_α .

Lemma 5.3.3. Let C be a field of characteristic zero and let W be a finite dimensional C vector space. Let $H \subset GL(W)$ be a connected reductive group that is the stabiliser of a collection of tensors $\{t_{\alpha}\}_{{\alpha}\in\mathscr{A}}\subset W^{\otimes}$. Then the Lie algebra $\mathfrak{h}\subset\mathfrak{gl}(W)$ is given by the subspace

$$\{H \in \mathfrak{gl}(W) : H^*(t_\alpha) = 0 \text{ for all } \alpha \in \mathscr{A}\}.$$

Proof. The Lie algebra is given by the kernel of the map $G(C[\epsilon]/(\epsilon^2)) \to G(C)$. Thus it consists of matrices of the form $1 + \epsilon M$, where $M \in \mathfrak{gl}(W)$, such that for $\alpha \in \mathscr{A}$ we have

$$(1 + \epsilon M)^*(t_{\alpha} \otimes 1) = t_{\alpha}.$$

But this is equivalent to $(\epsilon M)^*(t_\alpha \otimes 1) = 0$ or $M^*(t_\alpha) = 0$.

Let us write

$$(\mathfrak{g}, \operatorname{Ad} b\sigma)^{-1} \subset \mathscr{H}om(\mathbb{D}_{cov}(Y_0)[1/p], \mathbb{D}_{cov}(Y_1)[1/p]) = \mathbb{D}_{cov}(\mathcal{H}_{0,1,\overline{\mathbb{F}}_p})[1/p]$$
(5.3.2)

for the slope -1 subspace of the *F*-isocrystal (\mathfrak{g} , Ad $b\sigma$). Then by [Kim 2019, Lemma 3.1.3] and its proof, there is an inclusion of *p*-divisible groups

$$\mathcal{H}_{0,1,\overline{\mathbb{F}}_p}^G \subset \mathcal{H}_{0,1,\overline{\mathbb{F}}_p} \tag{5.3.3}$$

inducing (5.3.2) upon taking rational covariant Dieudonné modules. Since both of these p-divisible groups have étale Serre duals, there is a unique lift $\mathcal{H}_{0,1}^G$ of $\mathcal{H}_{0,1,\overline{\mathbb{F}}_p}^G$ to $\check{\mathbb{Z}}_p$ and a unique lift

$$\mathcal{H}_{0,1}^G \subset \mathcal{H}_{0,1}$$

of the inclusion (5.3.3).

Lemma 5.3.4. Let A be an f-semiperfect \mathbb{F}_p -algebra. Then the inclusion

$$\widetilde{\mathcal{H}}_{0,1,\overline{\mathbb{F}}_p}^G(A)\subset\widetilde{\mathcal{H}}_{0,1,\overline{\mathbb{F}}_p}(A)=\mathscr{H}om(Y_{0,A},Y_{1,A})[1/p]$$

identifies $\widetilde{\mathcal{H}}_{0,1,\overline{\mathbb{F}}_p}^G(A)$ with the subspace of those quasi-endomorphisms $f:Y_{0,A}\dashrightarrow Y_{1,A}$ such that the induced quasi-endomorphism

$$g = \begin{pmatrix} 0 & 0 \\ f & 0 \end{pmatrix} : Y_A \dashrightarrow Y_A$$

induces an endomorphism $\mathbb{D}_{\text{cov}}(Y_A)[1/p] \to \mathbb{D}_{\text{cov}}(Y_A)[1/p]$ satisfying $g^*(s_{\alpha,\text{cris}} \otimes 1) = 0$.

Proof. This follows from Lemma 5.2.5 in combination with Lemma 5.3.3.

Remark 5.3.5. The statement of Lemma 5.3.4 contradicts [Kim 2019, Lemma 3.1.3], which implies that the inclusion

$$\widetilde{\mathcal{H}}_{0,1,\overline{\mathbb{F}}_p}^G(A) \subset \widetilde{\mathcal{H}}_{0,1,\overline{\mathbb{F}}_p}(A) = \mathscr{H}om(Y_{0,A},Y_{1,A})[1/p]$$

identifies $\widetilde{\mathcal{H}}_{0,1,\overline{\mathbb{F}}_p}^G(A)$ with the subspace of those quasi-endomorphisms $f:Y_{0,A}\dashrightarrow Y_{1,A}$ such that the induced quasi-endomorphism

$$g = \begin{pmatrix} 0 & 0 \\ f & 0 \end{pmatrix} : Y_A \dashrightarrow Y_A$$

induces an endomorphism $\mathbb{D}_{\text{cov}}(Y_A)[1/p] \to \mathbb{D}_{\text{cov}}(Y_A)[1/p]$ satisfying $g^*(s_{\alpha,\text{cris}} \otimes 1) = s_{\alpha,\text{cris}} \otimes 1$. This cannot be correct because $\widetilde{\mathcal{H}}_{0,1,\overline{\mathbb{F}}_p}^G(A)$ is stable under addition and if g_1, g_2 both satisfy $g^*(s_{\alpha,\text{cris}} \otimes 1) = s_{\alpha,\text{cris}} \otimes 1$ then their sum $g_1 + g_2$ does not.

The following lemma and its corollary essentially follow from [Kim 2019, Proposition 3.2.4]. However the construction there is incorrect because of the error in [Kim 2019, Lemma 3.1.3] pointed out above. Once the subgroup in the statement of Lemma 5.3.6 has been shown to exist with the properties proved in Corollary 5.3.7, the rest of the arguments in [Kim 2019] go through without further changes.

Lemma 5.3.6. There is a closed subgroup

$$\operatorname{Aut}_G(\widetilde{Y}) \subset \operatorname{Aut}(\widetilde{Y})$$

such that on A-points for f-semiperfect $\overline{\mathbb{F}}_p$ -algebras A, it is the subgroup of those quasi-isogenies $g: Y_A \dashrightarrow Y_A$ that induce isomorphisms $g: \mathbb{D}_{cov}(Y_A)[1/p] \to \mathbb{D}_{cov}(Y_A)[1/p]$ satisfying

$$g^*(s_{\alpha, \text{cris}} \otimes 1) = s_{\alpha, \text{cris}} \otimes 1.$$

We will call such quasi-isogenies tensor-preserving quasi-isogenies.

Proof. First of all by [Caraiani and Scholze 2017, Lemma 4.2.10] the functor $\operatorname{Aut}(\widetilde{Y})$ satisfies

$$\operatorname{Aut}(\widetilde{Y})(R) = \operatorname{Aut}(\widetilde{Y})(R/p),$$

for all $R \in \text{Nilp}$. Thus we can define a closed subfunctor of $\mathbf{Aut}(\widetilde{Y})$ by specifying its values on $\overline{\mathbb{F}}_p$ -algebras. The matrix description of $\mathbf{Aut}(\widetilde{Y})$ in Section 4.2 gives us a semidirect product decomposition (see [Caraiani and Scholze 2017, Proposition 4.2.11, Remark 4.2.12])

$$\mathbf{Aut}(\widetilde{Y})_{\overline{\mathbb{F}}_p} := \mathscr{H}om(Y_0, Y_1)[1/p] \rtimes \big(\mathbf{Aut}(\widetilde{Y}_0)_{\overline{\mathbb{F}}_p} \times \mathbf{Aut}(\widetilde{Y}_1)_{\overline{\mathbb{F}}_p}\big).$$

Here we are using the map

$$\mathscr{H}om(Y_0, Y_1)[1/p] \to \operatorname{Aut}(\widetilde{Y})_{\overline{\mathbb{F}}_p}, \quad f \mapsto \begin{pmatrix} 1 & 0 \\ f & 1 \end{pmatrix}$$

to realise $\mathscr{H}om(Y_0,Y_1)[1/p]$ as the subgroup of lower triangular automorphisms of \widetilde{Y} . The condition that $\binom{1}{f}\binom{0}{1} = 1 + f$ satisfies $(1+f)^*(s_{\alpha,\mathrm{cris}}\otimes 1) = s_{\alpha,\mathrm{cris}}\otimes 1$ is equivalent to the condition that $f^*(s_{\alpha,\mathrm{cris}}\otimes 1) = 0$. Thus we see that the intersection $\mathscr{H}om(Y_0,Y_1)[1/p]$ with $\mathbf{Aut}_G(\widetilde{Y})$ is given by

$$\widetilde{\mathcal{H}}_{0,1,\overline{\mathbb{F}}_p}^G \subset \widetilde{\mathcal{H}}_{0,1,\overline{\mathbb{F}}_p} = \mathscr{H}om(Y_0,Y_1)[1/p].$$

By Lemma 5.3.4, this is representable by a closed subgroup.

We can identify the group $(\operatorname{Aut}(\tilde{Y}_1)_{\overline{\mathbb{F}}_p} \times \operatorname{Aut}(\tilde{Y}_0)_{\overline{\mathbb{F}}_p})$ with the locally profinite group scheme associated to the locally profinite group $\operatorname{Aut}(\tilde{Y})(\overline{\mathbb{F}}_p)$. Using Dieudonné theory, we can identify this locally profinite group with the σ -centraliser of b in $\operatorname{GL}(V^*)(\check{\mathbb{Q}}_p)$, where we recall that we have fixed an isomorphism $V_p^* \otimes_{\mathbb{Z}_p} \check{\mathbb{Z}}_p \simeq \mathbb{D}_{\operatorname{cov}}(Y)$ giving rise to $b \in G(\check{\mathbb{Q}}_p)$. The subgroup of tensor-preserving automorphisms of \widetilde{Y} over $\overline{\mathbb{F}}_p$ can be identified with $J_b(\mathbb{Q}_p)$, the σ -centraliser of $b \in G(\check{\mathbb{Q}}_p)$, which is a closed subgroup.

Note that $J_b(\mathbb{Q}_p) \subset G(\check{\mathbb{Q}}_p)$ stabilises $(\mathfrak{g}, \operatorname{Ad} b\sigma)^{-1}$ because it acts on \mathfrak{g} via automorphisms that preserve the slope decomposition. Using Lemma 5.3.4 we see that the closed subgroup

$$\widetilde{\mathcal{H}}_{0,1,\overline{\mathbb{F}}_p}^G \rtimes \underline{J_b(\mathbb{Q}_p)} \subset \operatorname{Aut}(\widetilde{Y})_{\overline{\mathbb{F}}_p},$$

has the required properties over $\overline{\mathbb{F}}_p$, and so we are done.

Since the *R*-points of $\widetilde{\mathcal{H}}_{0,1}^G$ and $J_b(\mathbb{Q}_p)$ both only depend on R/p, we see that

$$\widetilde{\mathcal{H}}_{0,1}^G \rtimes \underline{J_b(\mathbb{Q}_p)} = \mathbf{Aut}_G(\widetilde{Y})$$

describes the unique lift to $\check{\mathbb{Z}}_p$. This identifies $\widetilde{\mathcal{H}}_{0,1}^G$ with the neutral component $\mathbf{Aut}_G(\widetilde{Y})^\circ$ of $\mathbf{Aut}_G(\widetilde{Y})$.

Corollary 5.3.7. The identity component

$$\widetilde{\mathcal{H}}_{0,1,\overline{\mathbb{F}}_n}^G = \operatorname{Aut}_G(\widetilde{Y})^\circ \subset \operatorname{Aut}_G(\widetilde{Y})$$

is isomorphic to Spf S where S is the p-adic completion of $\check{\mathbb{Z}}_p[\![x_1^{1/p^{\infty}},\ldots,x_d^{1/p^{\infty}}]\!]$ for some d.

Proof. This is true for $\widetilde{\mathcal{H}}_{0,1,\overline{\mathbb{F}}_p}^G$ because it is the universal cover of a p-divisible group, see [Scholze and Weinstein 2013, Corollary 3.1.5, Section 6.4].

5.4. Serre–Tate coordinates for Hodge type Shimura varieties. Recall that $x \in \operatorname{Sh}_G(\overline{\mathbb{F}}_p)$ is an ordinary point with associated element $b = b_x \in G(\overline{\mathbb{Q}}_p)$. Recall also from Section 2.1.1 that we have a $G(\overline{\mathbb{Q}}_p)$ conjugacy class of cocharacters $\{\mu\}$ coming from the Shimura datum X and the fixed place v of E.

Lemma 5.4.1. The conjugacy class of fractional cocharacters $\{v_{[b]}\}$ defined by [b] is equal to $\{\mu^{-1}\}$.

Proof. The ordinary locus is equal to the μ -ordinary locus by Lemma 5.1.3. Therefore we have that $\{\nu_{[b]}\}=\{\bar{\mu}\}$, where $\{\bar{\mu}\}$ is the Galois-average of $\{\mu^{-1}\}$, see [Shankar and Zhou 2021, Section 2.1]. But since $G_{\mathbb{Q}_p}$ is unramified and the local reflex field E_v is equal to \mathbb{Q}_p , there is a cocharacter μ defined over \mathbb{Q}_p inducing the conjugacy class of cocharacters $\{\mu\}$. It follows that $\{\bar{\mu}\}=\{\mu^{-1}\}$.

Let $\{\lambda\}$ be a conjugacy class of (fractional) cocharacters of a connected reductive group H over an algebraically closed field C. Let T be a maximal torus, let λ be a representative of $\{\lambda\}$ factoring through T and let $B \supset T$ be a Borel. Let $\rho \in X^*(T)$ be the half sum of the positive roots with respect to B. Then the pairing $\langle 2\rho, \lambda \rangle$ does not depend on the choice of T, B or λ , and we denote it by $\langle 2\rho, \{\lambda\} \rangle$.

Corollary 5.4.2. The p-divisible formal group $\mathcal{H}_{0,1,\overline{\mathbb{F}}_n}^G$ has dimension $\langle 2\rho, \{\mu\} \rangle$

Proof. The dimension of $\mathcal{H}_{0,1,\overline{\mathbb{F}}_p}^G$ is equal to $\langle 2\rho, \nu_{[b]} \rangle$ by [Kim 2019, Proposition 3.1.4], which is equal to $\langle 2\rho, \{\mu^{-1}\} \rangle = \langle 2\rho, \{\mu\} \rangle$ by Lemma 5.4.1.

Proposition 5.4.3. The closed formal subscheme

$$\mathscr{S}_{G}^{/x} \hookrightarrow \mathcal{S}_{\mathrm{GSp}}^{/x} \hookrightarrow \widehat{\mathrm{Def}}(Y)$$

introduced in (5.1.2), is a p-divisible formal subgroup. The induced inclusion of p-divisible groups

$$\mathscr{S}_{G}^{/x}[p^{\infty}] \subset \widehat{\mathrm{Def}}(Y)[p^{\infty}] = \mathcal{H}_{0,1}$$

induces the inclusion

$$(\mathfrak{g}, \operatorname{Ad} b\sigma)^{-1} \subset \mathscr{H}om(\mathbb{D}_{cov}(Y_0)[1/p], \mathbb{D}_{cov}(Y_1)[1/p])$$

from (5.3.2) on rational covariant Dieudonné modules of their special fibres.

Proof. By [Kim 2019, Theorem 4.3.1], the closed formal subscheme $\mathscr{S}_G^{/x} \subset \widehat{\operatorname{Def}}(Y)$ is stable under the action of

$$\operatorname{Aut}_G(\widetilde{Y})^{\circ} \subset \operatorname{Aut}(\widetilde{Y})^{\circ}$$
.

We can identify these groups with

$$\widetilde{\mathcal{H}}_{0,1}^G \subset \widetilde{\mathcal{H}}_{0,1} \tag{5.4.1}$$

By Proposition 4.2.5, the action of $\widetilde{\mathcal{H}}_{0,1}$ on $\widehat{\mathrm{Def}}(Y)$ factors through the natural action of $\mathcal{H}_{0,1}$ on $\widehat{\mathrm{Def}}(Y)$ by left translation, via the natural quotient map

$$\widetilde{\mathcal{H}}_{0,1} \to \mathcal{H}_{0,1}$$
.

The inclusion $\mathcal{H}_{0,1}^G \subset \mathcal{H}_{0,1}$ induces an inclusion $T_p \mathcal{H}_{0,1}^G \subset T_p \mathcal{H}_{0,1}$ which induces (5.4.1) after inverting p. This implies that the action of $\widetilde{\mathcal{H}}_{0,1}^G$ on $\mathscr{S}_G^{/x}$ factors through an action of $\mathcal{H}_{0,1}^G$ via the natural quotient map

$$\widetilde{\mathcal{H}}_{0,1}^G \to \mathcal{H}_{0,1}^G$$
.

Now consider the closed point $\{x\} \in Sh_G^{/x}$. Then the associated orbit map gives a closed immersion

$$\mathcal{H}_{0,1,\overline{\mathbb{F}}_p} \hookrightarrow \mathrm{Def}(Y)_{\overline{\mathbb{F}}_p}.$$

This means that we similarly get a closed immersion

$$\mathcal{H}_{0,1,\overline{\mathbb{F}}_p}^G\subset \operatorname{Sh}_G^{/x}$$
.

By [Kim 2019, Proposition 3.1.4], the formal scheme $\operatorname{Def}(Y)_{G,\overline{\mathbb{F}}_p}$ has dimension $\langle 2\rho, \{\nu_{[b]}\}\rangle$, which is equal to $\langle 2\rho, \{\mu\}\rangle$ by Lemma 5.4.1, which in turn is equal to the dimension of $\operatorname{Sh}_G^{/x}$. It follows that the orbit map induces an isomorphism

$$\mathcal{H}_{0,1,\overline{\mathbb{F}}_p}^G \to \operatorname{Sh}_G^{/x}$$

and that $\operatorname{Sh}_G^{/x}$ is a formal subgroup of $\operatorname{Def}(Y)_{\overline{\mathbb{F}}_p}$ satisfying the conclusions of the proposition. It remains to show that $\mathscr{S}_G^{/x} \subset \operatorname{Def}(Y)$ is a formal subgroup, which follows from [Shankar and Zhou 2021, Theorem 1.1].

5.4.4. The action of automorphism groups. Let the notation be as in Section 5. Recall that we have fixed an isomorphism $\mathbb{D}_{cov}(Y) \simeq V_p^* \otimes \check{\mathbb{Z}}_p$ sending $s_\alpha \otimes 1$ to $s_{\alpha, cris}$. This gives us an element $b \in G(\check{\mathbb{Q}}_p) \subset GL(V^*)(\check{\mathbb{Q}}_p)$ corresponding to the Frobenius in $\mathbb{D}_{cov}(Y)[1/p]$.

Recall from Section 4 that there is an action of $\operatorname{Aut}(\widetilde{Y})$ on RZ_Y . Recall from the discussion before Corollary 4.2.3, that $\operatorname{\underline{Aut}(Y)(\overline{\mathbb{F}}_p)} \subset \operatorname{Aut}(\widetilde{Y})$ preserves the $\overline{\mathbb{F}}_p$ -point $y \in \operatorname{RZ}_Y(\overline{\mathbb{F}}_p)$ corresponding to the identity map of Y, and that this induces an action of the profinite group $\operatorname{Aut}(Y)(\overline{\mathbb{F}}_p)$ on $\operatorname{Def}(Y)$. This action is described in Corollary 4.2.3.

5.4.5. Recall that there are closed immersions of topological groups

$$J_b(\mathbb{Q}_p) & \longrightarrow G(\check{\mathbb{Q}}_p)$$

$$\downarrow \qquad \qquad \downarrow$$

$$GL(V^*)_b(\mathbb{Q}_p) & \longrightarrow GL(V^*)(\check{\mathbb{Q}}_p)$$

where $GL(V^*)_b(\mathbb{Q}_p) = \operatorname{Aut}(\widetilde{Y})(\overline{\mathbb{F}}_p)$ is the σ -centraliser of b in $GL(V^*)(\check{\mathbb{Q}}_p)$. Let us write $U_p \subset J_b(\mathbb{Q}_p)$ for the compact open subgroup given by the intersection

$$\operatorname{Aut}_G(\widetilde{Y})(\overline{\mathbb{F}}_p) \cap \operatorname{Aut}(Y)(\overline{\mathbb{F}}_p).$$

Then U_p acts on $\widetilde{\mathcal{H}}_{0,1}^G \subset \mathscr{H}om(Y_0,Y_1)[1/p]$ and preserves the action of $T_p\mathcal{H}_{0,1}^G$, and thus acts on the quotient $\mathcal{H}_{0,1} \simeq \mathscr{S}_G^{/x}$. By Proposition 5.4.3 and the proof of Lemma 5.3.6, the induced action on rational Dieudonné modules can be identified with the natural action of $U_p \subset J_b(\mathbb{Q}_p)$ on

$$(\mathfrak{g}, \operatorname{Ad} b\sigma)^{-1} \subset (\mathfrak{g}, \operatorname{Ad} b\sigma).$$

In order to apply the rigidity result of Chai [2008] we need to understand this action. We will do this in more generality in the next section.

5.5. Strongly nontrivial actions. Let G be a connected reductive group over \mathbb{Q}_p . Let $b \in G(\mathring{\mathbb{Q}}_p)$ be an element and consider the F-isocrystal $(\mathfrak{g}, \operatorname{Ad} b\sigma)$, where $\mathfrak{g} = \operatorname{Lie} G \otimes \mathring{\mathbb{Q}}_p$ equipped with its action of $J_b(\mathbb{Q}_p)$. If we replace b by a σ -conjugate b', then $J_b(\mathbb{Q}_p)$ and $J_{b'}(\mathbb{Q}_p)$ are conjugate in $G(\mathring{\mathbb{Q}}_p)$, and there is an isomorphism of isocrystals $(\mathfrak{g}, \operatorname{Ad} b\sigma) \simeq (\mathfrak{g}, \operatorname{Ad} b'\sigma)$.

Let $\lambda \in \mathbb{Q}$ and let $N_{\lambda} \subset (\mathfrak{g}, \operatorname{Ad} b\sigma)$ be the largest sub-*F*-isocrystal of slope λ . Then because $J_b(\mathbb{Q}_p)$ acts on $(\mathfrak{g}, \operatorname{Ad} b\sigma)$ via *F*-isocrystal automorphisms, it preserves the subspace N_{λ} . Let us also denote by *b* the image of *b* in $GL(\mathfrak{g})$, then there is a homomorphism of algebraic groups

$$J_b \to \mathrm{GL}(\mathfrak{g})_b$$
,

where $GL(\mathfrak{g})_b$ denotes the σ -centraliser of b in $GL(\mathfrak{g})$. There is a parabolic subgroup

$$P(\lambda) \subset GL(\mathfrak{g})$$

consisting of automorphisms of the $\tilde{\mathbb{Q}}_p$ -vector space \mathfrak{g} that preserve the slope filtration on the *F*-isocrystal $(\mathfrak{g}, \operatorname{Ad} b\sigma)$, and after potentially replacing b by a σ -conjugate, the image of b lands in $P(\lambda)$. There is

thus a group homomorphism

$$J_b \to P(\lambda)_b$$

where $P(\lambda)_b$ denotes the σ -centraliser of b in $P(\lambda)$. Since N_{λ} is a graded quotient of the slope filtration of the F-isocrystal $(\mathfrak{g}, \operatorname{Ad} b\sigma)$, there is an induced quotient map $P(\lambda) \to \operatorname{GL}(N_{\lambda})$ and this induces a group homomorphism

$$J_b \to \operatorname{GL}(N_\lambda)_b$$
,

where $GL(N_{\lambda})_b$ denotes the σ -centraliser of b in $GL(N_{\lambda})$. Let E be the \mathbb{Q}_p -algebra of endomorphisms of the F-isocrystal N_{λ} and let E^{\times} be the functor on \mathbb{Q}_p -algebras given by $R \mapsto (R \otimes E)^{\times}$. Then there is a natural isomorphism $E^{\times} \simeq GL(N_{\lambda})_b$.

Let GL(E) be the general linear group of E considered as a \mathbb{Q}_p -vector space and let $E^{\times} \to GL(E)$ be the natural map corresponding to the action of E on itself by left translation. Consider E as a \mathbb{Q}_p -linear representation of J_b via $J_b \to E^{\times}$, then the goal of this section is to prove the following result:

Proposition 5.5.1. Let $T \subset J_b$ be a maximal torus. If $\lambda \neq 0$, then the trivial representation of T does not occur in the representation of T given by E.

Proof of Proposition 5.5.1. After replacing b by a σ -conjugate we can arrange for it to satisfy (see [Kottwitz 1985, Section 4])

$$b\sigma(b)\cdots\sigma^{r-1}(b)=(r\nu_b)(p)$$

for some r. Here v_b is the Newton cocharacter of b, which is defined over \mathbb{Q}_{p^r} . Let $M_{v_b} \subset G \otimes \check{\mathbb{Q}}_p$ denote the centraliser of the cocharacter v_b . By [Kim 2019, Proposition 2.2.6], there is a unique isomorphism

$$J_b \otimes \check{\mathbb{Q}}_p \to M_{\nu_b}$$

such the composition $J_b(\mathbb{Q}_p) \subset J_b(\check{\mathbb{Q}}_p) \to M_{\nu_b}(\check{\mathbb{Q}}_p) \subset G(\check{\mathbb{Q}}_p)$ is the defining inclusion of $J_b(\mathbb{Q}_p)$ as the σ -centraliser of b in $G(\check{\mathbb{Q}}_p)$.

After tensoring up to $\check{\mathbb{Q}}_p$, there is a commutative diagram, where L^{reg} is the left regular representation of $GL(N_{\lambda})$ on $GL(\text{End}(N_{\lambda}))$,

$$GL(E_{\tilde{\mathbb{Q}}_p}) \xrightarrow{\simeq} GL(End(N_{\lambda}))$$

$$\uparrow \qquad \qquad \downarrow_{\operatorname{reg}} \qquad \downarrow_{\operatorname{reg}$$

If we show that the trivial representation of $T \otimes \check{\mathbb{Q}}_p$ does not occur in $E \otimes \check{\mathbb{Q}}_p$, then it follows that the trivial representation of T does not occur in E. The representation $W = \operatorname{End}(N_{\lambda})$ of $J_{b,\check{\mathbb{Q}}_p}$ defined by composition with the left regular representation is a direct sum of copies of the representation N_{λ} . Therefore it suffices to show that the representation N_{λ} of $T \otimes \check{\mathbb{Q}}_p$ does not contain the trivial representation.

We note that $T \otimes \check{\mathbb{Q}}_p =: T'$ is a maximal torus of M_{ν_b} acting on the associated graded of the slope filtration of the *F*-isocrystal $(\mathfrak{g}, \operatorname{Ad} b\sigma)$. Since ν_b is a central cocharacter of M_{ν_b} by definition, we see that $(r\nu_b)(p) \in T'(\check{\mathbb{Q}}_p)$. To determine the slope decomposition of the *F*-isocrystal $(\mathfrak{g}, \operatorname{Ad} b\sigma)$, it suffices to determine the slope decomposition of the *F*^r-isocrystal

$$(\mathfrak{g}, (\operatorname{Ad} b\sigma)^r)$$

for some positive integer r.

Let C be an algebraic closure of $\check{\mathbb{Q}}_p$ and consider the action of T'_C on \mathfrak{g}_C via the adjoint action of G_C . Then we have a decomposition

$$\mathfrak{g}_C \simeq \mathfrak{t}_C' \oplus \left(\bigoplus_{lpha \in \Phi} U_lpha
ight),$$

where $\Phi \subset X^*(T'_C)$ consists of the simple roots of T_C . There is a similar decomposition

$$\mathfrak{g} \simeq \mathfrak{t}' \oplus \left(\bigoplus_{\alpha_0 \in \Phi_0} U_{\alpha_0}\right),$$
 (5.5.1)

where $\Phi_0 \in X^*(T'_C)_I$ is the image of Φ and where $I = \operatorname{Gal}(C/\tilde{\mathbb{Q}}_p)$ is the inertia group.

Now we choose an integer r with the following properties: the isomorphism $J_b \otimes \tilde{\mathbb{Q}}_p \to M_{\nu_b}$ is defined over \mathbb{Q}_{p^r} , the equation

$$b\sigma(b)\cdots\sigma^{r-1}(b)=(r\nu_b)(p)$$

is satisfied, and the decomposition (5.5.1) is defined over \mathbb{Q}_{p^r} . Then each U_{α_0} is stable under the action of σ^r and $(\mathrm{Ad}\,b\sigma)^r$ acts on it by $(r\nu_b)(p)\sigma^r$. The operator $\mathrm{Ad}\,b\sigma$ moreover acts trivially on \mathfrak{t}' , and thus for nonzero λ we have that

$$N_{\lambda} \subset \bigoplus_{\alpha_0} U_{\alpha_0}.$$

After basechanging to C, we see that

$$N_{\lambda} \subset \bigoplus_{\alpha} U_{\alpha}$$
.

Thus T'_C acts on N_λ via a subset of the nontrivial characters given by the simple roots $\Phi \subset X^*(T'_C)$, and therefore the trivial representation of T' does not occur in N_λ and thus it does not occur in E.

6. Proof of the main theorems

There are two final ingredients that are introduced in this section. In Section 6.1, we prove the local stabiliser principle of Chai and Oort [2009, Theorem 9.5], which shows that the formal completion of the Zariski closure of a prime-to-p Hecke orbit is stable under the action of a large p-adic Lie group. In Section 6.2.2 we give a summary of results of [Chai 2003], which relates Serre–Tate coordinates of families of ordinary abelian varieties to the p-adic monodromy groups of these abelian varieties. Then in Section 6.3 we put everything together to prove Theorem I. In Section 6.4 we prove Corollary 6.4.1, which is a generalisation of Theorem I to Shimura varieties of abelian type.

We will use the notation introduced in Section 2 and Section 5.1 and moreover we will keep track of the level again. Moreover, all our schemes will now implicitly live over $\overline{\mathbb{F}}_p$. Let $x \in \operatorname{Sh}_{G,K^pK_p}(\overline{\mathbb{F}}_p)$ and let \tilde{x} be a lift of x to $\operatorname{Sh}_{G,K_p}(\overline{\mathbb{F}}_p)$. Then the *prime-to-p Hecke orbit* of x is defined to be the image $H_{K^p}(x) \subset \operatorname{Sh}_{G,K^pK_p}(\overline{\mathbb{F}}_p)$ of the orbit $G(\mathbb{A}_f^p) \cdot \tilde{x} \subset \operatorname{Sh}_{G,K_p}(\overline{\mathbb{F}}_p)$; it does not depend on the choice of lift \tilde{x} . For the rest of this section we fix x as above and we let $Z \subset \operatorname{Sh}_{G,\operatorname{ord},K^pK_p}$ be the closure of $H_{K^p}(x)$; note that Z is again $G(\mathbb{A}_f^p)$ -stable by Lemma 3.1.2.

6.1. Rigidity of Zariski closures of Hecke orbits. Let $z \in Z(\overline{\mathbb{F}}_p)$ be a smooth point of Z and let $I_z(\mathbb{Q})$ be the group of self-quasi-isogenies of z respecting the tensors, which was introduced in Section 2.2. Let $Y = A_z[p^{\infty}]$ and fix a choice of isomorphism $\mathbb{D}_{cov}(Y) \simeq V_p^* \otimes \mathbb{Z}_p$ sending $s_{\alpha} \otimes 1$ to $s_{\alpha, cris}$ as in Section 5.3. This gives rise to an element $b_z = b \in G(\mathbb{Q}_p)$ and we let $U_p \subset J_b(\mathbb{Q}_p)$ be the compact open subgroup introduced in Section 5.4.4. Let $I_z(\mathbb{Z}_{(p)})$ be the intersection of $I_z(\mathbb{Q})$ with U_p inside $J_b(\mathbb{Q}_p)$. We consider the closed immersion of formal neighbourhoods (where the notation is as in (5.1.2))

$$Z^{/z} \subset \operatorname{Sh}_{G,K^pK_n}^{/z} \subset \operatorname{Sh}_{\operatorname{GSp},\mathcal{K}^p\mathcal{K}_n}^{/z}$$
.

The goal of this section is to prove the following result.

Proposition 6.1.1 (local stabiliser principle). The closed subscheme $Z^{/z} \subset \operatorname{Sh}_{G,K^pK_p}^{/z}$ is stable under the action of $I_z(\mathbb{Z}_{(p)})$ via $I_z(\mathbb{Z}_{(p)}) \to U_p$.

6.1.2. There is a $G(\mathbb{A}_f^p)$ -equivariant closed immersion (using the fact that we have a closed immersion at finite level by the main theorem of [Xu 2020]) $\operatorname{Sh}_{G,K_p} \to \operatorname{Sh}_{\mathcal{G}_V,\mathcal{K}_p}$, where $G(\mathbb{A}_f^p)$ acts on the right hand side via the inclusion $G(\mathbb{A}_f^p) \to \mathcal{G}_V(\mathbb{A}_f^p)$. The space $\operatorname{Sh}_{\mathcal{G}_V,\mathcal{K}_p}$ is a moduli space of (weakly) polarised abelian varieties (A,λ) up to prime-to-p isogeny, equipped with an isomorphism $\mathbb{V}^pA \to V \otimes \underline{\mathbb{A}}_f^p$ compatible with the polarisation up to a scalar in $\underline{\mathbb{A}}_f^{p,\times}$.

Let \tilde{z} be a lift of z to $\operatorname{Sh}_{G,K_p}(\overline{\mathbb{F}}_p)$ as above, which defines an inclusion

$$I_z(\mathbb{Z}_{(p)}) \subset I_z(\mathbb{Q}) \subset G(\mathbb{A}_f^p).$$

The stabiliser in $\mathcal{G}_V(\mathbb{A}_f^p)$ of $\tilde{z} \in \operatorname{Sh}_{\mathcal{G}_V,\mathcal{K}_p}$ is given by $\operatorname{End}_{\lambda}(A_z)^{\times}$, which is the group of automorphisms of the abelian variety up to prime-to-p isogeny A that take λ to a $\mathbb{Z}_{(p)}^{\times}$ multiple of λ .

Lemma 6.1.3. The stabiliser inside $G(\mathbb{A}_f^p)$ of the point \tilde{z} is equal to $I_z(\mathbb{Z}_{(p)})$.

Proof. By [KMS 2022, Lemma 2.1.4], the stabiliser is contained in $I_z(\mathbb{Z}_{(p)})$. The stabiliser in $\mathcal{G}_V(\mathbb{A}_f^p)$ of the image of \tilde{z} in $\operatorname{Sh}_{\mathcal{G}_V,\mathcal{K}_p}$ is equal to $\operatorname{End}_{\lambda}(A_z)^{\times}$ and thus contains $I_z(\mathbb{Z}_{(p)})$. The result follows.

In order to prove Proposition 6.1.1, we first prove it for $Sh_{GSp,\mathcal{K}^p\mathcal{K}_p}$. See [Chai and Oort 2009, Theorem 9.5] for closely related results and arguments.

Let $\operatorname{Sh}_{\mathcal{G}_V,\mathcal{K}_p}^{/\bar{z}}$ be the formal completion of $\operatorname{Sh}_{\mathcal{G}_V,\mathcal{K}_p}$, considered as a formal algebraic space as in [Stacks 2020, Section 0AIX], and restrict its functor of points to Artin local $\overline{\mathbb{F}}_p$ -algebras R with residue field isomorphic to $\overline{\mathbb{F}}_p$. Then $\operatorname{Sh}_{\mathcal{G}_V,\mathcal{K}_p}^{/\bar{z}}(R)$ is the set of isomorphism classes of (weakly) polarised abelian varieties (A,λ) over R up to prime-to-p isogeny, equipped with an isomorphism $\epsilon: \mathbb{V}^p A \to V \otimes \underline{\mathbb{A}}_f^p$

compatible with the polarisation up to a scalar in $\underline{\mathbb{A}}_f^{p,\times}$, such that after basechanging to $\overline{\mathbb{F}}_p$ we recover the point given by the image of \tilde{z} .

This means that there is a (necessarily unique) isomorphism $\beta: A_{\overline{\mathbb{F}}_p} \to A_z$ making the following diagram commute:

The quadruple $(A, \lambda, \beta, \epsilon)$ is uniquely determined by (A, λ, β) because (pro-)étale sheaves on Artin local rings are determined by their restriction to the residue field. In particular, for all $R \in$ Art the natural forgetful map

$$\operatorname{Sh}_{\mathcal{G}_V,\mathcal{K}_p}^{/\tilde{z}}(R) \to \operatorname{Sh}_{\operatorname{GSp},\mathcal{K}^p\mathcal{K}_p}^{/z}(R)$$

is an isomorphism. This induces an action of $\operatorname{End}_{\lambda}(A_z)^{\times}$ on $\operatorname{Sh}_{\operatorname{GSp},\mathcal{K}^p\mathcal{K}_p}^{/z}$ that we will now identify.

6.1.4. Recall that there is an inclusion $\operatorname{End}_{\lambda}(A_z)^{\times} \subset \mathcal{G}_V(\mathbb{A}_f^p)$ determined by \tilde{z} or rather $\epsilon_{\tilde{z}}$. This means that an automorphism f of A_z acts on $V \otimes \mathbb{A}_f^p$ in a way that makes the following diagram commute:

$$\mathbb{V}^{p} A_{z} \xrightarrow{f} \mathbb{V}^{p} A_{z}
\downarrow^{\epsilon_{\bar{z}}} \qquad \downarrow^{\epsilon_{\bar{z}}}
V \otimes \mathbb{A}_{f}^{p} \xrightarrow{f} V \otimes \mathbb{A}_{f}^{p}$$

Since $\operatorname{End}_{\lambda}(A_z)^{\times}$ stabilises \tilde{z} , it acts on $\operatorname{Sh}_{\mathcal{G}_V,\mathcal{K}_p}^{/\tilde{z}}$. This action can be described as follows: An automorphism f sends a triple (A,λ,ϵ) to $(A,\lambda,f\circ\epsilon)$. It is straightforward to check that the unique upgrade $(A,\lambda,f\circ\epsilon)$ to a quadruple $(A,\lambda,\beta',f\circ\epsilon)$ is realised by taking $\beta'=f\circ\beta$. Therefore the induced action of $\operatorname{End}_{\lambda}(A_z)^{\times}$ on $\operatorname{Sh}_{\mathrm{GSp},\mathcal{K}^p\mathcal{K}_p}^{/z}$ is given by $(A,\lambda,\beta)\mapsto (A,\lambda,f\circ\beta)$.

6.1.5. Because deformations of abelian varieties are uniquely determined by deformations of their p-divisible groups, we can also identify

$$\operatorname{Sh}_{\mathcal{G}_V,\mathcal{K}_p}^{/\tilde{z}}(R)$$

with the space of triples (X, λ, β) where (X, λ) is a polarised p-divisible group and β is an isomorphism $(X, \lambda)_{\overline{\mathbb{F}}_p} \to (A_z[p^{\infty}], \lambda_z)$. The action of $\operatorname{End}_{\lambda}(A_z)^{\times}$ is then given by postcomposing β with f. There is a similar description of at finite level, and it follows that the natural map

$$\operatorname{Sh}_{\mathrm{GSp},\mathcal{K}^p\mathcal{K}_p}^{/z}\subset\operatorname{Def}(A_z[p^\infty])$$

is $\operatorname{End}_{\lambda}(A_z)^{\times}$ -equivariant, where $\operatorname{End}_{\lambda}(A_z)^{\times}$ acts on the right hand side via the inclusion

$$\operatorname{End}_{\lambda}(A_z)^{\times} \subset \operatorname{End}(A_z[p^{\infty}])^{\times},$$

followed by the natural action of $\operatorname{End}(A_z[p^{\infty}])^{\times}$ on $\operatorname{Def}(A_z[p^{\infty})]$.

Proof of Proposition 6.1.1. Let \tilde{z} be a lift of z to $Sh_{G,K_p}(\overline{\mathbb{F}}_p)$ as above, which defines an inclusion

$$I_z(\mathbb{Z}_{(p)}) \subset I_z(\mathbb{Q}) \subset G(\mathbb{A}_f^p).$$

It follows from Lemma 6.1.3 that $I_z(\mathbb{Z}_{(p)}) \subset G(\mathbb{A}_f^p)$ is the stabiliser of the point \tilde{z} under the action of $G(\mathbb{A}_f^p)$. Let \widetilde{Z} be the inverse image of Z in $\operatorname{Sh}_{G,K_p}$, it is stable under the action of $G(\mathbb{A}_f^p)$ by Lemma 3.1.2. There is a commutative diagram

$$\widetilde{Z} \longrightarrow \operatorname{Sh}_{G,K_p} \longrightarrow \operatorname{Sh}_{\mathcal{G}_V,\mathcal{K}_p}$$

$$\downarrow \qquad \qquad \downarrow \qquad \qquad \downarrow$$

$$Z \longrightarrow \operatorname{Sh}_{G,K^pK_p} \longrightarrow \operatorname{Sh}_{\operatorname{GSp},\mathcal{K}^p\mathcal{K}_p}$$

where the top right horizontal map is $G(\mathbb{A}_f^p)$ -equivariant via $G(\mathbb{A}_f^p) \to \mathcal{G}_V(\mathbb{A}_f^p)$.

Let $\widetilde{Z}^{/\tilde{z}}$ be the formal completion of \widetilde{Z} at the closed point corresponding to \widetilde{Z} , considered as a formal algebraic space as in [Stacks 2020, Section 0AIX]. This is per definition the subfunctor of \widetilde{Z} consisting of those morphisms Spec $T \to \widetilde{Z}$ that factor through \widetilde{z} on the level of topological spaces. Since $I_z(\mathbb{Z}_{(p)})$ stabilises \widetilde{z} , it acts on $\widetilde{Z}^{/\widetilde{z}}$.

By [Stacks 2020, Lemma 0CUF], there is a homeomorphism $|\widetilde{Z}| \simeq \varprojlim_{U^p} |Z_{U^p}|$ and thus we get an isomorphism

$$\widetilde{Z}^{/ ilde{z}} \simeq arprojlim_{U^p \subset G(\mathbb{A}_f^p)} Z_{U^p}^{/z},$$

where $z \in Z_{U^p}(\overline{\mathbb{F}}_p)$ is the image of \tilde{z} under $\widetilde{Z} \to Z_{U^p}$. The formal algebraic space $Z_{U^p}^{/z}$ can be identified with Spf $\widehat{\mathcal{O}}_{Z_{U^p},z}$, compatible with changing U^p . Since the transition morphisms are all finite étale, they induce isomorphisms of complete local rings. Therefore, all the transition maps in the inverse system $\varprojlim_{U^p \subset G(\mathbb{A}_f^p)} Z_{U^p}^{/z}$ are isomorphism. We conclude that

$$\widetilde{Z}^{/ ilde{z}}\simeq Z_{U^p}^{/z},$$

and so there is an action of $I_z(\mathbb{Z}_{(p)})$ on $Z_{U^p}^{/z}$. In the same way we can prove that there is an action of $I_z(\mathbb{Z}_{(p)})$ on $\mathrm{Sh}_{G,K^pK_p}^{/z}$. It remains for us to identify this action with the inclusion $I_z(\mathbb{Z}_{(p)}) \to U_p$ followed by the natural action of U_p on $\mathrm{Sh}_{G,K^pK_p}^{/z}$.

Let \tilde{z} be the image of \tilde{z} in $\operatorname{Sh}_{\mathcal{G}_V,\mathcal{K}_p}(\overline{\mathbb{F}}_p)$ and let $z \in \operatorname{Sh}_{\operatorname{GSp},\mathcal{K}^p\mathcal{K}_p}(\overline{\mathbb{F}}_p)$ be its image. Then the stabiliser of \tilde{z} can be identified with the group

$$\operatorname{End}_{\lambda}(A_{z})^{\times} \subset \mathcal{G}_{V}(\mathbb{A}_{f}^{p})$$

as before. The discussion above implies that we have an action of $\operatorname{End}_{\lambda}(A_z)^{\times}$ on $\operatorname{Sh}_{\mathrm{GSp},\mathcal{K}^p\mathcal{K}_p}^{/z}$ such that the closed immersion

$$\operatorname{Sh}_{G,K^pK_p}^{/z}\subset \operatorname{Sh}_{\operatorname{GSp},\mathcal{K}^p\mathcal{K}_p}^{/z}$$

is $I_z(\mathbb{Z}_{(p)})$ -equivariant for the action of $I_z(\mathbb{Z}_{(p)})$ on the right hand side via the map $I_z(\mathbb{Z}_{(p)}) \to \operatorname{End}_{\lambda}(A_z)^{\times}$. But we have seen in Section 6.1.5 that the action of $\operatorname{End}_{\lambda}(A_z)^{\times}$ on $\operatorname{Sh}_{\operatorname{GSp},\mathcal{K}^p\mathcal{K}_{\underline{p}}}^{/z}$ described above agrees with the action of $\operatorname{End}_{\lambda}(A_z)^{\times}$ via the inclusion $\operatorname{End}_{\lambda}(A_z)^{\times} \to \operatorname{Aut}_{\lambda}(A_z[p^{\infty}])(\mathbb{F}_p)$.

Note that the following diagram commutes by construction:

$$I_z(\mathbb{Z}_{(p)}) \longrightarrow \operatorname{End}_{\lambda}(A_z)^{\times}$$

$$\downarrow \qquad \qquad \downarrow$$

$$U_p \longrightarrow \operatorname{Aut}_{\lambda}(A_z[p^{\infty}])(\overline{\mathbb{F}}_p)$$

Thus we see that $Z_{K^p}^{/z}$ is stable under the action of $I_z(\mathbb{Z}_{(p)})$ on $\mathrm{Sh}_{G,K^pK_p}^{/x}$ given by the inclusion $I_z(\mathbb{Z}_{(p)}) \to U_p$ followed by the natural action of U_p on $\mathrm{Sh}_{G,K^pK_p}^{/x}$.

Corollary 6.1.6. Assume that $z \in \operatorname{Sh}_{G,K^pK_p}(\overline{\mathbb{F}}_p)$ is an ordinary point. Then $Z^{/z}$ is a formal subtorus of $\operatorname{Sh}_{G,K^pK_p}^{/z}$.

Proof. The compact open subgroup $U_p \subset J_b(\mathbb{Q}_p)$ acts on $\operatorname{Sh}_{G,K^pK_p}^{/z}$ as explained in Section 5.4.4. By Proposition 6.1.1 the closed subspace $Z^{/z} \subset \operatorname{Sh}_{G,K^pK_p}^{/z}$ is stable under the action of $I_z(\mathbb{Z}_{(p)}) \subset U_p$ and hence of its closure in U_p . The algebraic group $I_{\mathbb{Q}_p} \subset J_b$ has the same rank as J_b by [Kisin 2017, Corollary 2.1.7]. Let $T \subset I$ be a maximal torus, then [Platonov and Rapinchuk 1994, Theorem 7.9] tells us that the topological closure of $I_z(\mathbb{Q}_p)$ in $I_z(\mathbb{Q}_p)$ has finite index in $I_z(\mathbb{Q}_p)$. It follows from this that the closure of $I_z(\mathbb{Z}_{(p)})$ in $I_z(\mathbb{Q}_p)$ contains a compact open subgroup of a maximal torus $I_z(\mathbb{Q}_p)$.

Proposition 5.5.1 then tells us that the assumptions of [Chai 2008, Theorem 4.3] are satisfied. This theorem implies that $Z^{/z}$ is a p-divisible formal subgroup of $Sh_{G,K^pK_p}^{/z}$, in other words, it is a formal subtorus.

6.2. *Monodromy of linear subspaces.* The goal of this section is to prove the following result, which is a consequence of results of [Chai 2003]. Recall that the universal abelian variety A over $Sh_{GSp, ord, \mathcal{K}^p\mathcal{K}_p}$ gives rise to an F-isocrystal \mathcal{M} , see Section 3.3. Let $W \subset Sh_{GSp, ord, \mathcal{K}^p\mathcal{K}_p}$ be a connected smooth closed subscheme, then we say that W is *linear* at a smooth point $z \in W(\overline{\mathbb{F}}_p)$ if

$$W^{/z} \subset \operatorname{Sh}_{\mathrm{GSp},\mathcal{K}^p\mathcal{K}_p}^{/z}$$

is a p-divisible formal subgroup. Let U_W be the unipotent radical of the monodromy group $Mon(W, \mathcal{M}, z)$.

Proposition 6.2.1 (Chai). Let $z \in W(\overline{\mathbb{F}}_p)$ be a smooth point such that W is linear at z. Then we have the inequality

$$\dim W_7 > \dim U_W$$
,

where Dim W_z is the dimension of the local ring $\mathcal{O}_{W,z}$.

Chai proves the stronger statement that this inequality is actually an equality, but we will not need this stronger statement to prove Theorem I.

Our proof of Proposition 6.2.1 is a straightforward application of the results in [Chai 2003, Sections 2–4]. Since [Chai 2003] is an unpublished preprint from 2003, the referee has suggested we include another reference. Thus we give a second proof of Proposition 6.2.1 based on results of [D'Addezio and van Hoften 2022].

6.2.2. For our first proof of Proposition 6.2.1, we need to give a brief summary of [Chai 2003, Sections 2–4]. Consider the closed immersion.

$$W^{/z} \to \operatorname{Sh}_{\mathrm{GSp},\mathcal{K}^p\mathcal{K}_p}^{/z} \hookrightarrow \widehat{\operatorname{Def}}(Y)_{\overline{\mathbb{F}}_p}.$$

Write $R = \widehat{\mathcal{O}_{W,z}}$ and write M for the finite free \mathbb{Z}_p -module $T_p Y_0(\overline{\mathbb{F}}_p) \otimes_{\mathbb{Z}_p} T_p Y_1^{\vee}(\overline{\mathbb{F}}_p)$. Then the morphism $W^{/z} \to \widehat{\mathrm{Def}}(Y)$ corresponds to an element of

$$\widehat{\operatorname{Def}}(Y)(R) = \operatorname{Hom}(M, \widehat{\mathbb{G}}_m(R)) = \operatorname{Hom}(M, 1 + \mathfrak{m}_R)$$

where the first equality is [Katz 1981, Theorem 2.1]. Thus we get a homomorphism $f: M \to 1 + \mathfrak{m}_R$ and we let N_z^{\vee} be its kernel. By [Chai 2003, Proposition 4.2.1, Remark 2.5.1], the \mathbb{Z}_p -module N_z^{\vee} is finite free and the quotient M/N_z^{\vee} is torsion-free. Thus the map

$$W^{/z} \to \mathcal{H}om(M, \widehat{\mathbb{G}}_m)$$

factors through the subtorus

$$\mathscr{H}om(M/N_z^{\vee},\widehat{\mathbb{G}}_m)\subset \mathscr{H}om(M,\widehat{\mathbb{G}}_m),$$

which we can also write as $N_z \otimes_{\mathbb{Z}_p} \widehat{\mathbb{G}}_m \subset M^* \otimes_{\mathbb{Z}_p} \widehat{\mathbb{G}}_m$. Here the * denotes taking \mathbb{Z}_p -linear dual and the morphism $N_z \to M^*$ is the \mathbb{Z}_p -linear dual of the quotient

$$M \to M/N_z^{\vee}$$
.

The following lemma has the same statement as [Chai 2003, Remark 3.14].

Lemma 6.2.3. The subgroup $N_z \otimes_{\mathbb{Z}_p} \widehat{\mathbb{G}}_m$ is the smallest formal subtorus of $\widehat{\operatorname{Def}}(Y)_{\overline{\mathbb{F}}_p}$ through which the map from Spf R factors.

Proof. A subtorus corresponds to a free \mathbb{Z}_p -submodule $N \subset N_z$ such that the quotient N_z/N is torsion free. Write N^{\vee} for the kernel of the map

$$M \to M/N_7^{\vee} = N_7^* \to N^*.$$

If

$$\operatorname{Spf} R \to N_z \otimes_{\mathbb{Z}_p} \widehat{\mathbb{G}}_m$$

factors through $N \otimes_{\mathbb{Z}_p} \widehat{\mathbb{G}}_m$, then it factors through

$$\mathscr{H}om(M/N^{\vee},\widehat{\mathbb{G}}_m)\subset \mathscr{H}om(M,\widehat{\mathbb{G}}_m).$$

Since the kernel of the map $M \to \widehat{\mathbb{G}}_m(R)$ is given by N_z^{\vee} , it follows that $N_z^{\vee} \subset N^{\vee}$ and therefore $M/N^{\vee} = M/N_z^{\vee}$ and thus $N_z = N$.

Proof of Proposition 6.2.1. We specialise the discussion of Section 6.2.2 to the situation of Proposition 6.2.1. In particular, since $W^{/z}$ is assumed to be a formal subtorus, we are in the situation that

$$W^{/z} = N_z \otimes_{\mathbb{Z}_p} \widehat{\mathbb{G}}_m \subset \widehat{\mathrm{Def}}(Y)_{\overline{\mathbb{F}}_p}.$$

Chai [2003, Section 4, Theorem 4.4] proves that the dimension of U_W is equal to the rank of N_z . Thus the rank of N_z is certainly bounded from below by the dimension of U_W . But the rank of N_z is also the dimension of the formal scheme $W^{/z}$ which equals the Krull dimension of $\widehat{\mathcal{O}}_{W,z}$ and also the Krull dimension of $\mathcal{O}_{W,z}$, which proves the theorem.

Second proof of Proposition 6.2.1. By [D'Addezio and van Hoften 2022, Theorem II], the unipotent radical U_W of Mon (W, \mathcal{M}, z) is isomorphic to the monodromy group

$$Mon(W^{/z}, \mathcal{M}, z)$$
.

This monodromy group is defined as in Section 3.3 using the Tannakian category of isocrystals over the formal scheme $W^{/z}$ or equivalently the Tannakian category of isocrystals over the scheme Spec $\widehat{\mathcal{O}}_{W,z}$, see [D'Addezio and van Hoften 2022, Notation 2.2.5]). Thus it suffices to show that the dimension of $W^{/z}$ is bounded from below by the dimension of $\operatorname{Mon}(W^{/z},z)$.

Let $Y = A_z[p^{\infty}]$ as above and write $\mathfrak{a}^+ =: \mathbb{D}_{cov}(Y)$ and $\mathfrak{a} = \mathfrak{a}^+[1/p]$. Write $\mathfrak{b}^+ \subset \mathfrak{a}^+$ for the covariant Dieudonné module of the *p*-divisible group associated to $W^{/z}$ and $\mathfrak{b} = \mathfrak{b}^+[1/p]$. Then in the notation of [D'Addezio and van Hoften 2022, Section 5.5] we have

$$W^{/z} = Z(\mathfrak{b}^+).$$

Now [D'Addezio and van Hoften 2022, Theorem 5.5.3] tells us that there is an inclusion of algebraic groups over $\tilde{\mathbb{Q}}_p$,

$$\operatorname{Mon}(W^{/z}, \mathcal{M}, z) \subset U(\mathfrak{b}) := \mathfrak{b} \otimes_{\check{\mathbb{Q}}_p} \mathbb{G}_a.$$

In particular, the height of the isocrystal $\mathfrak b$ is bounded from below by the dimension of $\operatorname{Mon}(W^{/z}, \mathcal M, z)$. Since $\mathfrak b$ has slope 1, it follows that the dimension of the p-divisible group associated to $\mathfrak b^+$ is also bounded from below by the dimension of $\operatorname{Mon}(W^{/z}, \mathcal M, z)$.

6.3. *Monodromy and conclusion.* Recall from Section 3.2.2 the maps

$$B(G_{\mathbb{Q}_p}) \to B(G_{\mathbb{Q}_p}^{\mathrm{ad}}) \to \prod_{i=1}^n B(G_{i,\mathbb{Q}_p})$$

induced by the decomposition $G^{\mathrm{ad}} = \prod_{i=1}^n G_i$ of (2.2.1). Let $[b_{\mathrm{ord}}] \in B(G, \{\mu^{-1}\})$ be the σ -conjugacy class corresponding to the ordinary locus, and let $[b_{\mathrm{ord},i}]$ be the image of $[b_{\mathrm{ord}}]$ in $B(G_{i,\mathbb{Q}_p})$.

Lemma 6.3.1. For all i the element $[b_{ord,i}]$ is nonbasic.

Proof. By the axioms of a Shimura datum, the $G_i(\overline{\mathbb{Q}}_p)$ -conjugacy class of cocharacters $\{\mu_i^{-1}\}$ induced by $\{\mu^{-1}\}$ is nontrivial for all i. By Lemma 5.4.1, we have an equality $\{\mu_i^{-1}\} = \{\nu_{[b_{\mathrm{ord},i}]}\}$ and so the Newton

cocharacter of $[b_{\text{ord},i}]$ is noncentral for all i. In other words, the σ -conjugacy class $[b_{\text{ord},i}]$ is nonbasic for all i.

Proof of Theorem I. Let $x \in \operatorname{Sh}_{G,K^pK_p}(\overline{\mathbb{F}}_p)$ be an ordinary point and let Z be the Zariski closure (inside $\operatorname{Sh}_{G,\operatorname{ord},K^pK_p})$ of its prime-to-p Hecke orbit. Then Z is $G(\mathbb{A}_f^p)$ -stable by Lemma 3.1.2 and similarly its smooth locus $Z^{\operatorname{sm}} \subset Z$ is $G(\mathbb{A}_f^p)$ -stable by Lemma 3.1.1. Let X be the p-divisible group over Z^{sm} of the universal abelian variety and let \mathcal{M}^{\dagger} be the associated overconvergent F-isocrystal, see Section 3.3.

Let $z \in Z^{\mathrm{sm}}(\overline{\mathbb{F}}_p)$ and let $Z^{\circ} \subset Z^{\mathrm{sm}}$ be the connected component containing z. By Lemma 6.3.1, the element $[b_{\mathrm{ord}}]$ is \mathbb{Q} -nonbasic and by Lemma 2.3.2, we know that Hypothesis 2.3.1 is satisfied because K_p is hyperspecial. Therefore Corollary 3.3.3 tells us that the monodromy group of \mathcal{M}^{\dagger} over Z° is isomorphic to $G^{\mathrm{der}} \otimes \check{\mathbb{Q}}_p$. Corollary 3.3.5 tells us that unipotent radical of the monodromy group of \mathcal{M} over Z° is isomorphic to the unipotent radical of the parabolic subgroup $P_{\nu_{[b]}} \subset G \otimes \check{\mathbb{Q}}_p$ for any choice of $\nu_{[b]} \in \{\nu_{[b]}\}$.

By Lemma 5.4.1, this unipotent radical is isomorphic to the unipotent radical of the parabolic subgroup $P_{\mu} \subset G$ for any choice of representative μ of $\{\mu\}$. This unipotent radical has dimension equal to $\langle 2\rho, \{\mu\} \rangle$ (this notation was introduced after the statement of Lemma 5.4.1).

Corollary 6.1.6 tells us that $Z^{/z}$ is a formal subtorus. Applying Proposition 6.2.1 we see that the Krull dimension of $\mathcal{O}_{Z,z}$ is bounded from below by $\langle 2\rho, \{\mu\} \rangle$. Since the Shimura variety Sh_{G,K^pK_p} also has dimension $\langle 2\rho, \{\mu\} \rangle$, we conclude that

$$Z^{/z} = \operatorname{Sh}_{G,K^pK_p}^{/z}.$$

Because this is true for a dense set of points, it follows that Z is a union of connected components of $Sh_{G, \text{ord}, K^pK_p}$.

By Lemma 5.1.3, the ordinary locus is dense and thus $\pi_0(\operatorname{Sh}_{G,\operatorname{ord},K^pK_p}) = \pi_0(\operatorname{Sh}_{G,K^pK_p})$. Since $G(\mathbb{A}_f^p)$ acts transitively on $\pi_0(\operatorname{Sh}_{G,K_p})$, by [Kisin 2010, Lemma 2.2.5] in combination with [Madapusi Pera 2019, Corollary 4.1.11], it follows that $Z = \operatorname{Sh}_{G,\operatorname{ord},K^pK_p}$. We conclude that the prime-to-p Hecke orbit of x is dense in $\operatorname{Sh}_{G,K^pK_p}$ since $\operatorname{Sh}_{G,\operatorname{ord},K^pK_p}$ is dense in $\operatorname{Sh}_{G,K^pK_p}$.

6.4. Consequences for Shimura varieties of abelian type. Let (G, X) be a Shimura datum of abelian type with reflex field E, and let (G^{ad}, X^{ad}) be the induced adjoint Shimura datum with reflex field $E^{ad} \subset E$. Let p be a prime number, let $K_p \subset G(\mathbb{Q}_p)$ be a hyperspecial subgroup and let $K^p \subset G(\mathbb{A}_f^p)$ be a sufficiently small compact open subgroup. Let $\operatorname{Sh}_{G,K^pK_p}$ be the special fibre of the canonical integral model of the Shimura variety of level K^pK_p at a prime v above p of E, constructed by Kisin [2010] (see [Kim and Madapusi Pera 2016] for the case p = 2).

By [Shen and Zhang 2022, Theorem A], there is an open and dense $G(\mathbb{A}_f^p)$ -stable Newton stratum $\operatorname{Sh}_{G,K^pK_p,\mu-\operatorname{ord}}$ in $\operatorname{Sh}_{G,K^pK_p}$, called the μ -ordinary locus. If $(G,X)\subset (\mathcal{G}_V,\mathcal{H}_V)$ for some symplectic space V and $E_v=\mathbb{Q}_p$, then the μ -ordinary locus is equal to the ordinary locus by Lemma 5.1.3.

Corollary 6.4.1. If $E_v^{\mathrm{ad}} = \mathbb{Q}_p$, then the prime-to-p Hecke orbit of $x \in \mathrm{Sh}_{G,K^pK_p,\mu-\mathrm{ord}}(\overline{\mathbb{F}}_p)$ is dense in Sh_{G,K^pK_p} .

Remark 6.4.2. This corollary is more general than Theorem I even for Shimura varieties of Hodge type. Indeed, there are (many) examples of Shimura data (G, X) of Hodge type and primes p and v such that $E_v \neq \mathbb{Q}_p$ but $E_v^{\mathrm{ad}} = \mathbb{Q}_p$.

Lemma 6.4.3. Corollary 6.4.1 holds for (G, X) if and only if it holds for (G^{ad}, X^{ad}) .

Proof. The image K_p^{ad} in $G^{\mathrm{ad}}(\mathbb{Q}_p)$ is a hyperspecial subgroup. We can choose $K^{p,\mathrm{ad}} \subset G^{\mathrm{ad}}(\mathbb{A}_f^p)$ containing the image of K^p such that there is a morphism

$$\mathbf{Sh}_{G,K^pK_p}(G,X) \to \mathbf{Sh}_{G^{\mathrm{ad}},K^{p,\mathrm{ad}}K_p^{\mathrm{ad}}}(G^{\mathrm{ad}},X^{\mathrm{ad}}) \otimes_{E^{\mathrm{ad}}} E,$$

inducing a morphism on geometric special fibres of integral canonical models

$$\operatorname{Sh}_{G,K^{p}K_{p}} \to \operatorname{Sh}_{G^{\operatorname{ad}},K^{p,\operatorname{ad}}K_{p}^{\operatorname{ad}}}, \tag{6.4.1}$$

where we are taking the canonical integral model of (G^{ad}, X^{ad}) at the place v^{ad} of E^{ad} induced by v. This morphism induces a $G(\mathbb{A}_f^p)$ -equivariant morphism

$$\operatorname{Sh}_{G,K_p} \to \operatorname{Sh}_{G^{\operatorname{ad}},K_p^{\operatorname{ad}}},$$

where $G(\mathbb{A}_f^p)$ acts on the left hand side via the natural map $G(\mathbb{A}_f^p) \to G^{\mathrm{ad}}(\mathbb{A}_f^p)$. Since the Newton stratification on Shimura varieties of abelian type can be constructed using the *F*-crystals with *G*-structure of Lovering [2017], which are functorial for morphisms of Shimura data, it follows that there is an induced map

$$\mathrm{Sh}_{G,K^pK_p,\mu\mathrm{-ord}} o \mathrm{Sh}_{G^{\mathrm{ad}},K^{p,\mathrm{ad}}K_n^{\mathrm{ad}},\mu\mathrm{-ord}}$$
.

Moreover, since the natural map $B(G, \{\mu^{-1}\}) \to B(G^{\mathrm{ad}}, \{\mu^{-1}\})$ is a bijection as explained in [Kottwitz 1997, Section 6.5], it is in fact true that $\mathrm{Sh}_{G,K^pK_p,\mu-\mathrm{ord}}$ is the inverse image of $\mathrm{Sh}_{G^{\mathrm{ad}},K^{p,\mathrm{ad}}K_p^{\mathrm{ad}},\mu-\mathrm{ord}}$ under (6.4.1). By construction of the integral canonical models of Shimura varieties of abelian type, see [Kisin 2010, Section 3.4.9; 2017, Appendix E.7], the connected components of $\mathrm{Sh}_{G^{\mathrm{ad}},K^{p,\mathrm{ad}}K_p^{\mathrm{ad}}}$ are quotients of connected components of Sh_{G,K^pK_p} by free actions of finite groups. In particular, the map (6.4.1) is finite étale and thus closed.

Because the map (6.4.1) is closed and takes prime-to-p Hecke orbits to prime-to-p Hecke orbits, it must takes Zariski closures of prime-to-p Hecke orbits to Zariski closure of prime-to-p Hecke orbits. Thus for $x \in \operatorname{Sh}_{G,K^pK_p,\mu-\operatorname{ord}}(\overline{\mathbb{F}}_p)$ the Zariski closure of its Hecke orbit in $\operatorname{Sh}_{G,K^pK_p,\mu-\operatorname{ord}}$ has the same dimension as the Zariski closure of its Hecke orbit in $\operatorname{Sh}_{G^{\operatorname{ad}},K^p,\operatorname{ad}}K_p^{\operatorname{ad}}$ Moreover in both cases the prime-to-p Hecke operators act transitively on $\pi_0(\operatorname{Sh}_{G,K^pK_p})$ by [Kisin 2010, Lemma 2.2.5] in combination with [Madapusi Pera 2019, Corollary 4.1.11]. Thus prime-to-p Hecke orbits in $\operatorname{Sh}_{G,K^pK_p,\mu-\operatorname{ord}}$ are dense if

⁶The result [Madapusi Pera 2019, Corollary 4.1.11] states that for Shimura varieties of Hodge type and hyperspecial level, Hypothesis 3.4.1 holds. Since the canonical integral models of Shimura varieties of abelian type are constructed from the canonical integral models of Shimura varieties of Hodge type, the statement therefore also holds for canonical integral models of Shimura varieties of abelian type.

and only if their images under (6.4.1) are dense. In particular, if the corollary holds for (G^{ad}, X^{ad}) , then it holds for (G, X).

To prove the converse, we note that a point in the Shimura variety for (G^{ad}, X^{ad}) can, by [Kisin 2010, Lemma 2.2.5] in combination with [Madapusi Pera 2019, Corollary 4.1.11], be moved to a connected component which is in the image of (6.4.1). Therefore every prime-to-p Hecke orbit can be lifted to the Shimura variety for (G, X), and we are done.

Proof of Corollary 6.4.1. By Lemma 6.4.3, we may assume that G is adjoint. Then by the proof of [Kisin and Pappas 2018, Lemma 4.6.22] we can choose a Shimura datum of Hodge type (G_2, X_2) and a morphism of Shimura data $(G_2, X_2) \to (G, X)$ such that: the group G_{2,\mathbb{Q}_p} is quasi-split and split over an unramified extension and the prime v of E splits in the reflex field $E_2 \supset E$ of (G_2, X_2) . The upshot is that we can choose a prime w of E_2 satisfying $E_{2,w} = \mathbb{Q}_p$ and thus the μ -ordinary locus in the special fibre of the canonical integral model for (G_2, X_2) at this prime is equal to the ordinary locus for a choice of Hodge embedding $(\mathcal{G}_V, \mathcal{H}_V)$.

Then Theorem I implies that Corollary 6.4.1 holds for (G_2, X_2) and Lemma 6.4.3 tells us that it also holds for (G, X) which concludes the proof.

Acknowledgements

It is clear that our approach owes a substantial intellectual debt to the work of Chai and Oort, and in fact our main results were conceived after reading a remark in [Chai 2006]. We are very grateful to the referee for pointing out a serious error in a previous version, for a very detailed reading of a second version and for many helpful comments. We thank Sean Cotner for helpful discussions.

References

[Berthelot 1996] P. Berthelot, "Cohomologie rigide et cohomologie rigide à supports propres", prépublication 96-03, Institut de recherche mathématique de Rennes, 1996, https://perso.univ-rennes1.fr/pierre.berthelot/publis/Cohomologie_Rigide_I.pdf.

[Borel 1991] A. Borel, Linear algebraic groups, 2nd ed., Graduate Texts in Mathematics 126, Springer, 1991. MR Zbl

[Borel and Tits 1965] A. Borel and J. Tits, "Groupes réductifs", Inst. Hautes Études Sci. Publ. Math. 27 (1965), 55–150. MR Zbl

[Caraiani and Scholze 2017] A. Caraiani and P. Scholze, "On the generic part of the cohomology of compact unitary Shimura varieties", *Ann. of Math.* (2) **186**:3 (2017), 649–766. MR Zbl

[Chai 1995] C.-L. Chai, "Every ordinary symplectic isogeny class in positive characteristic is dense in the moduli", *Invent. Math.* **121**:3 (1995), 439–479. MR Zbl

[Chai 2003] C.-L. Chai, "Families of ordinary abelian varieties: canonical coordinates, *p*-adic monodromy, Tate-linear subvarieties and Hecke orbits", preprint, University of Pennsylvania, 2003, https://www2.math.upenn.edu/~chai/papers_pdf/fam_ord_av.pdf.

[Chai 2005] C.-L. Chai, "Monodromy of Hecke-invariant subvarieties", Pure Appl. Math. Q. 1:2 (2005), 291–303. MR Zbl

[Chai 2006] C.-L. Chai, "Hecke orbits as Shimura varieties in positive characteristic", pp. 295–312 in *International Congress of Mathematicians* (Madrid, 2006), vol. 2: Invited lectures, edited by M. Sanz-Solé et al., Eur. Math. Soc., Zürich, 2006. MR Zbl

[Chai 2008] C.-L. Chai, "A rigidity result for p-divisible formal groups", Asian J. Math. 12:2 (2008), 193–202. MR Zbl

[Chai and Oort 2009] C.-L. Chai and F. Oort, "Moduli of abelian varieties and *p*-divisible groups", pp. 441–536 in *Arithmetic geometry* (Göttingen, Germany, 2006), edited by H. Darmon et al., Clay Math. Proc. **8**, Amer. Math. Soc., Providence, RI, 2009. MR Zbl

[Conrad et al. 2015] B. Conrad, O. Gabber, and G. Prasad, *Pseudo-reductive groups*, 2nd ed., New Mathematical Monographs **26**, Cambridge University Press, 2015. MR Zbl

[D'Addezio 2020] M. D'Addezio, "The monodromy groups of lisse sheaves and overconvergent *F*-isocrystals", *Selecta Math.* (*N.S.*) **26**:3 (2020), art. id. 45. MR Zbl

[D'Addezio 2023] M. D'Addezio, "Parabolicity conjecture of F-isocrystals", Ann. of Math. (2) 198:2 (2023), 619–656. MR Zbl

[D'Addezio and van Hoften 2022] M. D'Addezio and P. van Hoften, "Hecke orbits on Shimura varieties of Hodge type", preprint, 2022. arXiv 2205.10344

[Deligne 1980] P. Deligne, "La conjecture de Weil, II", Inst. Hautes Études Sci. Publ. Math. 52 (1980), 137-252. MR Zbl

[Étesse 2002] J.-Y. Étesse, "Descente étale des *F*-isocristaux surconvergents et rationalité des fonctions *L* de schémas abéliens", *Ann. Sci. École Norm. Sup.* (4) **35**:4 (2002), 575–603. MR Zbl

[Glockner 2016] H. Glockner, "Lectures on Lie groups over local fields", preprint, 2016. arXiv 0804.2234

[He et al. 2021] X. He, R. Zhou, and Y. Zhu, "Stabilizers of irreducible components of affine Deligne–Lusztig varieties", preprint, 2021. arXiv 2109.02594

[van Hoften 2020] P. van Hoften, "Mod p points on Shimura varieties of parahoric level (with an appendix by Rong Zhou)", preprint, 2020. arXiv 2010.10496

[Howe 2020] S. Howe, "A unipotent circle action on *p*-adic modular forms", *Trans. Amer. Math. Soc. Ser. B* **7** (2020), 186–226. MR Zbl

[Kasprowitz 2012] R. Kasprowitz, "Monodromy of subvarieties of PEL-Shimura varieties", preprint, 2012. arXiv 1209.5891

[Katz 1979] N. M. Katz, "Slope filtration of F-crystals", pp. 113–163 in Journées de Géométrie Algébrique de Rennes, I (Rennes, France, 1978), Astérisque 63, Soc. Math. France, Paris, 1979. MR Zbl

[Katz 1981] N. Katz, "Serre-Tate local moduli", pp. 138–202 in *Surfaces algébriques* (Orsay, France, 1976–78), edited by J. Giraud et al., Lecture Notes in Math. **868**, Springer, 1981. MR Zbl

[Kim 2019] W. Kim, "On central leaves of Hodge-type Shimura varieties with parahoric level structure", *Math. Z.* **291**:1-2 (2019), 329–363. MR Zbl

[Kim and Madapusi Pera 2016] W. Kim and K. Madapusi Pera, "2-adic integral canonical models", Forum Math. Sigma 4 (2016), art. id. e28. Correction in 8 (2020), e14. MR Zbl

[Kisin 2010] M. Kisin, "Integral models for Shimura varieties of abelian type", J. Amer. Math. Soc. 23:4 (2010), 967–1012. MR Zbl

[Kisin 2017] M. Kisin, "Mod p points on Shimura varieties of abelian type", J. Amer. Math. Soc. 30:3 (2017), 819–914. MR Zbl

[Kisin and Pappas 2018] M. Kisin and G. Pappas, "Integral models of Shimura varieties with parahoric level structure", *Publ. Math. Inst. Hautes Études Sci.* **128** (2018), 121–218. MR Zbl

[Kisin and Zhou 2021] M. Kisin and R. Zhou, "Independence of ℓ for Frobenius conjugacy classes attached to abelian varieties", preprint, 2021. arXiv 2103.09945

[Kisin et al. 2021] M. Kisin, S. W. Shin, and Y. Zhu, "The stable trace formula for Shimura varieties of abelian type", preprint, 2021. arXiv 2110.05381

[KMS 2022] M. Kisin, K. Madapusi Pera, and S. W. Shin, "Honda–Tate theory for Shimura varieties", *Duke Math. J.* 171:7 (2022), 1559–1614. MR Zbl

[Kottwitz 1982] R. E. Kottwitz, "Rational conjugacy classes in reductive groups", Duke Math. J. 49:4 (1982), 785–806. MR Zbl

[Kottwitz 1985] R. E. Kottwitz, "Isocrystals with additional structure", Compositio Math. 56:2 (1985), 201–220. MR Zbl

[Kottwitz 1997] R. E. Kottwitz, "Isocrystals with additional structure, II", Compositio Math. 109:3 (1997), 255–339. MR Zbl

[Kret and Shin 2021] A. Kret and S. W. Shin, " H^0 of Igusa varieties via automorphic forms", preprint, 2021. arXiv 2102.10690 [Lee 2018] D. U. Lee, "Nonemptiness of Newton strata of Shimura varieties of Hodge type", *Algebra Number Theory* 12:2 (2018), 259–283. MR Zbl

[Lovering 2017] T. Lovering, "Filtered F-crystals on Shimura varieties of abelian type", preprint, 2017. arXiv 1702.06611

[Madapusi Pera 2019] K. Madapusi Pera, "Toroidal compactifications of integral models of Shimura varieties of Hodge type", *Ann. Sci. Éc. Norm. Supér.* (4) **52**:2 (2019), 393–514. MR Zbl

[Maulik et al. 2022] D. Maulik, A. N. Shankar, and Y. Tang, "Picard ranks of K3 surfaces over function fields and the Hecke orbit conjecture", *Invent. Math.* **228**:3 (2022), 1075–1143. MR Zbl

[Milne 2005] J. S. Milne, "Introduction to Shimura varieties", pp. 265–378 in *Harmonic analysis, the trace formula, and Shimura varieties* (Toronto, 2003), edited by J. Arthur et al., Clay Math. Proc. **4**, Amer. Math. Soc., Providence, RI, 2005. MR Zbl

[Noot 1996] R. Noot, "Models of Shimura varieties in mixed characteristic", *J. Algebraic Geom.* **5**:1 (1996), 187–207. MR Zbl [Pál 2022] A. Pál, "The *p*-adic monodromy group of abelian varieties over global function fields of characteristic *p*", *Doc. Math.* **27** (2022), 1509–1579. MR Zbl

[Platonov and Rapinchuk 1994] V. Platonov and A. Rapinchuk, *Algebraic groups and number theory*, Pure and Applied Mathematics **139**, Academic Press, Boston, 1994. MR Zbl

[Rapoport and Richartz 1996] M. Rapoport and M. Richartz, "On the classification and specialization of *F*-isocrystals with additional structure", *Compositio Math.* **103**:2 (1996), 153–181. MR Zbl

[Rapoport and Zink 1996] M. Rapoport and T. Zink, *Period spaces for p-divisible groups*, Annals of Mathematics Studies **141**, Princeton University Press, 1996. MR Zbl

[Scholze and Weinstein 2013] P. Scholze and J. Weinstein, "Moduli of *p*-divisible groups", *Camb. J. Math.* 1:2 (2013), 145–237. MR Zbl

[Shankar 2016] A. N. Shankar, "The Hecke-orbit conjecture for 'modèles étranges", preprint, 2016.

[Shankar and Zhou 2021] A. N. Shankar and R. Zhou, "Serre-Tate theory for Shimura varieties of Hodge type", *Math. Z.* **297**:3-4 (2021), 1249–1271. MR Zbl

[Shen and Zhang 2022] X. Shen and C. Zhang, "Stratifications in good reductions of Shimura varieties of abelian type", *Asian J. Math.* **26**:2 (2022), 167–226. MR Zbl

[Stacks 2020] "The Stacks project", electronic reference, 2020, http://stacks.math.columbia.edu.

[Wortmann 2013] D. Wortmann, "The μ -ordinary locus for Shimura varieties of Hodge type", preprint, 2013. arXiv 1310.6444 [Xiao Xiao 2020] L. Xiao Xiao, "On the Hecke orbit conjecture for PEL type Shimura varieties", preprint, 2020. arXiv 2006.06859

[Xu 2020] Y. Xu, "Normalization in integral models of Shimura varieties of Hodge type", preprint, 2020. arXiv 2007.01275

[Zhou 2023] R. Zhou, "Motivic cohomology of quaternionic Shimura varieties and level raising", *Ann. Sci. Éc. Norm. Supér.* (4) **56**:4 (2023), 1231–1297. MR

Communicated by Frank Calegari

Received 2022-01-26 Revised 2023-02-20 Accepted 2023-07-20

p.van.hoften@vu.nl Mathematics Department, Stanford University, Stanford, CA, United States

Algebra & Number Theory

msp.org/ant

EDITORS

MANAGING EDITOR Antoine Chambert-Loir Université Paris-Diderot France EDITORIAL BOARD CHAIR

David Eisenbud

University of California

Berkeley, USA

BOARD OF EDITORS

Jason P. Bell	University of Waterloo, Canada	Philippe Michel	École Polytechnique Fédérale de Lausanne
Bhargav Bhatt	University of Michigan, USA	Martin Olsson	University of California, Berkeley, USA
Frank Calegari	University of Chicago, USA	Irena Peeva	Cornell University, USA
J-L. Colliot-Thélène	CNRS, Université Paris-Saclay, France	Jonathan Pila	University of Oxford, UK
Brian D. Conrad	Stanford University, USA	Anand Pillay	University of Notre Dame, USA
Samit Dasgupta	Duke University, USA	Bjorn Poonen	Massachusetts Institute of Technology, USA
Hélène Esnault	Freie Universität Berlin, Germany	Victor Reiner	University of Minnesota, USA
Gavril Farkas	Humboldt Universität zu Berlin, Germany	Peter Sarnak	Princeton University, USA
Sergey Fomin	University of Michigan, USA	Michael Singer	North Carolina State University, USA
Edward Frenkel	University of California, Berkeley, USA	Vasudevan Srinivas	Tata Inst. of Fund. Research, India
Wee Teck Gan	National University of Singapore	Shunsuke Takagi	University of Tokyo, Japan
Andrew Granville	Université de Montréal, Canada	Pham Huu Tiep	Rutgers University, USA
Ben J. Green	University of Oxford, UK	Ravi Vakil	Stanford University, USA
Christopher Hacon	University of Utah, USA	Akshay Venkatesh	Institute for Advanced Study, USA
Roger Heath-Brown	Oxford University, UK	Melanie Matchett Wood	Harvard University, USA
János Kollár	Princeton University, USA	Shou-Wu Zhang	Princeton University, USA
Michael J. Larsen	Indiana University Bloomington, USA		

PRODUCTION

production@msp.org

Silvio Levy, Scientific Editor

See inside back cover or msp.org/ant for submission instructions.

The subscription price for 2024 is US \$525/year for the electronic version, and \$770/year (+\$65, if shipping outside the US) for print and electronic. Subscriptions, requests for back issues and changes of subscriber address should be sent to MSP.

Algebra & Number Theory (ISSN 1944-7833 electronic, 1937-0652 printed) at Mathematical Sciences Publishers, 798 Evans Hall #3840, c/o University of California, Berkeley, CA 94720-3840 is published continuously online.

ANT peer review and production are managed by EditFLow® from MSP.

PUBLISHED BY

mathematical sciences publishers

nonprofit scientific publishing

http://msp.org/

© 2024 Mathematical Sciences Publishers

Algebra & Number Theory

Volume 18 No. 5 2024

On the ordinary Hecke orbit conjecture POL VAN HOFTEN	847
Locally analytic vector bundles on the Fargues–Fontaine curve GAL PORAT	899
Multiplicity structure of the arc space of a fat point RIDA AIT EL MANSSOUR and GLEB POGUDIN	947
Theta correspondence and simple factors in global Arthur parameters CHENYAN WU	969
Equidistribution theorems for holomorphic Siegel cusp forms of general degree: the level aspect HENRY H. KIM, SATOSHI WAKATSUKI and TAKUYA YAMAUCHI	993